UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549 **FORM 10-Q**

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) [X] OF THE SECURITIES EXCHANGE ACT OF 1934

	For the quarterly period e	ended June 30, 2018
	OR	
[]	TRANSITION REPORT PURSUAN OF THE SECURITIES EXC	` '
	For the transition period fro	m to
	Commission File No	o.: 000-51826
	MERCER INTERNA (Exact name of Registrant as s	
	Washington	47-0956945
	(State or other jurisdiction	(I.R.S. Employer

Identification No.) of incorporation or organization)

Suite 1120, 700 West Pender Street, Vancouver, British Columbia, Canada, V6C 1G8 (Address of office)

•	604) 684-1099 hone number, including area code)
	and all reports required to be filed by Section 13 or 15(d) of the Securities for such shorter period that the Registrant was required to file such reports), past 90 days. YES \boxtimes NO \square
·	nitted electronically and posted on its corporate Web site, if any, every rsuant to Rule 405 of Regulation S-T during the preceding 12 months or for t and post such files). YES \boxtimes NO \square
·	e accelerated filer, an accelerated filer, a non-accelerated filer, a smaller lefinitions of "large accelerated filer", "accelerated filer", "non-accelerated company" in Rule 12b-2 of the Exchange Act. (Check one):
Large Accelerated Filer ⊠ Accelerated Filer □	Non-Accelerated Filer
Emerging Growth Company	
	if the registrant has elected not to use the extended transition period for adards provided pursuant to Section 13(a) of the Exchange Act. \Box

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). YES □ NO ☒

The Registrant had 65,201,661 shares of common stock outstanding as at July 25, 2018.

PART I. <u>FINANCIAL INFORMATION</u> ITEM 1. FINANCIAL STATEMENTS

MERCER INTERNATIONAL INC. INTERIM CONSOLIDATED FINANCIAL STATEMENTS FOR THE SIX MONTHS ENDED JUNE 30, 2018

(Unaudited)

MERCER INTERNATIONAL INC. INTERIM CONSOLIDATED STATEMENTS OF OPERATIONS

(Unaudited)

(In thousands of U.S. dollars, except per share data)

	 Three Moi Jun			Six Mont Jun		
	2018		2017	2018		2017
Revenues	\$ 346,532	\$	283,177	\$ 714,435	\$	525,961
Costs and expenses						
Operating costs, excluding depreciation and amortization	271,134		230,534	525,419		403,130
Operating depreciation and amortization	22,906		20,521	46,115		39,637
Selling, general and administrative expenses	 15,016		13,259	 29,377		22,985
Operating income	 37,476	_	18,863	113,524		60,209
Other income (expenses)						
Interest expense	(12,128)		(13,320)	(24,243)		(27,199)
Loss on settlement of debt (Note 4(a))	_			(21,515)		(10,696)
Legal cost award (Note 11(c))	_		_	(6,951)		_
Other income (expenses)	(132)		137	(369)		573
Total other expenses	(12,260)		(13,183)	(53,078)		(37,322)
Income before provision for income taxes	25,216		5,680	60,446		22,887
Provision for income taxes	(8,461)		(7,784)	(18,042)		(15,265)
Net income (loss)	\$ 16,755	\$	(2,104)	\$ 42,404	\$	7,622
Net income (loss) per common share						
Basic and diluted	\$ 0.26	\$	(0.03)	\$ 0.65	\$	0.12
Dividends declared per common share	\$ 0.125	\$	0.115	\$ 0.250	\$	0.230

INTERIM CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (LOSS)

(Unaudited) (In thousands of U.S. dollars)

	 Three Moi Jun				Ended ,		
	 2018 2017				2018		2017
Net income (loss)	\$ 16,755	\$	(2,104)	\$	42,404	\$	7,622
Other comprehensive income (loss), net of taxes ⁽¹⁾							
Foreign currency translation adjustment	(55,807)		58,471		(39,522)		69,640
Change in unrecognized losses and prior service costs related							
to defined benefit pension plan	(574)		309		(728)		602
Change in unrealized gains/losses on marketable securities	26		3		27		5
Other comprehensive income (loss), net of taxes ⁽¹⁾	(56,355)		58,783		(40,223)		70,247
Total comprehensive income (loss)	\$ (39,600)	\$	56,679	\$	2,181	\$	77,869

⁽¹⁾ Balances are net of tax effects of \$nil in all periods.

The accompanying notes are an integral part of these interim consolidated financial statements.

MERCER INTERNATIONAL INC. INTERIM CONSOLIDATED BALANCE SHEETS

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

		June 30, 2018	De	ecember 31, 2017
ASSETS				
Current assets				
Cash and cash equivalents	\$	269,482	\$	143,299
Restricted cash to redeem senior notes (Note 4(a))				317,439
Accounts receivable		190,807		206,027
Inventories		189,233		176,601
Prepaid expenses and other		12,041		8,973
Total current assets		661,563		852,339
Property, plant and equipment, net		836,116		844,848
Intangible and other assets		23,299		26,147
Deferred income tax		1,411		1,376
Total assets	\$	1,522,389	\$	1,724,710
LIABILITIES AND SHAREHOLDERS' EQUITY				
Current liabilities				
Accounts payable and other	\$	187,532	\$	133,557
Pension and other post-retirement benefit obligations	Ψ	939	Ψ	985
Senior notes to be redeemed with restricted cash (Note 4(a))				295,924
Total current liabilities		188,471		430,466
Debt		699,261		662,997
Pension and other post-retirement benefit obligations		21,579		21,156
Capital leases and other		37,671		27,464
Deferred income tax		36,905	_	31,961
Total liabilities		983,887	_	1,174,044
Shareholders' equity				
Common shares \$1 par value; 200,000,000 authorized;				
65,202,000 issued and outstanding (2017 – 65,017,000)		65,171		64,974
Additional paid-in capital		340,450		338,695
Retained earnings		232,105		205,998
Accumulated other comprehensive loss		(99,224)		(59,001)
Total shareholders' equity		538,502		550,666
Total liabilities and shareholders' equity	\$	1,522,389	\$	1,724,710
Commitments and contingencies (Note 11)				
Subsequent events (Note 4(c) and Note 8)				
1				

The accompanying notes are an integral part of these interim consolidated financial statements.

MERCER INTERNATIONAL INC. INTERIM CONSOLIDATED STATEMENTS OF CASH FLOWS

(Unaudited) (In thousands of U.S. dollars)

	Three Months Ended June 30,			Six Months June 3				
		2018		2017		2018		2017
Cash flows from (used in) operating activities								
Net income (loss)	\$	16,755	\$	(2,104)	\$	42,404	\$	7,622
Adjustments to reconcile net income (loss) to cash flows from operating act	iviti	es						
Depreciation and amortization		23,014		20,625		46,333		39,846
Deferred income tax provision		1,204		4,196		6,016		8,405
Loss on settlement of debt		_		_		21,515		10,696
Defined benefit pension plan and other post-retirement benefit plan								
expense		432		540		871		1,066
Stock compensation expense		1,759		912		1,952		751
Other		1,311		(153)		2,131		525
Defined benefit pension plan and other post-retirement benefit plan								
contributions		(60)		(319)		(105)		(851)
Changes in working capital								
Accounts receivable		13,475		(37,426)		8,343		(43,714)
Inventories		(12,221)		(5,294)		(19,043)		4,131
Accounts payable and accrued expenses		36,906		36,954		54,933		43,835
Other		3,170		(1,846)		(3,228)		(2,842)
Net cash from (used in) operating activities		85,745		16,085		162,122		69,470
The cash from (asea in) operating activities	_	05,715		10,000	_	102,122	_	05,170
Cash flows from (used in) investing activities								
Purchase of property, plant and equipment		(28,655)		(19,743)		(44,839)		(27,907)
Purchase of intangible assets		(153)		(165)		(320)		(405)
Acquisition of Friesau Facility		`		(61,627)		`		(61,627)
Other		67		77		67		77
Net cash from (used in) investing activities		(28,741)		(81,458)		(45,092)		(89,862)
1100 Guida Holla (Goods III) III (Gotting Gott (III)		(=0,7.11)	_	(01,100)	_	(10,0)2)	_	(0),002)
Cash flows from (used in) financing activities								
Redemption of senior notes		_		_		(317,439)		(234,945)
Proceeds from issuance of notes		_		_		_		250,000
Proceeds from revolving credit facilities, net		17,665		26,525		37,736		26,525
Dividend payments		(8,147)		(7,472)		(16,274)		(14,912)
Payment of interest rate derivative liability				(3,789)		<u> </u>		(3,789)
Payment of debt issuance costs		_		(1,008)		(1,390)		(6,132)
Other		(771)		1,879		(1,619)		958
Net cash from (used in) financing activities		8,747		16,135		(298,986)		17,705
Tive takin nom (akta m) manting attitives		0,7.7	_	10,100	_	(=> 0,> 00)		17,700
Effect of exchange rate changes on cash, cash equivalents and restricted cash		(9,835)		5,916		(9,300)		6,434
Net increase (decrease) in cash, cash equivalents and restricted cash		55,916		(43,322)		(191,256)		3,747
Cash, cash equivalents and restricted cash, beginning of period		213,566		187,965		460,738		140,896
Cash, cash equivalents and restricted cash, end of period	\$	269,482	\$	144,643	\$	269,482	\$	144,643
cash, sash equivalents and restricted easil, ond or period	Ψ	207,402	Ψ	111,073	Ψ	207,702	Ψ	111,043
Supplemental cash flow disclosure								
Cash paid for interest	\$	4,424	\$	16,425	\$	15,696	\$	20,881
Cash paid for income taxes	\$	2,742	\$	2,677	\$	4,220	\$	5,204
		ĺ				, i		·
Supplemental schedule of non-cash investing and financing activities	_							
Leased production equipment	\$	12,126	\$	_	\$	12,126	\$	139

The accompanying notes are an integral part of these interim consolidated financial statements.

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 1. The Company and Summary of Significant Accounting Policies

Nature of Operations and Basis of Presentation

The Interim Consolidated Financial Statements contained herein include the accounts of Mercer International Inc. ("Mercer Inc.") and all of its subsidiaries (collectively the "Company"). The Company's shares of common stock are quoted and listed for trading on the NASDAQ Global Market.

The Interim Consolidated Financial Statements have been prepared by the Company pursuant to the rules and regulations of the United States Securities and Exchange Commission (the "SEC"). The year-end Consolidated Balance Sheet data was derived from audited financial statements. The footnote disclosure included herein has been prepared in accordance with accounting principles generally accepted for interim financial statements in the United States ("GAAP"). The unaudited Interim Consolidated Financial Statements should be read together with the audited Consolidated Financial Statements and accompanying notes included in the Company's latest Annual Report on Form 10-K for the fiscal year ended December 31, 2017. In the opinion of the Company, the unaudited Interim Consolidated Financial Statements contained herein contain all adjustments necessary for a fair statement of the results of the interim periods included. The results for the periods included herein may not be indicative of the results for the entire year.

In these Interim Consolidated Financial Statements, unless otherwise indicated, all amounts are expressed in United States dollars ("U.S. dollars" or "\$"). The symbol "€" refers to euros and the symbol "C\$" refers to Canadian dollars.

Use of Estimates

Preparation of financial statements and related disclosures in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in the financial statements and accompanying notes. Significant management judgment is required in determining the accounting for, among other things, pension and other post-retirement benefit obligations, deferred income taxes (valuation allowance and permanent reinvestment), depreciation and amortization, future cash flows associated with impairment testing for long-lived assets, the allocation of the purchase price in a business combination to the assets acquired and liabilities assumed, legal liabilities and contingencies. Actual results could differ materially from these estimates, and changes in these estimates are recorded when known.

New Accounting Pronouncements

Accounting Pronouncements Implemented

In May 2014, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update 2014-09, Revenue Recognition – Revenue from Contracts with Customers that requires companies to recognize revenue when a customer obtains control rather than when companies have transferred substantially all risks and rewards of a good or service. Additionally, the update provides presentation and disclosure requirements which are more detailed in regards to the nature, amount, timing, and uncertainty of revenue and cash flows arising from contracts with customers. The Company adopted this standard as at January 1, 2018 using the modified retrospective method. The new standard does not change the timing of when the Company recognizes revenue as the majority of the Company's revenue arises from contracts with customers in which the sale of goods is the main performance obligation. The Company's revised revenue recognition disclosure has been included in the Significant Accounting Policies and the Business Segment Information Note.

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 1. The Company and Summary of Significant Accounting Policies (continued)

In March 2017, the FASB issued Accounting Standards Update 2017-07, Improving the Presentation of Net Periodic Pension Cost and Net Periodic Post-Retirement Benefit Cost which requires that an employer report the service cost component in the same line item or items as other compensation costs arising from services rendered by the pertinent employees during the period. The other components of net benefit cost are required to be presented in the income statement separately from the service cost component and outside a subtotal of income from operations. The Company adopted this standard as at January 1, 2018. For the three and six month periods ended June 30, 2018, \$289 and \$582 of the net benefit cost, respectively, has been recorded in other income (expenses) in the Interim Consolidated Statement of Operations. For the three and six month periods ended June 30, 2017, \$376 and \$736, respectively, has been reclassified from operating costs, excluding depreciation and amortization to other income (expenses) in the Interim Consolidated Statement of Operations.

In January 2018, the FASB released guidance on the accounting for tax on the global intangible low-taxed income ("GILTI") provisions of the Tax Cuts and Jobs Act (the "Act"). The GILTI provisions impose a tax on foreign income in excess of a deemed return on tangible assets of foreign corporations. The Company has elected to treat any potential GILTI inclusions as a period cost.

Accounting Pronouncements Not Yet Implemented

In February 2016, the FASB issued Accounting Standards Update 2016-02, Leases ("ASU 2016-02") which requires lessees to recognize virtually all of their leases on the balance sheet, by recording a right-of-use asset and liability. This update is effective for financial statements issued for fiscal years beginning after December 15, 2018, with early adoption permitted at the beginning of an interim or annual reporting period. The Company is currently assessing the impact the adoption of ASU 2016-02 will have on its consolidated financial statements.

In February 2018, the FASB issued Accounting Standards Update 2018-02, Income Statement - Reporting Comprehensive Income which allows a reclassification from accumulated other comprehensive income to retained earnings for stranded tax effects resulting from the Act. This standard is effective for fiscal years beginning after December 15, 2018, and should be applied either in the period of adoption or retrospectively to each period (or periods) in which the effect of the change in the U.S. federal corporate income tax rate in the Act is recognized. The Company believes this new standard will not have an impact on its consolidated financial statements.

In June 2018, the FASB issued Accounting Standards Update 2018-07, Compensation - Stock Compensation - Improvements to Nonemployee Share-Based Payment Accounting which both clarifies and modifies accounting requirements relating to nonemployee share based payment transactions. The Company believes this new standard will not have an impact on its consolidated financial statements.

Significant Accounting Policies

Revenue Recognition

The Company recognizes revenue when obligations under the terms of a contract with its customer are satisfied; generally this occurs with the transfer of control of the products sold. Transfer of control to the customer is based on the standardized shipping terms in the contract as this determines when the Company has the right to payment, the customer has legal title to the asset and the customer has the risks of ownership. Payment terms are defined in the contract and payment is typically due within three months after control has transferred to the customer. The contracts do not have a significant financing component.

The Company has elected to exclude value added, sales and other taxes it collects concurrent with revenue-producing activities from revenues.

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 1. The Company and Summary of Significant Accounting Policies (continued)

The Company may arrange shipping and handling activities as part of the sale of its products. The Company has elected to account for shipping and handling activities that occur after the customer has obtained control of the product as a fulfillment cost rather than as an additional promised service.

The following is a description of the principal activities from which the Company generates its revenues. For a breakdown of revenues by product and geographic location see the Business Segment Information Note.

Pulp and Lumber Revenues

For European sales sent by truck or train from the mills directly to the customer, the contracted sales terms are such that control transfers once the truck or train leaves the mill. For orders sent by ocean freighter, the contract terms state that control transfers at the time the product passes the ships rail. For North American sales shipped by truck or train, the contracts state that control transfers once the truck or train has arrived at the customer's specified location.

The transaction price is included in the sales contract and is net of customer discounts, rebates and other selling concessions.

The Company's pulp sales are to tissue and paper producers and the Company's lumber sales are to manufacturers and retailers. The Company's sales to Europe and North America are direct to the customer. The Company's pulp sales to overseas customers are primarily through third party sales agents and the Company's lumber sales to overseas customers are either direct to the customer or through third party sales agents.

By-Product Revenues

Energy sales are to utility companies in Canada and Germany. Sales of energy are recognized as the electricity is consumed by the customer and is based on contractual usage rates and meter readings that measure electricity consumption.

Chemicals and wood residuals are sold into the European market direct to the customer and have shipping terms where control transfers once the chemicals or wood residuals are loaded onto the truck at the mill.

Note 2. Inventories

	·	June 30, 2018	D	December 31, 2017
Raw materials	\$	77,844	\$	49,137
Finished goods		42,518		58,364
Spare parts and other		68,871		69,100
	\$	189,233	\$	176,601

Note 3. Accounts Payable and Other

	June 30, 2018	De	ecember 31, 2017
Trade payables	\$ 49,175	\$	36,151
Accrued expenses	90,733		67,528
Interest payable	16,505		10,093
Income tax payable	10,941		4,324
Legal cost award payable (Note 11(c))	6,951		_
Dividends payable	8,150		8,126
Other	5,077		7,335
	\$ 187,532	\$	133,557

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 4. Debt

	June 30, 2018	De	cember 31, 2017
2022 Senior Notes, principal amount, \$100,000 (a)	\$ 98,780	\$	394,565
2024 Senior Notes, principal amount, \$250,000 (a)	245,776		245,398
2026 Senior Notes, principal amount, \$300,000 (a)	294,203		293,773
Revolving credit facilities			
€75.0 million (b)	_		_
C\$40.0 million (c)	_		_
€70.0 million (d)	34,974		25,185
€5.0 million (e)	_		
€25.0 million (f)	25,528		_
	\$ 699,261	\$	958,921
As at June 30, 2018, the maturities of the principal portion of debt are as follows:			
2018		\$	_
2019			
2020			25,528
2021			
2022			134,974
Thereafter			550,000
		¢	710 502

Certain of the Company's debt instruments were issued under agreements which, among other things, may limit its ability and the ability of its subsidiaries to make certain payments, including dividends. These limitations are subject to specific exceptions. As at June 30, 2018, the Company is in compliance with the terms of its debt agreements.

(a) On December 20, 2017, the Company issued \$300,000 in aggregate principal amount of 5.50% senior notes which mature on January 15, 2026 ("2026 Senior Notes"). The 2026 Senior Notes were issued at a price of 100.00% of their principal amount. The net proceeds of the offering were \$293,795, after deducting the underwriter's discount and offering expenses.

In January 2018, the Company used the net proceeds, together with cash on hand, to redeem \$300,000 in aggregate principal amount of 2022 Senior Notes (herein defined below). In connection with this redemption the Company recorded a loss on settlement of debt of \$21,515 in the Interim Consolidated Statement of Operations. As at December 31, 2017, the total cash used to redeem the 2022 Senior Notes was classified as restricted cash and the carrying value of the 2022 Senior Notes was classified as a current liability in the Consolidated Balance Sheet.

On February 3, 2017, the Company issued \$225,000 in aggregate principal amount of 6.50% senior notes which mature on February 1, 2024 ("2024 Senior Notes") and on March 16, 2017, the Company issued an additional \$25,000 in aggregate principal amount of its 2024 Senior Notes. The 2024 Senior Notes were issued at a price of 100.00% of their principal amount. The net proceeds of the offerings were \$244,711, after deducting the underwriter's discount and offering expenses. The net proceeds from the 2024 Senior Notes, together with cash on hand, were used to redeem \$227,000 of remaining aggregate principal amount of outstanding senior notes due 2019, to finance the acquisition of a German sawmill and bio-mass power plant near Friesau Germany (the "Friesau Facility") and for general working capital purposes. In connection with the redemption the Company recorded a loss on settlement of debt of \$10,696 in the Interim Consolidated Statement of Operations.

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 4. Debt (continued)

On November 26, 2014, the Company issued \$400,000 in aggregate principal amount of 7.75% senior notes which mature on December 1, 2022 ("2022 Senior Notes" and collectively with the 2024 Senior Notes and 2026 Senior Notes, the "Senior Notes").

The Senior Notes are general unsecured senior obligations of the Company. They rank equal in right of payment with all existing and future unsecured senior indebtedness of the Company and are senior in right of payment to any current or future subordinated indebtedness of the Company. The Senior Notes are effectively junior in right of payment to all existing and future secured indebtedness, to the extent of the assets securing such indebtedness, and all indebtedness and liabilities of the Company's subsidiaries.

The Company may redeem all or a part of the 2026 Senior Notes, upon not less than 10 days' or more than 60 days' notice, at the redemption prices (expressed as percentages of principal amount) discussed below, plus accrued and unpaid interest to (but not including) the applicable redemption date. The Company may redeem all or a part of the 2024 Senior Notes or 2022 Senior Notes, upon not less than 30 days' or more than 60 days' notice, at the redemption prices (expressed as percentages of principal amount) discussed below, plus accrued and unpaid interest to (but not including) the applicable redemption date. The 2026 Senior Notes redemption prices are equal to 102.750% for the twelve month period beginning on January 15, 2021, 101.375% for the twelve month period beginning on January 15, 2023 and at any time thereafter. The 2024 Senior Notes redemption prices are equal to 103.250% for the twelve month period beginning on February 1, 2020, 101.625% for the twelve month period beginning on February 1, 2022 and at any time thereafter. The 2022 Senior Notes redemption prices are equal to 105.813% for the twelve month period beginning on December 1, 2017, 103.875% for the twelve month period beginning on December 1, 2018, 101.938% for the twelve month period beginning on December 1, 2020 and at any time thereafter.

- (b) A €75.0 million revolving credit facility at the Stendal mill that matures in October 2019. Borrowings under the facility are collateralized by the mill's inventory and accounts receivable and bear interest at Euribor plus 3.50%. As at June 30, 2018, approximately €2.2 million (\$2,592) of this facility was supporting bank guarantees leaving approximately €72.8 million (\$84,843) available.
- (c) A C\$40.0 million revolving credit facility at the Celgar mill that matures in May 2019. Borrowings under the facility are collateralized by the mill's inventory and accounts receivable and are restricted by a borrowing base calculated on the mill's inventory and accounts receivable. Canadian dollar denominated amounts bear interest at bankers acceptance plus 1.50% or Canadian prime. U.S. dollar denominated amounts bear interest at LIBOR plus 1.50% or U.S. base. As at June 30, 2018, approximately C\$1.7 million (\$1,290) was supporting letters of credit and approximately C\$38.3 million (\$29,086) was available.

In July 2018, the Company amended the credit facility including extending its maturity date to July 2023 and reducing the applicable margin on interest rates for Canadian and U.S. dollar denominated balances by 0.25% when the remaining borrowing capacity is greater than or equal to 25% of the total facility.

(d) A €70.0 million joint revolving credit facility that matures in April 2022. The Rosenthal mill has full access to the available amount under the facility and the Company's wholly owned subsidiary, Mercer Timber Products GmbH has access to a maximum of €45.0 million. Borrowings under the facility are collateralized by the borrowers' inventory and accounts receivable and bear interest at Euribor plus 2.95%. As at June 30, 2018, approximately €30.0 million (\$34,974) of this facility was drawn and accruing interest at a rate of 2.95% and approximately €11.6 million (\$13,478) of this facility was supporting bank guarantees leaving approximately €28.4 million (\$33,154) available.

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 4. Debt (continued)

- (e) A €5.0 million revolving credit facility at the Rosenthal mill that matures in December 2018. Borrowings under this facility bear interest at the rate of the three-month Euribor plus 2.50% and are secured by certain land at the Rosenthal mill. As at June 30, 2018 approximately €2.6 million (\$2,975) of this facility was supporting bank guarantees leaving approximately €2.4 million (\$2,854) available.
- (f) A €25.0 million revolving credit facility for the Company's wholly owned German subsidiary, Mercer Holz GmbH ("Mercer Holz"), that matures in February 2020. Borrowings under this facility bear interest at Euribor plus 3.30% and are secured by Mercer Holz's inventory and accounts receivable. As at June 30, 2018, approximately €21.9 million (\$25,528) of this facility was drawn and accruing interest at a rate of 3.30% and approximately €0.3 million (\$373) of this facility was supporting bank guarantees leaving approximately €2.8 million (\$3,244) available.

Note 5. Pension and Other Post-Retirement Benefit Obligations

Defined Benefit Plans

Included in pension and other post-retirement benefit obligations are amounts related to the Company's Celgar and Rosenthal mills. The largest component of these obligations is with respect to the Celgar mill which maintains a defined benefit pension plan and other post-retirement benefit plans for certain employees (the "Celgar Defined Benefit Plans").

Pension benefits are based on employees' earnings and years of service. The Celgar Defined Benefit Plans are funded by contributions from the Company based on actuarial estimates and statutory requirements.

The components of the net benefit costs relating to the Celgar Defined Benefit Plans for the three and six month periods ended June 30, 2018 and 2017 were as follows:

		Three Months Ended June 30,									
		20	018			20	17				
	P	Other Post- Retirement Pension Benefits				Pension	Ret	ner Post- irement enefits			
Service cost	\$	26	\$	117	\$	23	\$	141			
Interest cost		317		178		324		229			
Expected return on plan assets		(384)		_		(486)					
Amortization of unrecognized items		230		(52)		270		39			
Net benefit costs	\$	189	\$	243	\$	131	\$	409			

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 5. Pension and Other Post-Retirement Benefit Obligations (continued)

		Six Months Ended June 30,									
		2	018			20	17				
	Po	ension	Other Post- Retirement Ision Benefits			Pension	Re	her Post- tirement senefits			
Service cost	\$	52	\$	237	\$	46	\$	284			
Interest cost		640		360		652		461			
Expected return on plan assets		(777)		_		(979)		_			
Amortization of unrecognized items		464		(105)		527		75			
Net benefit costs	\$	379	\$	492	\$	246	\$	820			

Defined Contribution Plan

Effective December 31, 2008, the Celgar Defined Benefit Plans were closed to new members. In addition, the defined benefit service accrual ceased on December 31, 2008, and members began to receive pension benefits, at a fixed contractual rate, under a new defined contribution plan effective January 1, 2009. During the three and six month periods ended June 30, 2018, the Company made contributions of \$217 and \$435, respectively (2017 – \$192 and \$459), to this plan.

Multiemployer Plan

The Company participates in a multiemployer plan for the hourly-paid employees at the Celgar mill. The contributions to the plan are determined based on a percentage of pensionable earnings pursuant to a collective bargaining agreement. The Company has no current or future contribution obligations in excess of the contractual contributions. During the three and six month periods ended June 30, 2018, the Company made contributions of \$658 and \$1,145, respectively (2017 – \$568 and \$1,046), to this plan.

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 6. Income Taxes

The income tax provision attributable to income before provision for income taxes in the Interim Consolidated Statements of Operations differs from the amounts computed by applying the U.S. Federal statutory income tax rate of 21% (2017 - 35%) for the three and six month periods ended June 30, 2018 and 2017 as a result of the following:

	Three Months Ended June 30,					Six Months Ended June 30,			
		2018		2017		2018		2017	
U.S. Federal statutory rate		21%		35%		21%		35%	
	•	(7.500)	Φ.	(4.000)		(12.50.1)	•	(0.040)	
U.S. Federal statutory rate on income before provision for income taxes	\$	(5,296)	\$	(1,988)	\$	(12,694)	\$	(8,010)	
Tax differential on foreign income		(2,639)		550		(7,228)		2,967	
Effect of foreign earnings		(5,740)		_		(8,457)		_	
Change in undistributed earnings		_		(2,983)				(5,465)	
Valuation allowance		3,773		(5,200)		22,018		(9,354)	
Tax benefit of partnership structure		1,380		1,230		2,277		2,446	
Non-taxable foreign subsidies		732		550		1,488		1,109	
True-up of prior year taxes		(426)		(490)		(14,493)		(110)	
Foreign exchange on valuation allowance		(352)		926		(674)		1,163	
Foreign exchange on settlement of debt		_				_		550	
Other		107		(379)		(279)		(561)	
	\$	(8,461)	\$	(7,784)	\$	(18,042)	\$	(15,265)	
Comprised of:									
Current income tax provision	\$	(7,257)	\$	(3,588)	\$	(12,026)	\$	(6,860)	
Deferred income tax provision		(1,204)		(4,196)		(6,016)		(8,405)	
	\$	(8,461)	\$	(7,784)	\$	(18,042)	\$	(15,265)	

During the three month period ended June 30, 2018, the Company amended previously filed tax returns with the Canadian tax authorities. As a result, the Company reversed its unrecognized tax benefit of \$16,677 recorded during the three month period ended March 31, 2018 and increased its true-up of prior year taxes. The liability related to the amendment is fully offset by a reduction in the valuation allowance and the Company did not incur any interest or penalties.

The Act enacted on December 22, 2017 resulted in substantial changes including reducing the U.S. federal corporate income tax rate from 35% to 21% and requiring companies to pay a one-time transition tax on earnings of certain foreign subsidiaries that were previously tax deferred. The Company applied the guidance in Staff Accounting Bulletin No. 118 and at December 31, 2017 calculated its best estimate of the impact of the Act in its year end income tax provision. During the three and six month periods ended June 30, 2018, the Company did not record any measurement period adjustments to the provisional estimates recorded at December 31, 2017. Final accounting for these impacts is expected in the third quarter of 2018 subsequent to the Company's completion of the 2017 tax return.

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 7. Net Income (Loss) Per Common Share

	T	hree Months	Ended June 30,	Six Months E	Six Months Ended June 30,		
		2018	2017	2018	2017		
Net income (loss)							
Basic and diluted	\$	16,755	\$ (2,104)	\$ 42,404	\$ 7,622		
Net income (loss) per common share							
Basic and diluted	\$	0.26	\$ (0.03)	\$ 0.65	\$ 0.12		
Weighted average number of common shares outstanding:							
Basic ⁽¹⁾		65,140,802	64,945,675	65,095,788	64,857,300		
Effect of dilutive shares:							
Performance Share Units ("PSUs")		520,920	_	506,475	438,204		
Restricted shares		20,959	_	26,737	22,539		
Diluted		65,682,681	64,945,675	65,629,000	65,318,043		

⁽¹⁾ For the three and six month periods ended June 30, 2018, the basic weighted average number of common shares outstanding excludes 31,130 restricted shares which have been issued, but have not vested as at June 30, 2018 (2017 – 43,635 restricted shares).

The calculation of diluted net income (loss) per common share does not assume the exercise of any instruments that would have an anti-dilutive effect on net income (loss) per common share. The following table summarizes the instruments excluded from the calculation of net income (loss) per common share because they were anti-dilutive:

	Three Months E	nded June 30,	Six Months Ended June 30,		
	2018	2017	2018	2017	
PSUs		1,953,308			
Restricted shares		43,635	_	_	

Note 8. Shareholders' Equity

Dividends

During the six month period ended June 30, 2018, the Company's Board of Directors declared the following quarterly dividends:

		Di	vidend Per		
	Date Declared	Con	nmon Share	 Amount	
February 15, 2018		\$	0.125	\$ 8,147	
May 3, 2018			0.125	8,150	
		\$	0.250	\$ 16,297	

Dividends are paid in the quarter subsequent to the quarter in which they were declared.

In July 2018, the Company's Board of Directors declared a quarterly dividend of \$0.125 per common share. Payment of the dividend will be made on October 3, 2018 to all shareholders of record on September 26, 2018. Future dividends are subject to approval by the Board of Directors and may be adjusted as business and industry conditions warrant.

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 8. Shareholders' Equity (continued)

Stock Based Compensation

In June 2010, the Company adopted a stock incentive plan which provides for options, restricted stock rights, restricted shares, performance shares, PSUs and stock appreciation rights to be awarded to employees, consultants and non-employee directors. During the six month period ended June 30, 2018, there were no issued and outstanding options, restricted stock rights, performance shares or stock appreciation rights. As at June 30, 2018, after factoring in all allocated shares, there remain approximately 2.8 million common shares available for grant.

PSUs

PSUs comprise rights to receive common shares at a future date that are contingent on the Company and the grantee achieving certain performance objectives. The performance objective period is generally three years. For the three and six month periods ended June 30, 2018, the Company recognized an expense of \$1,630 and \$1,694, respectively related to PSUs (2017 – \$804 and \$555).

The following table summarizes PSU activity during the period:

	Number of PSUs
Outstanding as at January 1, 2018	1,867,158
Granted	652,548
Vested and issued	(153,243)
Forfeited	(330,455)
Outstanding as at June 30, 2018	2,036,008

Restricted Shares

Restricted shares generally vest at the end of one year. Expense recognized for the three and six month periods ended June 30, 2018 was \$129 and \$258 (2017 – \$108 and \$196). As at June 30, 2018, the total remaining unrecognized compensation cost related to restricted shares amounted to approximately \$477 which will be amortized over the remaining vesting periods.

The following table summarizes restricted share activity during the period:

	Number of
	Restricted
	Shares
Outstanding as at January 1, 2018	43,635
Granted	31,130
Vested	(43,635)
Outstanding as at June 30, 2018	31,130

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 8. Shareholders' Equity (continued)

Accumulated Other Comprehensive Loss

The components of accumulated other comprehensive loss are as follows:

		Foreign Turrency anslation Ljustment]	efined Benefit Pension and Other Post- Retirement Benefit Items	Unrealized Gains / Losses In Marketable Securities	Total
Balance as at January 1, 2018	\$	(50,083)	\$	(8,900)	\$ (18)	\$ (59,001)
Other comprehensive income (loss) before reclassifications		(39,522)		(1,087)	27	(40,582)
Amounts reclassified from accumulated other comprehensive loss				359	<u> </u>	359
Other comprehensive income (loss)		(39,522)		(728)	27	(40,223)
Balance as at June 30, 2018	\$	(89,605)	\$	(9,628)	\$ 9	\$ (99,224)

Note 9. Business Segment Information

The Company is managed based on the primary products it manufactures: pulp and wood products. Accordingly, the Company's three pulp mills are aggregated into the pulp business segment and the Friesau Facility from its acquisition date of April 12, 2017 is a separate reportable business segment, wood products.

None of the income or loss items following operating income in the Company's Interim Consolidated Statement of Operations are allocated to the segments, since those items are reviewed separately by management.

The following tables shows information by reportable business segments for the three and six month periods ended June 30, 2018 and 2017:

Three Months Ended June 30, 2018	Pulp	Wood Products	Corporate and Other	Consolidated
Revenues from external customers	\$ 291,632	\$ 54,900	\$ 	\$ 346,532
Operating income (loss)	\$ 36,976	\$ 4,322	\$ (3,822)	\$ 37,476
Depreciation and amortization	\$ 21,127	\$ 1,779	\$ 108	\$ 23,014
Revenues by major products				
Pulp	\$ 279,939	\$ _	\$ _	\$ 279,939
Lumber	_	48,991	_	48,991
Energy and chemicals	11,693	3,255	_	14,948
Wood residuals	_	2,654		2,654
Total revenues	\$ 291,632	\$ 54,900	\$ 	\$ 346,532
Revenues by geographical markets				
	\$ 5,653	\$ 15,249	\$ _	\$ 20,902
Germany	116,205	22,094	_	138,299
China	75,356	_	_	75,356
Other countries	94,418	17,557	_	111,975
Total revenues	\$ 291,632	\$ 54,900	\$ _	\$ 346,532

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 9. Business Segment Information (continued)

Three Months Ended June 30, 2017	Pulp			Wood roducts		orporate d Other	Consolidated		
Revenues from external customers	\$	265,886	\$	17,291	\$	_	\$	283,177	
Operating income (loss)	\$	21,069	\$	81	\$	(2,287)	\$	18,863	
Depreciation and amortization	\$	19,387	\$	1,134	\$	104	\$	20,625	
Revenues by major products									
Pulp	\$	244,684	\$	_	\$	_	\$	244,684	
Lumber		_		13,593		_		13,593	
Energy and chemicals		21,202		2,645		_		23,847	
Wood residuals				1,053				1,053	
Total revenues	\$	265,886	\$	17,291	\$	<u> </u>	\$	283,177	
Revenues by geographical markets									
U.S.	\$	6,694	\$	_	\$	_	\$	6,694	
Germany		103,438		11,636				115,074	
China		74,779		_		_		74,779	
Other countries		80,975		5,655		<u> </u>		86,630	
Total revenues	\$	265,886	\$	17,291	\$	<u> </u>	\$	283,177	
Six Months Ended June 30, 2018		Pulp		Wood Products		orporate nd Other	C	onsolidated	
Six Months Ended June 30, 2018 Revenues from external customers		Pulp 605,867	P	roducts	an				
Revenues from external customers	\$ \$	Pulp 605,867 111,030		Products 108,568		od Other	\$ \$	714,435	
· ·	\$ \$ \$ \$	605,867	\$ \$ \$	7,304	\$ \$		\$		
Revenues from external customers Operating income (loss)	\$	605,867 111,030	<u>P</u> \$	Products 108,568	an \$	(4,810)	\$	714,435 113,524	
Revenues from external customers Operating income (loss) Depreciation and amortization	\$ \$	605,867 111,030 42,650	\$ \$ \$	7,304 3,465	\$ \$ \$	(4,810) 218	\$ \$ \$	714,435 113,524 46,333	
Revenues from external customers Operating income (loss) Depreciation and amortization Total assets	\$ \$	605,867 111,030 42,650	\$ \$ \$	7,304 3,465	\$ \$ \$	(4,810) 218	\$ \$ \$	714,435 113,524 46,333	
Revenues from external customers Operating income (loss) Depreciation and amortization Total assets Revenues by major products Pulp Lumber	\$ \$ \$	605,867 111,030 42,650 1,351,996	\$ \$ \$ \$	7,304 3,465 131,092	\$ \$ \$ \$	(4,810) 218	\$ \$ \$ \$	714,435 113,524 46,333 1,522,389	
Revenues from external customers Operating income (loss) Depreciation and amortization Total assets Revenues by major products Pulp	\$ \$ \$	605,867 111,030 42,650 1,351,996	\$ \$ \$ \$	108,568 7,304 3,465 131,092	\$ \$ \$ \$	(4,810) 218	\$ \$ \$ \$	714,435 113,524 46,333 1,522,389 570,490	
Revenues from external customers Operating income (loss) Depreciation and amortization Total assets Revenues by major products Pulp Lumber	\$ \$ \$	605,867 111,030 42,650 1,351,996 570,490	\$ \$ \$ \$	108,568 7,304 3,465 131,092	\$ \$ \$ \$	(4,810) 218	\$ \$ \$ \$	714,435 113,524 46,333 1,522,389 570,490 97,159	
Revenues from external customers Operating income (loss) Depreciation and amortization Total assets Revenues by major products Pulp Lumber Energy and chemicals	\$ \$ \$	605,867 111,030 42,650 1,351,996 570,490	\$ \$ \$ \$	108,568 7,304 3,465 131,092 97,159 6,036	\$ \$ \$ \$	(4,810) 218	\$ \$ \$ \$	714,435 113,524 46,333 1,522,389 570,490 97,159 41,413	
Revenues from external customers Operating income (loss) Depreciation and amortization Total assets Revenues by major products Pulp Lumber Energy and chemicals Wood residuals	\$ \$ \$	605,867 111,030 42,650 1,351,996 570,490 — 35,377	\$ \$ \$ \$ \$	97,159 6,036 5,373	\$ \$ \$ \$ \$ \$ \$	(4,810) 218	\$ \$ \$ \$	714,435 113,524 46,333 1,522,389 570,490 97,159 41,413 5,373	
Revenues from external customers Operating income (loss) Depreciation and amortization Total assets Revenues by major products Pulp Lumber Energy and chemicals Wood residuals Total revenues	\$ \$ \$	605,867 111,030 42,650 1,351,996 570,490 — 35,377	\$ \$ \$ \$ \$	97,159 6,036 5,373	\$ \$ \$ \$ \$ \$ \$	(4,810) 218	\$ \$ \$ \$	714,435 113,524 46,333 1,522,389 570,490 97,159 41,413 5,373	
Revenues from external customers Operating income (loss) Depreciation and amortization Total assets Revenues by major products Pulp Lumber Energy and chemicals Wood residuals Total revenues Revenues by geographical markets U.S. Germany	\$ \$ \$ \$	605,867 111,030 42,650 1,351,996 570,490 — 35,377 — 605,867	\$ \$ \$ \$ \$ \$ \$ \$ \$	97,159 6,036 5,373 108,568	\$ \$ \$ \$ \$ \$ \$	(4,810) 218	\$ \$ \$ \$	714,435 113,524 46,333 1,522,389 570,490 97,159 41,413 5,373 714,435	
Revenues from external customers Operating income (loss) Depreciation and amortization Total assets Revenues by major products Pulp Lumber Energy and chemicals Wood residuals Total revenues Revenues by geographical markets U.S.	\$ \$ \$ \$	605,867 111,030 42,650 1,351,996 570,490 — 35,377 — 605,867	\$ \$ \$ \$ \$ \$ \$ \$ \$	97,159 6,036 5,373 108,568	\$ \$ \$ \$ \$ \$ \$	(4,810) 218	\$ \$ \$ \$	714,435 113,524 46,333 1,522,389 570,490 97,159 41,413 5,373 714,435	

605,867

108,568

714,435

FORM 10-Q QUARTERLY REPORT - PAGE 17

Total revenues

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 9. Business Segment Information (continued)

				Wood		orporate		
Six Months Ended June 30, 2017		Pulp	P	roducts	and Other		Co	nsolidated
Revenues from external customers	\$	508,670	\$	17,291	\$		\$	525,961
Operating income (loss)	\$	63,429	\$	81	\$	(3,301)	\$	60,209
Depreciation and amortization	\$	38,503	\$	1,134	\$	209	\$	39,846
Revenues by major products								
Pulp	\$	465,496	\$	_	\$	_	\$	465,496
Lumber		_		13,593		_		13,593
Energy and chemicals		43,174		2,645		_		45,819
Wood residuals				1,053		<u> </u>		1,053
Total revenues	\$	508,670	\$	17,291	\$	_	\$	525,961
Revenues by geographical markets		 -	<u> </u>				-	
U.S.	\$	11,118	\$	_	\$	_	\$	11,118
Germany		201,463		11,636		_		213,099
China		133,676		_		_		133,676
Other countries		162,413		5,655		_		168,068
Total revenues	\$	508,670	\$	17,291	\$	_	\$	525,961

Revenues between segments are accounted for at prices that approximate fair value. These include revenues from the sale of residual fiber from the wood products segment to the pulp segment for use in the pulp production process and from the sale of residual fuel from the pulp segment to the wood products segment for use in energy production. For the three and six month periods ended June 30, 2018, the pulp segment sold \$556 and \$910, respectively of residual fuel to the wood products segment (2017 - \$nil) and the wood products segment sold \$5,096 and \$10,045, respectively of residual fiber to the pulp segment (2017 - \$2,986 and \$2,986).

As at December 31, 2017, the Company had total assets of \$1,253,545 in the pulp segment, \$116,320 in the wood products segment and \$354,845 in corporate and other.

Note 10. Financial Instruments and Fair Value Measurement

Due to their short-term maturity, the carrying amounts of cash and cash equivalents, restricted cash, accounts receivable and accounts payable and other approximates their fair value.

The fair value of the senior notes classified as Level 2 was determined using quoted prices in a dealer market, or using recent market transactions.

The following tables present a summary of the Company's outstanding financial instruments and their estimated fair values under the fair value hierarchy:

	Fair value measurements as at June 30, 2018 using:									
Description		Level 1		Level 2		Level 3		Total		
Revolving credit facilities	\$		\$	60,502	\$		\$	60,502		
Senior notes		_		649,125		_		649,125		
	\$		\$	709,627	\$	_	\$	709,627		

(Unaudited)

(In thousands of U.S. dollars, except share and per share data)

Note 10. Financial Instruments and Fair Value Measurement (continued)

	Fair value measurements as at December 31, 2017 using:									
Description	Level 1	Level 2	Level 3	Total						
Revolving credit facilities	\$ —	\$ 25,185	\$ —	\$ 25,185						
Senior notes	_	989,125		989,125						
	\$ —	\$ 1,014,310	\$ <u> </u>	\$ 1,014,310						

Credit Risk

The Company's credit risk is primarily attributable to cash held in bank accounts and accounts receivable. The Company maintains cash balances in foreign financial institutions in excess of insured limits. The Company limits its credit exposure on cash held in bank accounts by periodically investing cash in excess of short-term operating requirements and debt obligations in low risk government bonds, or similar debt instruments. The Company's credit risk associated with the sale of pulp, lumber and other wood residuals is managed through setting credit limits, the purchase of credit insurance and for certain customers a letter of credit is received prior to shipping the product. Concentrations of credit risk on the sale of pulp, lumber and other wood residuals are with customers and agents based primarily in Germany, China and Italy.

The carrying amount of cash and cash equivalents of \$269,482 and accounts receivable of \$190,807 recorded in the Interim Consolidated Balance Sheet, net of any allowances for losses, represents the Company's maximum exposure to credit risk.

Note 11. Commitments and Contingencies

- (a) The Company is involved in legal actions and claims arising in the ordinary course of business. While the outcome of any legal actions and claims cannot be predicted with certainty, it is the opinion of management that the outcome of any such claims which are pending or threatened, either individually or on a combined basis, will not have a material adverse effect on the consolidated financial condition, results of operations or liquidity of the Company.
- (b) The Company is subject to regulations that require the handling and disposal of asbestos in a prescribed manner if a property undergoes a major renovation or demolition. Otherwise, the Company is not required to remove asbestos from its facilities. Generally asbestos is found on steam and condensate piping systems as well as certain cladding on buildings and in building insulation throughout older facilities. The Company's obligation for the proper removal and disposal of asbestos products from the Company's mills is a conditional asset retirement obligation. As a result of the longevity of the Company's mills, due in part to the maintenance procedures and the fact that the Company does not have plans for major changes that require the removal of asbestos, the timing of the asbestos removal is indeterminate. As a result, the Company is currently unable to reasonably estimate the fair value of its asbestos removal and disposal obligation. The Company will recognize a liability in the period in which sufficient information is available to reasonably estimate its fair value.
- (c) In March 2018, the Company announced it had received the decision of the tribunal in respect of its previously initiated claim in January 2012 against the Government of Canada under the North American Free Trade Agreement ("NAFTA"). The basis of the claim was that the Celgar mill had received discriminatory treatment regarding its ability to purchase and sell energy compared to other pulp mills and entities that generate and sell electricity within the Province of British Columbia. The tribunal ruled that there was no violation of NAFTA and as is customary in these matters, the tribunal awarded costs to the Government of Canada of approximately \$6,951.

NON-GAAP FINANCIAL MEASURES

This quarterly report on Form 10-Q contains "non-GAAP financial measures", that is, financial measures that either exclude or include amounts that are not excluded or included in the most directly comparable measure calculated and presented in accordance with the generally accepted accounting principles in the United States, referred to as "GAAP". Specifically, we make use of the non-GAAP measure "Operating EBITDA".

Operating EBITDA is defined as operating income (loss) plus depreciation and amortization and non-recurring capital asset impairment charges. We use Operating EBITDA as a benchmark measurement of our own operating results and as a benchmark relative to our competitors. We consider it to be a meaningful supplement to operating income as a performance measure primarily because depreciation expense and non-recurring capital asset impairment charges are not actual cash costs, and depreciation expense varies widely from company to company in a manner that we consider largely independent of the underlying cost efficiency of our operating facilities. In addition, we believe Operating EBITDA is commonly used by securities analysts, investors and other interested parties to evaluate our financial performance.

Operating EBITDA does not reflect the impact of a number of items that affect our net income (loss), including financing costs and the effect of derivative instruments. Operating EBITDA is not a measure of financial performance under GAAP, and should not be considered as an alternative to net income (loss) or income (loss) from operations as a measure of performance, or as an alternative to net cash from operating activities as a measure of liquidity. Operating EBITDA is an internal measure and therefore may not be comparable to other companies.

Operating EBITDA has significant limitations as an analytical tool, and should not be considered in isolation, or as a substitute for analysis of our results as reported under GAAP. Some of these limitations are that Operating EBITDA does not reflect: (i) our cash expenditures, or future requirements, for capital expenditures or contractual commitments; (ii) changes in, or cash requirements for, working capital needs; (iii) the significant interest expense, or the cash requirements necessary to service interest or principal payments, on our outstanding debt; (iv) the impact of realized or marked to market changes in our derivative positions, which can be substantial; and (v) the impact of impairment charges against our investments or assets. Because of these limitations, Operating EBITDA should only be considered as a supplemental performance measure and should not be considered as a measure of liquidity or cash available to us to invest in the growth of our business. Because all companies do not calculate Operating EBITDA in the same manner, Operating EBITDA as calculated by us may differ from Operating EBITDA or EBITDA as calculated by other companies. We compensate for these limitations by using Operating EBITDA as a supplemental measure of our performance and by relying primarily on our GAAP financial statements.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

In this document: (i) unless the context otherwise requires, references to "we", "our", "us", the "Company" or "Mercer" mean Mercer International Inc. and its subsidiaries; (ii) references to "Mercer Inc." mean the Company excluding its subsidiaries; (iii) information is provided as of June 30, 2018, unless otherwise stated; (iv) our reporting currency is dollars and references to "€" mean euros and "C\$" mean Canadian dollars; (v) "ADMTs" refers to air-dried metric tonnes; (vi) "MW" refers to megawatts and "MWh" refers to megawatt hours; (vii) "Mfbm" refers to thousand board feet of lumber and "MMfbm" mean million board feet of lumber; and (viii) our lumber metrics are converted from cubic meters to Mfbm using a conversion ratio of 1.6 cubic meters to 1 Mfbm, which is the ratio commonly used in the industry.

Due to rounding, numbers presented throughout this report may not add up precisely to totals we provide and percentages may not precisely reflect the absolute figure.

The following discussion and analysis of our results of operations and financial condition for the three and six months ended June 30, 2018 should be read in conjunction with our interim consolidated financial statements and related notes included in this quarterly report, as well as our most recent annual report on Form 10-K for the fiscal year ended December 31, 2017 filed with the Securities and Exchange Commission, referred to as the "SEC".

Results of Operations

General

We have two reportable operating segments, being the pulp business and, since April 2017, the wood products business.

Each segment offers primarily different products and requires different manufacturing processes, technology and sales and marketing.

Current Market Environment

In the second quarter of 2018, pulp prices in Europe and North America increased and pulp prices in China were generally flat compared to prior quarter of 2018 due to continued steady demand, producer maintenance downtime, low customer inventories and continued lower levels of recycled fiber availability in China due to import restrictions.

Overall, our average pulp sales realizations were approximately 5% higher in the second quarter of 2018 compared to the prior quarter of 2018.

At the end of the current quarter, list prices in Europe, China and North America were approximately \$1,230, \$910 and \$1,330 per ADMT, respectively.

Currently, the NBSK pulp market is generally balanced with world producer inventories at about 28 days' supply. Looking forward, we believe the new pulp production capacity that has or is coming online will not materially adversely impact the market in the near term as a result of continued steady demand growth, producer downtime and the continued restrictions on the import of recovered or waste paper in China.

In the third quarter of 2018, we have 14 days of scheduled maintenance downtime (which will reduce production by approximately 14,800 ADMTs) at our Rosenthal mill and in the fourth quarter of 2018, we have three days of scheduled maintenance downtime (which will reduce production by approximately 5,700 ADMTs) at our Stendal mill.

In the second quarter of 2018, European and U.S. lumber markets continued to be strong with prices near multiyear highs. Looking ahead into the third quarter, we expect lumber prices to moderately adjust from their multiyear highs.

Summary Financial Highlights

		Three Months Ended June 30,			Six Months Ended June 30,			
		2018 2		2017	2018			2017
			usar		an per share amounts)			
Pulp segment revenues	\$	291,632	\$	265,886	\$,	\$	508,670
Wood products segment revenues		54,900		17,291		108,568		17,291
Total revenues	\$	346,532	\$	283,177	\$	714,435	\$	525,961
Pulp segment operating income	\$	36,976	\$	21,069(1)	\$	111,030	\$	63,429(1)
Wood products segment operating income	4	4,322	4	81	4	7,304	Ψ.	81
Corporate and other operating loss		(3,822)		(2,287)		(4,810)		(3,301)
Total operating income	\$	37,476	\$	18,863	\$	113,524	\$	60,209
Pulp segment depreciation and amortization	\$	21,127	\$	19,387	\$	42,650	\$	38,503
Wood products segment depreciation and amortization		1,779		1,134		3,465		1,134
Corporate and other depreciation and amortization		108		104		218		209
Total depreciation and amortization	\$	23,014	\$	20,625	\$	46,333	\$	39,846
Operating EBITDA ⁽²⁾	\$	60,490	\$	39,488(1)	\$	159,857	\$	100,055(1)
Loss on settlement of debt	\$	00,490	\$	39,400(1)	\$	21,515(3)	\$	100,035(1)
Legal cost award	\$	=	\$	-	\$	6,951	\$	10,090(1)
Provision for income taxes	\$	8,461	\$	7,784	\$	18,042	\$	15,265
	\$		\$		\$		\$	
Net income (loss)	Ф	16,755	Ф	(2,104)	Ф	42,404	Ф	7,622
Net income (loss) per common share	\$	0.26	Ф	(0.02)	¢	0.65	Ф	0.12
Basic and diluted	Þ	0.26	\$	(0.03)	\$	0.65	\$	
Common shares outstanding at period end		65,202		65,017		65,202		65,017

⁽¹⁾ Adjusted as a result of our adoption of Accounting Standards Update 2017-07, Improving the Presentation of Net Periodic Pension Cost and Net Periodic Post-Retirement Benefit Cost, in the current year. See Note 1 to our Interim Consolidated Financial Statements.

⁽²⁾ The following table provides a reconciliation of net income (loss) to operating income and Operating EBITDA for the periods indicated:

	Three Months Ended June 30,				onths Ended une 30,		
	 2018		2017		2018		2017
			(in tho	usands			
Net income (loss)	\$ 16,755	\$	(2,104)	\$	42,404	\$	7,622
Provision for income taxes	8,461		7,784		18,042		15,265
Interest expense	12,128		13,320		24,243		27,199
Loss on settlement of debt	-		-		21,515		10,696
Legal cost award	-		-		6,951		-
Other (income) expenses	 132		(137)		369		(573)
Operating income	37,476		18,863		113,524		60,209
Add: Depreciation and amortization	 23,014		20,625		46,333		39,846
Operating EBITDA	\$ 60,490	\$	39,488	\$	159,857	\$	100,055

⁽³⁾ Redemption of 7.75% senior notes due 2022, referred to as the "2022 Senior Notes".
(4) Redemption of 7.00% senior notes due 2019, referred to as the "2019 Senior Notes".

Selected Production, Sales and Other Data

	Three Mont June	Six Month June		
Pulp Segment	2018	2017	2018	2017
Pulp production ('000 ADMTs)	309.7	362.7	674.2	736.4
Annual maintenance downtime ('000 ADMTs)	55.4	32.5	55.4	32.5
Annual maintenance downtime (days)	37	22	37	22
Pulp sales ('000 ADMTs)	338.3	388.8	705.4	763.9
Average NBSK pulp list prices in Europe (\$/ADMT) ⁽¹⁾	1,200	880	1,148	852
Average NBSK pulp list prices in China (\$/ADMT)(1)	910	670	910	658
Average NBSK pulp list prices in North America (\$/ADMT) ⁽¹⁾	1,310	1,093	1,272	1,063
Average pulp sales realizations (\$/ADMT) ⁽²⁾	821	624	801	604
Energy production ('000 MWh)	294.7	448.7	732.7	920.9
Energy sales ('000 MWh)	84.6	193.5	260.3	396.1
Average energy sales realizations (\$/MWh)	99	89	104	90

	Three Mon June		Six Montl June	
Wood Products Segment	2018	2017	2018	2017
Lumber production (MMfbm)	112.0	67.5	215.3	67.5
Lumber sales (MMfbm)	113.1	41.5	228.2	41.5
Average lumber sales realizations (\$/Mfbm)	433	328	426	328
Energy production ('000 MWh)	25.6	24.0	46.2	24.0
Energy sales ('000 MWh)	25.6	24.0	46.2	24.0
Average energy sales realizations (\$/MWh)	127	110	131	110
Average Spot Currency Exchange Rates				
\$/ € ⁽³⁾	1.1922	1.1008	1.2103	1.0838
\$ / C\$ ⁽³⁾	0.7750	0.7438	0.7826	0.7496

⁽¹⁾ Source: RISI pricing report.

Consolidated - Three Months Ended June 30, 2018 Compared to Three Months Ended June 30, 2017

Total revenues for the three months ended June 30, 2018 increased by approximately 22% to \$346.5 million from \$283.2 million in the same quarter of 2017 primarily due to a 32% increase in pulp sales realizations and \$37.6 million of higher revenues from our wood products segment.

Costs and expenses in the current quarter increased by approximately 17% to \$309.1 million from \$264.3 million in the second quarter of 2017 primarily due to costs associated with a 66% increase in production in our wood products segment and higher maintenance and per unit fiber costs partially offset by the impact of lower pulp sales volumes.

In the second quarter of 2018, operating depreciation and amortization increased to \$22.9 million from \$20.5 million in the same quarter of 2017 primarily due to the negative impact of a weaker dollar on our euro and Canadian dollar denominated depreciation expense.

⁽²⁾ Sales realizations after customer discounts, rebates and other selling concessions. Incorporates the effect of pulp price variations occurring between the order and shipment dates.

⁽³⁾ Average Federal Reserve Bank of New York Noon Buying Rates over the reporting period.

Selling, general and administrative expenses increased to \$15.0 million in the second quarter of 2018 from \$13.3 million in the same quarter of 2017 primarily due to the impact of a higher share price on our stock compensation expense and the negative impact of a weaker dollar.

In the second quarter of 2018, our operating income increased by approximately 98% to \$37.5 million from \$18.9 million in the same quarter of 2017 as higher pulp sales realizations more than offset lower energy and pulp sales volumes and higher maintenance and per unit fiber costs.

Interest expense in the current quarter decreased to \$12.1 million from \$13.3 million in the same quarter of 2017 primarily as a result of a lower interest rate on our outstanding senior notes.

During the second quarter of 2018, income tax expense increased to \$8.5 million from \$7.8 million in the same quarter of 2017 due to higher taxable income for our German mills.

For the second quarter of 2018, our net income increased to \$16.8 million, or \$0.26 per share, from a net loss of \$2.1 million, or \$0.03 per share, in the same quarter of 2017.

In the second quarter of 2018, Operating EBITDA increased by approximately 53% to \$60.5 million from \$39.5 million in the same quarter of 2017 as higher pulp sales realizations more than offset lower energy and pulp sales volumes and higher maintenance and per unit fiber costs.

Operating Results by Business Segment

None of the income or loss items following operating income in our Interim Consolidated Statement of Operations are allocated to our segments, since those items are reviewed separately by management.

Pulp Segment - Three Months Ended June 30, 2018 Compared to Three Months Ended June 30, 2017

Selected Financial Information

	Three Months Ended June 30,			
	 2018 2017			
	 (in thousands)			
Pulp revenues	\$ 279,939	\$	244,684	
Energy and chemical revenues	\$ 11,693	\$	21,202	
Depreciation and amortization	\$ 21,127	\$	19,387	
Operating income	\$ 36,976	\$	21,069	

Pulp revenues in the second quarter of 2018 increased by approximately 14% to \$279.9 million from \$244.7 million in the same quarter of 2017 due to higher sales realizations partially offset by lower sales volumes.

Energy and chemical revenues decreased by approximately 45% to \$11.7 million in the second quarter of 2018 from \$21.2 million in the same quarter of 2017 due to lower production as a result of the maintenance downtime. Additionally, one of the turbines at the Stendal mill was taken offline for a scheduled major maintenance in April and did not resume service until late July 2018.

Pulp production decreased by approximately 15% to 309,668 ADMTs in the current quarter from 362,665 ADMTs in the same quarter of 2017. In the current quarter of 2018, we had an aggregate of 37 days (approximately 55,400 ADMTs) of annual maintenance downtime, of which 25 days (approximately 36,300 ADMTs) was at our Celgar mill and 12 days (approximately 19,100 ADMTs) was at our Stendal mill. The 25 days of maintenance downtime at the Celgar mill included 11 unplanned days primarily to complete additional identified maintenance inside the recovery boiler. This contributed to a slower restart of the mill than planned. In the second quarter of 2017, we had an aggregate of 22 days (approximately 32,500 ADMTs) of annual maintenance downtime, of which 20 days (approximately 28,700 ADMTs) was at our Celgar mill and two days (approximately 3,800 ADMTs) was at our Stendal mill.

We estimate that annual maintenance downtime in the current quarter adversely impacted our operating income by approximately \$59.1 million, comprised of approximately \$36.6 million in direct out-of-pocket expenses and the balance in reduced production. Many of our competitors that report their financial results using International Financial Reporting Standards ("IFRS") capitalize their direct costs of maintenance downtime.

Pulp sales volumes decreased by approximately 13% to 338,308 ADMTs in the current quarter from 388,792 ADMTs in the same quarter of 2017 primarily due to lower production.

In the current quarter of 2018, list prices for NBSK pulp increased from the same quarter of 2017, largely as a result of overall steady demand. Average list prices for NBSK pulp in Europe were approximately \$1,200 per ADMT in the second quarter of 2018 compared to approximately \$880 per ADMT in the same quarter of 2017. Average list prices for NBSK pulp in China and North America were approximately \$910 per ADMT and \$1,310 per ADMT, respectively, in the current quarter compared to approximately \$670 per ADMT and \$1,093 per ADMT, respectively, in the same quarter of 2017. NBSK pulp prices are cyclical and are at or near record highs.

Average pulp sales realizations increased by approximately 32% to \$821 per ADMT in the second quarter of 2018 from approximately \$624 per ADMT in the same quarter of 2017 primarily due to higher list prices.

As a result of the effect of the dollar strengthening at the end of the current quarter against the euro and Canadian dollar on our dollar denominated cash and receivables held at our operations, we recorded a net overall positive impact of approximately \$1.5 million due to foreign exchange despite the negative impact of a weaker dollar on costs and expenses during the current quarter.

Costs and expenses for our pulp segment in the current quarter increased by approximately 4% to \$255.2 million from \$244.8 million in the second quarter of 2017 primarily due to higher maintenance and per unit fiber costs partially offset by the impact of lower sales volumes.

In the second quarter of 2018, pulp segment operating depreciation and amortization increased to \$21.1 million from \$19.4 million in the same quarter of 2017 primarily due the negative impact of a weaker dollar on our euro and Canadian dollar denominated depreciation expense.

On average, in the current quarter overall per unit fiber costs increased by approximately 19% from the same quarter of 2017 primarily as a result of the negative impact of a weaker dollar on our euro and Canadian dollar denominated fiber costs and strong demand. In the current quarter low producer inventories caused by unfavorable winter harvesting conditions and strong demand resulted in the higher prices. We currently expect a moderate decline in per unit fiber costs in the third quarter of 2018 as a result of improved harvesting conditions.

Transportation costs for our pulp segment decreased by approximately 12% to \$17.6 million in the current quarter from \$20.1 million in the same quarter of 2017 primarily due to lower sales volumes.

In the second quarter of 2018, pulp segment operating income increased by approximately 75% to \$37.0 million from \$21.1 million in the same quarter of 2017 as higher pulp sales realizations more than offset lower sales volumes, higher maintenance costs and higher per unit fiber costs.

Wood Products Segment - Three Months Ended June 30, 2018 Compared to Three Months Ended June 30, 2017

Selected Financial Information

		Three Months Ended June 30,			
		2018 20			
	<u></u>	(in thousands)			
Lumber revenues	\$	48,991	\$	13,593	
Energy revenues	\$	3,255	\$	2,645	
Wood residual revenues	\$	2,654	\$	1,053	
Depreciation and amortization	\$	1,779	\$	1,134	
Operating income	\$	4,322	\$	81	

We entered into the wood products business on April 12, 2017.

In the second quarter of 2018, lumber revenues increased to \$49.0 million from \$13.6 million, due to higher sales volume and higher sales realizations. In the current quarter approximately 22% of sales volumes were in the U.S. market and substantially all remaining sales were in Europe, which such sales are generally made in euros.

Energy and wood residual revenues increased to \$5.9 million in the second quarter of 2018 from \$3.7 million in the same quarter of 2017 primarily due to higher sales volumes and the positive impact of a weaker dollar on our euro denominated revenues.

Production increased to 112.0 MMfbm of lumber in the current quarter from 67.5 MMfbm in the same quarter of 2017.

Average lumber sales realizations increased by approximately 32% to \$433 per Mfbm in the second quarter of 2018 from approximately \$328 per Mfbm in the same quarter of 2017 primarily due to higher prices in Europe and increased sales to the U.S. where we realized higher sale prices.

Fiber costs are approximately 80% of our cash production costs. In the current quarter per unit fiber costs increased by approximately 17% from the same quarter of 2017 primarily as a result of the negative impact of a weaker dollar on our euro denominated fiber costs and strong demand. In the current quarter low producer inventories caused by unfavorable winter harvesting conditions and strong demand resulted in higher prices. We currently expect a moderate decline in per unit fiber costs in the third quarter of 2018 as a result of improved harvesting conditions.

In the second quarter of 2018, lumber segment operating depreciation and amortization increased to \$1.8 million from \$1.1 million in the same quarter of 2017 primarily due to the acquisition of production equipment.

Transportation costs for our wood products segment increased to \$5.9 million in the current quarter from \$0.8 million in the same quarter of 2017 primarily due to higher sales volumes and higher sales to the U.S.

In the second quarter of 2018, our wood products segment operating income increased to \$4.3 million from \$0.1 million in the same quarter of 2017 primarily due to higher lumber sales realizations partially offset by higher transportation and per unit fiber costs.

Consolidated - Six Months Ended June 30, 2018 Compared to Six Months Ended June 30, 2017

Total revenues for the first half of 2018 increased by approximately 36% to \$714.4 million from \$526.0 million in the first half of 2017 primarily due to a 33% increase in pulp sales realizations and the inclusion of an additional \$91.3 million of wood products segment revenues.

Costs and expenses in the first half of 2018 increased by approximately 29% to \$600.9 million from \$465.8 million in the first half of 2017 primarily due to additional costs and expenses from our wood products segment, the negative impact of a weaker dollar on our euro denominated costs and expenses and higher maintenance and per unit fiber costs partially offset by lower pulp sales volumes.

In the first half of 2018, operating depreciation and amortization increased to \$46.1 million from \$39.6 million in the same period of 2017 primarily due to the negative impact of a weaker dollar on our euro denominated depreciation expense and the inclusion of depreciation for the wood products segment for the full period.

Selling, general and administrative expenses increased to \$29.4 million in the first half of 2018 from \$23.0 million in the same period of 2017 primarily due to the inclusion of our wood products segment for the full period and the negative impact of a weaker dollar.

In the first half of 2018, operating income increased by approximately 89% to \$113.5 million from \$60.2 million in the same period of 2017 as higher pulp sales realizations more than offset lower energy and pulp sales volumes, the negative impact of a weaker dollar on our euro denominated costs and expenses and higher maintenance and per unit fiber costs.

In December 2017, we issued \$300.0 million of 5.50% senior notes due 2026, referred to as the "2026 Senior Notes" and, on January 5, 2018, we utilized the proceeds, together with cash on hand, to redeem \$300.0 million of our 7.75% senior notes due 2022 at a cost, including premium, of \$317.4 million and recorded a loss on such redemption of \$21.5 million (being \$0.33 per share).

Interest expense in the first half of 2018 decreased to \$24.2 million from \$27.2 million in the same period of 2017 primarily as a result of a lower interest rate on our outstanding senior notes.

In the first half of 2018, we recognized an expense of \$7.0 million, or \$0.11 per share, in connection with the legal cost award made by the tribunal in our claim against the Government of Canada under the North American Free Trade Agreement, referred to as "NAFTA".

During the first half of 2018, income tax expense increased to \$18.0 million from \$15.3 million in the same period of 2017 due to higher taxable income for our German mills.

For the first half of 2018, after giving effect to costs of \$28.5 million, or \$0.44 per basic and \$0.43 per diluted share, for the redemption of senior notes and the NAFTA legal cost award, our net income increased to \$42.4 million, or \$0.65 per share, from \$7.6 million, or \$0.12 per share, after giving effect to costs of \$10.7 million for the redemption of senior notes in the same period of 2017.

In the first half of 2018, Operating EBITDA increased by approximately 60% to \$159.9 million from \$100.1 million in the same period of 2017 as higher pulp sales realizations more than offset lower energy and pulp sales volumes, the negative impact of a weaker dollar relative to the euro and higher maintenance and per unit fiber costs.

Pulp Segment – Six Months Ended June 30, 2018 Compared to Six Months Ended June 30, 2017

Selected Financial Information

	Six Months Ended June 30,			
	 2018 201			
	 (in thousands)			
Pulp revenues	\$ 570,490	\$	465,496	
Energy and chemical revenues	\$ 35,377	\$	43,174	
Depreciation and amortization	\$ 42,650	\$	38,503	
Operating income	\$ 111,030	\$	63,429	

Pulp revenues in the first half of 2018 increased by approximately 23% to \$570.5 million from \$465.5 million in the same period of 2017 due to higher sales realizations partially offset by lower sales volumes.

Energy and chemical revenues decreased by approximately 18% to \$35.4 million in the first half of 2018 from \$43.2 million in the same period of 2017 due to lower production as a result of the maintenance downtime. Additionally, one of the turbines at the Stendal mill was taken offline for a scheduled major maintenance commencing in April and did not resume service until late July 2018.

Pulp production decreased by approximately 8% to 674,154 ADMTs in the first half of 2018 from 736,430 ADMTs in the same period of 2017. We had annual maintenance of 37 days (approximately 55,400 ADMTs) in the first half of 2018, compared to 22 days (approximately 32,500 ADMTs) in the first half of 2017.

We estimate that annual maintenance downtime in the first half of 2018 adversely impacted our operating income by approximately \$59.1 million, comprised of approximately \$36.6 million in direct out-of-pocket expenses and the balance in reduced production. Many of our competitors that report their financial results using IFRS capitalize their direct costs of maintenance downtime.

Pulp sales volumes decreased by approximately 8% to 705,382 ADMTs in the first half of 2018 compared to 763,896 ADMTs in the same period of 2017 primarily due to lower production.

In the first half of 2018, list prices for NBSK pulp increased from the same period of 2017, largely as a result of continued steady demand. Average list prices for NBSK pulp in Europe were approximately \$1,148 per ADMT in the first half of 2018, compared to approximately \$852 per ADMT in the same period of 2017. Average list prices for NBSK pulp in China and North America were approximately \$910 per ADMT and \$1,272 per ADMT, respectively, in the first half of 2018, compared to approximately \$658 per ADMT and \$1,063 per ADMT, respectively, in the first half of 2017.

Average pulp sales realizations increased by approximately 33% to \$801 per ADMT in the first half of 2018 from approximately \$604 per ADMT in the same period of 2017 primarily due to higher list prices.

In the first half of 2018, the dollar was weaker against the euro which increased the dollar cost of our euro denominated costs and expenses and contributed to a negative foreign exchange impact on operating income of approximately \$19.9 million when compared to the same period of the prior year.

Costs and expenses in the first half of 2018 increased by approximately 11% to \$495.7 million from \$445.2 million in the first half of 2017 primarily due to the negative impact of a weaker dollar on our euro denominated costs and expenses and higher maintenance and per unit fiber costs partially offset by lower sales volumes.

In the first half of 2018, depreciation and amortization increased to \$42.7 million from \$38.5 million in the same period of 2017 primarily due the negative impact of a weaker dollar on our euro denominated depreciation expense.

On average, in the first half of 2018 overall per unit fiber costs increased by approximately 24% from the same period of 2017 primarily as a result of the negative impact of a weaker dollar on our euro and Canadian dollar denominated fiber costs, increased demand from Scandinavian pulp mills in our European fiber procurement areas resulting from weather and lower harvesting activities. Harvesting activities in both Germany and British Columbia were impacted by short-term interruptions resulting from unseasonably wet winter conditions. Additionally, in British Columbia, sawmills focused harvesting activities on rebuilding low sawlog inventories resulting from the 2017 summer wildfires and unfavorable winter harvesting conditions, which resulted in lower pulp log availability. We expect a moderate decline in per unit fiber costs in the third quarter of 2018 as a result of improved harvesting conditions.

Transportation costs for our pulp segment were \$37.4 million in the first half of 2018, flat when compared to the same period of 2017 as the decrease due to lower sales volume was offset by the negative impact of a weaker dollar on our euro and Canadian dollar denominated transportation costs.

In the first half of 2018, pulp segment operating income increased by approximately 75% to \$111.0 million from \$63.4 million in the same period of 2017 primarily due to higher pulp sales realizations partially offset by lower sales volumes, higher per unit fiber costs, the negative impact of a weaker dollar relative to the euro and higher maintenance costs.

Wood Products Segment - Six Months Ended June 30, 2018 Compared to Six Months Ended June 30, 2017

Selected Financial Information

		Six Months Ended June 30,				
		2018 2017 (in thousands)				
Lumber revenues	\$	97,159	\$	13,593		
Energy revenues	\$	6,036	\$	2,645		
Wood residual revenues	\$	5,373	\$	1,053		
Depreciation and amortization	\$	3,465	\$	1,134		
Operating income	\$	7,304	\$	81		

We entered into the wood products business on April 12, 2017.

In the first half of 2018, lumber revenues were \$97.2 million, of which approximately 24% of sales volumes were in the U.S. market and substantially all remaining sales were in Europe, which such sales are made in euros. European and U.S. lumber markets were generally strong with prices steady and near multi-year highs during the first half of 2018.

In the first half of 2018, lumber production increased to 215.3 MMfbm from 67.5 MMfbm in the prior period of 2017.

In the first six months of 2018, lumber sales volumes increased to 228.2 MMfbm from 41.5 MMfbm in the same period of 2017.

In the first half of 2018, average lumber sales realizations increased by 30% to \$426 per Mfbm from \$328 per Mfbm in the same period of 2017.

In the first half of 2018, energy and other by-product revenues increased to \$11.4 million from \$3.7 million in the first six months of 2017.

Our fiber costs were approximately 80% of our cash production costs. Unseasonably wet winter weather conditions in Germany resulted in lower harvesting activities and high fiber costs in the first half of 2018 compared to the same period of 2017. As harvesting activities increase we expect modestly lower fiber costs in the third quarter of 2018.

In the first half of 2018, depreciation and amortization for our wood products segment was \$3.5 million compared to \$1.1 million in the first half of 2017.

In the first half of 2018, our wood products segment operating income increased to approximately \$7.3 million from \$0.1 million in the same period of 2017.

Liquidity and Capital Resources

Summary of Cash Flows

	 Six Months Ended June 30,			
	2018		2017	
	 (in thousands)			
Net cash from operating activities	\$ 162,122	\$	69,470	
Net cash used in investing activities	(45,092)		(89,862)	
Net cash from financing activities	18,453(1)		17,705	
Effect of exchange rate changes on cash, cash equivalents and restricted cash	 (9,300)		6,434	
Net increase in cash, cash equivalents and restricted cash	\$ 126,183	\$	3,747	

⁽¹⁾ Excludes cash of \$317.4 million used for the redemption of 2022 Senior Notes.

Cash Flows from Operating Activities. We operate in a cyclical industry and our operating cash flows vary accordingly. Our principal operating cash expenditures are for labor, fiber, chemicals and debt service. Working capital levels fluctuate throughout the year and are affected by maintenance downtime, changing sales patterns, seasonality and the timing of receivables and sales and the payment of payables and expenses.

Cash provided by operating activities was \$162.1 million in the six months ended June 30, 2018 and \$69.5 million in the comparative period of 2017. An increase in accounts payable and accrued expenses provided cash of \$54.9 million, including the legal cost award in our NAFTA claim, in the first half of 2018 compared to \$43.8 million in the same period of 2017. An overall increase in inventories used cash of \$19.0 million in the first half of 2018 compared to a decrease in inventories providing cash of \$4.1 million in the same period of 2017. A decrease in accounts receivable provided cash of \$8.3 million in the first half of 2018 compared to an increase in accounts receivable using cash of \$43.7 million in the same period of 2017.

Cash Flows from Investing Activities. Investing activities in the six months ended June 30, 2018 used cash of \$45.1 million primarily related to capital expenditures of \$44.8 million. In the first half of 2018, capital expenditures included spending on large maintenance projects, improvements to the digester performance at our Celgar mill, a project to reduce nitrogen in wastewater at our Stendal mill and the replacement of mobile equipment, planer mill upgrades and saw line optimization at our Friesau Facility. In the same period of 2017, investing activities used cash of \$89.9 million, primarily related to the acquisition of our Friesau Facility for \$61.6 million and capital expenditures of \$27.9 million. In the first half of 2017, capital expenditures primarily related to a rail car acceptance system for logs and additional land for raw material storage at our Rosenthal mill, a pre-bleach press system upgrade and large maintenance projects at our Celgar mill and various other smaller projects.

Cash Flows from Financing Activities. In the first half of 2018, financing activities used cash of \$299.0 million primarily in connection with the redemption of the 2022 Senior Notes, which used cash of \$317.4 million. In the first half of 2018, advances of \$37.7 million on our revolving credit facilities were used primarily to finance wood procurement activities. In the first half of 2018, we paid dividends of \$16.3 million and \$1.4 million of debt issuance costs related to the 2026 Senior Notes. In the same period of 2017, financing activities provided cash of \$17.7 million, including an aggregate of \$250.0 million from the issuance of our 6.50% senior notes due 2024, referred to as the "2024 Senior Notes", which was primarily used to redeem our 2019 Senior Notes at a cost of \$234.9 million. In the first half of 2017, debt issuance costs for the 2024 Senior Notes used cash of \$6.1 million and dividend payments used cash of \$14.9 million and a scheduled payment in respect of our interest rate derivative used cash of \$3.8 million. In the first half of 2017, we also drew \$26.5 million on our €70.0 million revolving credit facility to partially finance the acquisition of the Friesau Facility and the build in working capital associated with its ramp up of operations.

Balance Sheet Data

The following table is a summary of selected financial information as at the dates indicated:

	_	June 30, December 2018 2017 (in thousands)		
Financial Position				
Cash and cash equivalents	\$	269,482	\$	143,299(1)
Working capital	\$	473,092	\$	421,873
Total assets	\$	1,522,389	\$	1,407,271(1)
Long-term liabilities	\$	795,416	\$	743,578
Total equity	\$	538,502	\$	550,666

⁽¹⁾ Excludes restricted cash of \$317.4 million held to redeem \$300.0 million of 2022 Senior Notes on January 5, 2018.

As a result of the strengthening of the dollar versus the euro and Canadian dollar as at June 30, 2018, we recorded a net non-cash decrease in the carrying value of our net assets, consisting primarily of our fixed assets denominated in euros and Canadian dollars. This non-cash decrease of approximately \$39.5 million does not affect our net income (loss), Operating EBITDA or cash flows but is reflected in our other comprehensive income (loss) and as a decrease to our total equity.

Sources and Uses of Funds

Our principal sources of funds are cash flows from operations and cash and cash equivalents on hand. Our principal uses of funds consist of operating expenses, capital expenditures and semi-annual interest payments on our outstanding senior notes.

The following table sets out our total capital expenditures and interest expense for the periods indicated:

	Six Mon Jur	ths Ei ie 30,	nded
	2018		2017
	 (in the	usano	ds)
Capital expenditures	\$ 44,839	\$	27,907
Cash paid for interest expense ⁽¹⁾	\$ 15,696	\$	20,881
Interest expense ⁽²⁾	\$ 24,243	\$	27,199

⁽¹⁾ Amounts differ from interest expense which includes non-cash items. See supplemental disclosure of cash flow information from our Interim Consolidated Statement of Cash Flows included in this report.

⁽²⁾ Interest on our 2022 Senior Notes is paid semi-annually in June and December of each year. In January 2018, we redeemed \$300.0 million of our 2022 Senior Notes. Interest on our 2024 Senior Notes is paid semi-annually in February and August of each year and interest on our 2026 Senior Notes is paid semi-annually in January and July of each year, commencing July 2018.

In the first half of 2018, we expended \$16.3 million to pay quarterly dividends of \$0.125 per common share.

As at June 30, 2018, our cash and cash equivalents increased to \$269.5 million from \$143.3 million at the end of 2017. As at June 30, 2018, we had approximately \$153.2 million available under our revolving credit facilities.

Based upon the current level of operations and our current expectations for future periods in light of the current economic environment, and in particular, current and expected pulp and lumber pricing and foreign exchange rates, we believe that cash flow from operations and available cash, together with available borrowings under our revolving credit facilities, will be adequate to finance the capital requirements for our business including the payment of our quarterly dividend during the next 12 months.

As at June 30, 2018, we had no material commitments to acquire assets or operating businesses.

In the future we may make acquisitions of businesses or assets or commitments to additional capital projects. To achieve the long-term goals of expanding our assets and earnings, including through acquisitions, capital resources will be required. Depending on the size of a transaction, the capital resources that will be required can be substantial. The necessary resources will be generated from cash flow from operations, cash on hand, borrowing against our assets or the issuance of securities.

Debt Covenants

Certain of our long-term obligations contain various financial tests and covenants customary to these types of arrangements. See our annual report on Form 10-K for the fiscal year ended December 31, 2017.

As at June 30, 2018, we were in full compliance with all of the covenants of our indebtedness.

Off-Balance Sheet Arrangements

At June 30, 2018, we did not have any off-balance sheet arrangements (as defined in Item 303(a)(4)(ii) of Regulation S-K).

Contractual Obligations and Commitments

There were no material changes outside the ordinary course to any of our material contractual obligations during the six months ended June 30, 2018.

Foreign Currency

As a majority of our assets, liabilities and expenditures are held or denominated in euros or Canadian dollars, our consolidated financial results are subject to foreign currency exchange rate fluctuations.

We translate foreign denominated assets and liabilities into dollars at the rate of exchange on the balance sheet date. Equity accounts are translated using historical exchange rates. Unrealized gains or losses from these translations are recorded in other comprehensive income (loss) and do not affect our net earnings.

As a result of the strengthening of the dollar versus the euro and Canadian dollar as at June 30, 2018, we recorded a net non-cash decrease of \$39.5 million in the carrying value of our net assets, consisting primarily of our fixed assets denominated in euros and Canadian dollars. As a result, our accumulated other comprehensive loss increased to \$99.2 million.

Based upon the exchange rate as at June 30, 2018, the dollar has strengthened by approximately 3% against the euro and by approximately 5% against the Canadian dollar since December 31, 2017. See "Quantitative and Qualitative Disclosures about Market Risk".

Critical Accounting Policies

The preparation of financial statements and related disclosures in conformity with GAAP requires management to make estimates and assumptions that affect both the amount and the timing of the recording of assets, liabilities, revenues, and expenses in the consolidated financial statements and accompanying note disclosures. Our management routinely makes judgments and estimates about the effects of matters that are inherently uncertain. As the number of variables and assumptions affecting the probable future resolution of the uncertainties increases, these judgments become even more subjective and complex.

Our significant accounting policies are disclosed in Note 1 to our audited annual financial statements included in our annual report on Form 10-K for the fiscal year ended December 31, 2017. While all of the significant accounting policies are important to the consolidated financial statements, some of these policies may be viewed as having a high degree of judgment. On an ongoing basis using currently available information, management reviews its estimates, including those related to accounting for, among other things, pension and other post-retirement benefit obligations, deferred income taxes (valuation allowance and permanent reinvestment), depreciation and amortization, future cash flows associated with impairment testing for long-lived assets, the allocation of the purchase price in a business combination to the assets acquired and liabilities assumed, legal liabilities and contingencies. Actual results could differ materially from these estimates, and changes in these estimates are recorded when known.

We have identified certain accounting policies that are the most important to the portrayal of our current financial condition and results of operations.

For information about both our significant and critical accounting policies, see our annual report on Form 10-K for the fiscal year ended December 31, 2017.

Cautionary Statement Regarding Forward-Looking Information

The statements in this report that are not reported financial results or other historical information are "forward-looking statements" within the meaning of the *Private Securities Litigation Reform Act of 1995*, as amended.

Generally, forward-looking statements can be identified by the fact that they do not relate strictly to historical or current facts. They often include words such as "expects", "anticipates", "intends", "plans", "believes", "seeks", "estimates", or words of similar meaning, or future or conditional verbs, such as "will", "should", "could", or "may", although not all forward-looking statements contain these identifying words. Forward-looking statements are based on expectations, forecasts and assumptions by our management and involve a number of risks, uncertainties and other factors, many of which are beyond our control, that could cause actual conditions, events or results to differ significantly from those described in the forward-looking statements. These factors include, but are not limited to, the following:

- our business is highly cyclical in nature;
- a weakening of the global economy, including capital and credit markets, could adversely affect our business and financial results and have a material adverse effect on our liquidity and capital resources;
- our level of indebtedness could negatively impact our financial condition, results of operations and liquidity;
- cyclical fluctuations in the price and supply of our raw materials, particularly fiber, could adversely affect our business;
- we face intense competition in our markets;
- we are exposed to currency exchange rate fluctuations;
- we are subject to extensive environmental regulation and we could incur substantial costs as a result of compliance with, violations of or liabilities under applicable environmental laws and regulations;
- our business is subject to risks associated with climate change and social and government responses thereto:
- our operations require substantial capital and we may be unable to maintain adequate capital resources to provide for such capital requirements;
- our acquisition of the Friesau Facility and other future acquisitions may result in additional risks and uncertainties in our business:
- fluctuations in prices and demand for lumber could adversely affect our business;
- adverse housing market conditions may increase the credit risk from customers of our Friesau Facility;
- our Friesau Facility's lumber products are vulnerable to declines in demand due to competing technologies or materials;
- changes in credit ratings issued by nationally recognized statistical rating organizations could adversely affect our cost of financing and have an adverse effect on the market price of our securities;
- we rely on government grants and participate in German statutory energy programs;
- we are subject to risks related to our employees;
- we are dependent on key personnel;
- we may experience material disruptions to our production;

- if our long-lived assets become impaired, we may be required to record non-cash impairment charges that could have a material impact on our results of operations;
- we may incur losses as a result of unforeseen or catastrophic events, including the emergence of a pandemic, terrorist attacks or natural disasters;
- our insurance coverage may not be adequate;
- we rely on third parties for transportation services;
- we periodically use derivatives to manage certain risks which has caused significant fluctuations in our operating results;
- failures or security breaches of our information technology systems could disrupt our operations and negatively impact our business;
- the price of our common stock may be volatile;
- a small number of our shareholders could significantly influence our business;
- our international sales and operations are subject to applicable laws relating to trade, export controls and foreign corrupt practices, the violation of which could adversely affect our operations;
- we are exposed to interest rate fluctuations; and
- political uncertainty, the rise of populist political parties and an increase in trade protectionism could have a material adverse effect on global macro-economic activities and trade and adversely affect our business, results of operations and financial condition.

Given these uncertainties, you should not place undue reliance on our forward-looking statements. The forgoing review of important factors is not exhaustive or necessarily in order of importance and should be read in conjunction with the risks and assumptions including those set forth in reports and other documents we have filed with or furnished to the SEC, including in our annual report on Form 10-K for the fiscal year ended December 31, 2017. We advise you that these cautionary remarks expressly qualify in their entirety all forward-looking statements attributable to us or persons acting on our behalf. Unless required by law, we do not assume any obligation to update forward-looking statements based on unanticipated events or changed expectations. However, you should carefully review the reports and other documents we file from time to time with the SEC.

Cyclical Nature of Business

Revenues

The pulp and lumber businesses are highly cyclical in nature and markets are characterized by periods of supply and demand imbalance, which in turn can materially affect prices. Pulp and lumber markets are sensitive to cyclical changes in the global economy, industry capacity and foreign exchange rates, all of which can have a significant influence on selling prices and our operating results. The length and magnitude of industry cycles have varied over time but generally reflect changes in macro-economic conditions and levels of industry capacity. Pulp and lumber are commodities that are generally available from other producers. Because commodity products have few distinguishing qualities from producer to producer, competition is generally based upon price, which is generally determined by supply relative to demand.

Industry capacity can fluctuate as changing industry conditions can influence producers to idle production capacity or permanently close mills. In addition, to avoid substantial cash costs in idling or closing a mill, some producers will choose to operate at a loss, sometimes even a cash loss, which can prolong weak pricing environments due to oversupply. Oversupply of our products can also result from producers introducing new capacity in response to favorable pricing trends. Certain integrated pulp and paper producers have the ability to discontinue paper production by idling their paper machines and selling their NBSK pulp production on the market, if market conditions, prices and trends warrant such actions.

Demand for each of pulp and lumber has historically been determined primarily by general global macro-economic conditions and has been closely tied to overall business activity. NBSK pulp prices have been and are likely to continue to be volatile and can fluctuate widely over time. Between 2008 and 2018, European list prices for NBSK pulp have fluctuated between a low of approximately \$575 per ADMT in 2009 to a high of \$1,230 per ADMT in 2018.

Our mills and operations voluntarily subject themselves to third-party certification as to compliance with internationally recognized, sustainable management standards because end use paper and lumber customers have shown an increased interest in understanding the origin of products they purchase. Demand for our products could be adversely affected if we, or our suppliers, are unable to achieve compliance, or are perceived by the public as failing to comply, with these standards or if our customers require compliance with alternate standards for which our operations are not certified.

A producer's actual sales price realizations are list prices net of customer discounts, rebates and other selling concessions. Over the last three years, these have increased as producers compete for customers and sales.

Accordingly, prices for pulp and lumber are driven by many factors outside our control, and we have little influence over the timing and extent of price changes, which are often volatile. Because market conditions beyond our control determine the prices for pulp and lumber, prices may fall below our cash production costs, requiring us to either incur short-term losses on product sales or cease production at one or more of our mills. Therefore, our profitability depends on managing our cost structure, particularly raw materials which represent a significant component of our operating costs and can fluctuate based upon factors beyond our control. If the prices of our products decline, or if prices for our raw materials increase, or both, our results of operations and cash flows could be materially adversely affected.

Costs

Our production costs are influenced by the availability and cost of raw materials, energy and labor, and our plant efficiencies and productivity. Our main raw material is fiber in the form of wood chips, pulp logs and sawlogs. Wood chip, pulp log and sawlog costs are primarily affected by the supply of, and demand for, lumber and pulp, which are both highly cyclical. Higher fiber prices could affect producer profit margins if they are unable to pass along price increases to pulp and lumber customers or purchasers of surplus energy.

Currency

We have manufacturing operations in Germany and Canada. Most of the operating costs and expenses of our German mills are incurred in euros and those of our Celgar mill in Canadian dollars. However, the majority of our sales are in products quoted in dollars. Our results of operations and financial condition are reported in dollars. As a result, our costs generally benefit from a strengthening dollar but are adversely affected by a decrease in the value of the dollar relative to the euro and to the Canadian dollar. Such declines in the dollar relative to the euro and the Canadian dollar reduce our operating margins and the cash flow available to fund our operations and to service our debt. This could have a material adverse effect on our business, financial condition, results of operations and cash flows.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

We are exposed to market risks from changes in interest rates and foreign currency exchange rates, particularly the exchange rates between the dollar and the euro and Canadian dollar. Changes in these rates may affect our results of operations and financial condition and, consequently, our fair value. We seek to manage these risks through internal risk management policies as well as the periodic use of derivatives.

For additional information, please refer to Part II, Item 7A. Quantitative and Qualitative Disclosures about Market Risk included in our annual report on Form 10-K for the fiscal year ended December 31, 2017.

ITEM 4. CONTROLS AND PROCEDURES

Disclosure Controls and Procedures

Our management, with the participation of our principal executive officer and principal financial officer, has evaluated the effectiveness of our disclosure controls and procedures (as such term is defined in Rules 13a-15(e) and 15d-15(e) under the *Securities Exchange Act of 1934*, as amended, referred to as the "Exchange Act"), as of the end of the period covered by this report. Disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed in the reports we file or submit under the Exchange Act is accumulated and communicated to management, including our principal executive officer and principal financial officer, as appropriate, to allow timely decisions regarding required disclosure. Based on such evaluation, our principal executive officer and principal financial officer have concluded that, as of the end of the period covered by this report, our disclosure controls and procedures are effective in recording, processing, summarizing and reporting, on a timely basis, information required to be disclosed by us in the reports that we file or submit under the Exchange Act.

It should be noted that any system of controls is based in part upon certain assumptions designed to obtain reasonable (and not absolute) assurance as to its effectiveness and there can be no assurance that any design will succeed in achieving its stated goals.

Changes in Internal Controls

There have been no changes in our internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) during the period covered by this report that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

We are subject to routine litigation incidental to our business, including that which is described in our latest annual report on Form 10-K for the fiscal year ended December 31, 2017. We do not believe that the outcome of such litigation will have a material adverse effect on our business or financial condition.

ITEM 1A. RISK FACTORS

There have been no material changes to the factors disclosed in Item 1A. Risk Factors in our annual report on Form 10-K for the fiscal year ended December 31, 2017, other than the addition of the risk factor in item (i) below and the update in item (ii) below:

(i) Political uncertainty, the rise of populist political parties and an increase in trade protectionism could have a material adverse effect on global macro-economic activities and trade and adversely affect our business, results of operations and financial condition.

The current rise of populist political parties, economic nationalist sentiments and trade protectionism has led to increasing political uncertainty and unpredictability throughout the world. In 2016, the United Kingdom held a referendum at which the electorate voted to leave the Council of the European Union (the "EU"). It is unclear whether any other EU member states will hold such referendums, but such referendums could result in one or more other countries leaving the EU or in major reforms being made to the EU. The current U.S. presidential administration has publicly stated its intention to renegotiate or withdraw from the NAFTA and imposed tariffs on various goods from various countries, including China, Canada and those in the EU and announced intentions to impose further more significant tariffs. These potential developments, market perceptions concerning these and related issues and the attendant regulatory uncertainty regarding, for example, the posture of governments with respect to international trade, could have a material adverse effect on global trade and economic growth which, in turn can adversely affect our business, results of operation and financial condition.

The rise of populist political parties in some countries and the dominance of single-party political power in other countries may also lead to increased trade barriers, trade protectionism and restrictions on trade. Increased trade protectionism could materially adversely affect our business. If the current continuing global recovery is undermined by downside risks and there is a prolonged economic downturn, governments, especially populist governments, may turn to trade barriers to protect their domestic industries against imports, thereby depressing demand. Changes in U.S. trade policy, such as the announcement of unilateral tariffs on imported products, have already triggered retaliatory actions from affected countries, resulting in "trade wars" that could have a material adverse effect on global trade and economic growth.

Protectionist developments, or the perception they may occur, may have a material adverse effect on global economic conditions, and may significantly reduce global trade. Increasing trade protectionism in the markets could increase the risks associated with exporting goods to such markets. These developments could have a material adverse effect on our business, results of operations and financial condition.

(ii) We rely on government grants and participate in German statutory energy programs.

Our German mills, including our pulp mills, sell surplus green energy at fixed prices or "tariffs" pursuant to the Renewable Energy Act.

In 2014, in response to an investigation by the European Commission into whether portions of such Act constituted unpermitted state aid, the German government amended the same, which amendments permitted our German mills to continue to sell green energy into the market at stipulated prices or "tariffs" and were exempted, as "existing installations", from certain surcharges on the consumption of energy that they generate, or "auto-generation".

The German government further amended the Renewable Energy Act effective January 1, 2017, so that funding for renewable energy is to be allocated through an auction system, primarily to create a competitive bidding process for new installations of wind, solar and biomass energy. Our Friesau Facility's tariff expires in 2029. However, the amendments provide that existing pulp mills, including our German pulp mills, are ineligible for such auction process and instead will have their tariffs renewed upon expiry of their initial 20-year terms for a further 10-year period, based upon the price received in the last year prior to renewal regressing at a rate of 8% per annum. Our Rosenthal mill's initial 20-year tariff expires on December 31, 2019 and our Stendal mill's initial 20-year tariff expires on December 31, 2024. Such 10-year extensions for such pulp mills have been notified by the German government to the European Commission for review for compliance with applicable state aid rules. We have been advised by German governmental authorities that such extensions may not be permitted under EU rules. As a result, we cannot currently predict whether such promulgated amendments to the Renewable Energy Act will become effective. If they do not become effective, we cannot predict what further resulting amendments the German government may put into effect and their effect on our German mills' sale or consumption of energy after the expiry of their current terms of December 31, 2019 for Rosenthal and December 31, 2024 for Stendal.

Our costs of energy for our pulp operations in Germany could increase in the event that the auto-generation surcharge exemption is removed or reduced in the future. Additionally, if the stipulated tariffs for energy sold by our German mills are reduced in the future or sales are on an auction or market basis, we cannot provide assurances that our energy sales in Germany will be as profitable. Any of the foregoing situations or any combination of them could have a material adverse effect on our results of operations.

ITEM 2. UNREGISTERED SALE OF EQUITY SECURITIES AND USE OF PROCEEDS

None.

ITEM 3. DEFAULTS UPON SENIOR SECURITIES

None.

ITEM 4. MINE SAFETY DISCLOSURES

None.

ITEM 5. OTHER INFORMATION

None.

ITEM 6. EXHIBITS

Exhibit No.	Description
10.1	Third Amended and Restated Credit Agreement dated as of July 16, 2018 among Zellstoff Celgar Limited Partnership, as borrower, and the lenders from time to time parties thereto, as lenders, and Canadian Imperial Bank of Commerce, as agent.
31.1	Section 302 Certification of Chief Executive Officer
31.2	Section 302 Certification of Chief Financial Officer
32.1*	Section 906 Certification of Chief Executive Officer
32.2*	Section 906 Certification of Chief Financial Officer
101	The following financial statements from the Company's Form 10-Q for the fiscal period ended June 30, 2018, formatted in XBRL: (i) Interim Consolidated Statements of Operations; (ii) Interim Consolidated Statements of Comprehensive Income (Loss); (iii) Interim Consolidated Balance Sheets; (iv) Interim Consolidated Statements of Cash Flows; and (v) Notes to Interim Consolidated Financial Statements.

^{*} In accordance with Release No. 33-8212 of the SEC, these Certifications: (i) are "furnished" to the SEC and are not "filed" for the purposes of liability under the Securities Exchange Act of 1934, as amended; and (ii) are not to be subject to automatic incorporation by reference into any of the Company's registration statements filed under the Securities Act of 1933, as amended, for the purposes of liability thereunder or any offering memorandum, unless the Company specifically incorporates them by reference therein.

SIGNATURES

Pursuant to the requirements of the *Securities Exchange Act of 1934*, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

MERCER INTERNATIONAL INC.

By: /s/ David M. Gandossi

David M. Gandossi Chief Executive Officer and President

Date: July 26, 2018

EXHIBIT 10.1

EXECUTION VERSION

THIRD AMENDED AND RESTATED CREDIT AGREEMENT

dated as of

July 16, 2018

among

ZELLSTOFF CELGAR LIMITED PARTNERSHIP

as Borrower

and

THE LENDERS FROM TIME TO TIME PARTIES HERETO

as Lenders

and

CANADIAN IMPERIAL BANK OF COMMERCE

as Agent

	Pag
ARTICLE 1 DEFINITIONS	1
1.1 Defined Terms	
1.2 Classification of Loans and Borrowings	29
1.3 Terms Generally	29
1.4 Accounting Terms; GAAP	29
<u>1.5 Time</u>	30
1.6 Permitted Liens	30
1.7 Joint and Several	30
1.8 Effect of Amendment and Restatement of Existing Credit Agreement	30
1.9 Reaffirmation	30
ARTICLE 2 THE CREDITS	3:
2.1 Commitments	3:
2.2 Loans and Borrowings	3:
2.3 Requests for Borrowings	32
2.4 Funding of Borrowings	33
2.5 Interest	3:
2.6 Termination and Reduction of Commitments	3′
2.7 Repayment of Loans	3′
2.8 Evidence of Debt	3′
2.9 Prepayments	38
2.10 Fees	39
2.11 B/A Borrowings	40
2.12 Increased Costs; Illegality	42
2.13 Break Funding Payments	44
<u>2.14 Taxes</u>	44
2.15 Payments Generally; Pro Rata Treatment; Sharing of Set-offs	4:
2.16 Currency Indemnity	4
2.17 Collection of Accounts	47
2.18 Letters of Credit	49
ARTICLE 3 REPRESENTATIONS AND WARRANTIES	52
3.1 Organization; Powers	52
3.2 Authorization; Enforceability	52
3.3 Governmental Approvals; No Conflicts	52
3.4 Financial Condition; No Material Adverse Effect	52

(continued)

3.5 Litigation 3.6 Compliance with Applicable Laws and Agreements 3.7 Ownership 3.8 Taxes 3.9 Titles to Real Property 3.10 Titles to Personal Property 3.11 Pension Plans 3.12 Disclosure	53 53 53 53 53 54 54 55 55 55 55 56
3.7 Ownership 3.8 Taxes 3.9 Titles to Real Property 3.10 Titles to Personal Property 3.11 Pension Plans 3.12 Disclosure	53 53 53 54 54 55 55 55 55
3.8 Taxes 3.9 Titles to Real Property 3.10 Titles to Personal Property 3.11 Pension Plans 3.12 Disclosure	53 53 54 54 55 55 55 55
3.9 Titles to Real Property 3.10 Titles to Personal Property 3.11 Pension Plans 3.12 Disclosure	53 54 54 55 55 55 55
3.10 Titles to Personal Property 3.11 Pension Plans 3.12 Disclosure	54 54 55 55 55 55 56
3.11 Pension Plans 3.12 Disclosure	54 55 55 55 55 56
3.12 Disclosure	55 55 55 55 56
	55 55 55 56
	55 55 56
3.13 Defaults	55 56
3.14 Casualties; Taking of Properties	56
3.15 Subsidiaries	
3.16 Insurance	56
3.17 Solvency	30
3.18 Material Contracts	56
3.19 Environmental Matters	56
3.20 Employee Matters	57
3.21 Fiscal Year	58
3.22 Intellectual Property Rights	58
3.23 Residency of Borrower for Tax Purposes	58
3.24 Distributions	58
3.25 Debt	58
3.26 Workers' Compensation	58
3.27 Bank Accounts	59
3.28 Real Property and Leases	59
3.29 Further Real Property Matters	59
3.30 Jurisdictions of Credit Parties	59
ARTICLE 4 CONDITIONS	59
4.1 Effective Date	59
4.2 Each Credit Event	63
ARTICLE 5 AFFIRMATIVE COVENANTS	64
5.1 Financial Statements and Other Information	64
5.2 Existence; Conduct of Business	69
5.3 Payment of Obligations	70
5.4 Maintenance of Properties	70
5.5 Books and Records; Inspection Rights	70

(continued)

	Page
5.6 Compliance with Applicable Laws and Material Contracts	70
5.7 Use of Proceeds and Letters of Credit	70
5.8 Further Assurances	70
5.9 Insurance	71
5.10 Operation and Maintenance of Property	71
5.11 Additional Subsidiaries; Additional Liens	72
5.12 Financial Covenants	72
5.13 Post Closing Undertakings	72
5.14 Canadian Pension Plans	73
5.15 Application under the CCAA	73
ARTICLE 6 NEGATIVE COVENANTS	73
<u>6.1 Indebtedness</u>	73
6.2 Liens	74
6.3 Fundamental Changes	74
6.4 Investments, Loans, Advances, Guarantees and Acquisitions	75
6.5 Swap Agreements	75
6.6 Restricted Payments	75
6.7 Transactions with Affiliates	76
6.8 Repayment of Debt	77
6.9 Restrictive Agreements	77
6.10 Capital Lease Obligations; Sale/Leaseback Obligations	77
6.11 Pension Plan Compliance	77
6.12 Sale or Discount of Receivables	78
6.13 Unconditional Purchase Obligations	78
6.14 [Intentionally Deleted]	78
6.15 No Amendments to Material Contracts	78
ARTICLE 7 EVENTS OF DEFAULT	79
7.1 Events of Default	79
ARTICLE 8 THE AGENT	84
8.1 Appointment of Agent	84
8.2 Limitation of Duties of Agent	84
8.3 Lack of Reliance on the Agent	85
8.4 Certain Rights of the Agent	85
8.5 Reliance by Agent	85
8.6 Indemnification of Agent	85

(continued)

	Page
8.7 The Agent in its Individual Capacity	86
8.8 May Treat Lender as Owner	86
8.9 Successor Agent	86
8.10 No Independent Legal Action by Lenders	87
8.11 Quebec Security	87
ARTICLE 9 MISCELLANEOUS	88
9.1 Notices	88
9.2 Waivers; Amendments	89
9.3 Expenses; Indemnity; Damage Waiver	90
9.4 Successors and Assigns	92
9.5 Survival	95
9.6 Counterparts; Integration; Effectiveness	95
9.7 Severability	95
9.8 Right of Set-Off	95
9.9 Governing Law; Jurisdiction; Consent to Service of Process	96
9.10 WAIVER OF JURY TRIAL	96
9.11 Headings	96
9.12 Confidentiality	97
9.13 Press Releases and Related Materials	97
9.14 Anti-Money Laundering Legislation	98
9.15 No Strict Construction	98
9.16 Paramountcy	98
9.17 LIMITATION OF LIABILITY	98

THIRD AMENDED AND RESTATED CREDIT AGREEMENT

THIRD AMENDED AND RESTATED CREDIT AGREEMENT dated as of July 16, 2018, among ZELLSTOFF CELGAR LIMITED PARTNERSHIP, as Borrower, the Lenders from time to time parties hereto as lenders, and CANADIAN IMPERIAL BANK OF COMMERCE, as Issuing Bank, FX Bank and Agent.

RECITALS

A. Reference is made to the Credit Agreement dated as of May 19, 2006 (as amended and restated by the Amended and Restated Credit Agreement dated as of November 27, 2009, and as further amended and restated by the Second Amended and Restated Credit Agreement dated as of May 2, 2013, and as further amended from time to time, the "Existing Credit Agreement") among the Borrower, the Lenders and Canadian Imperial Bank of Commerce, as Administrative Agent.

- B. The Borrower has requested that the \$40,000,000 revolving credit facility established under the Existing Credit Agreement continue to be made available to the Borrower under this Agreement.
- C. The Lenders, the Administrative Agent, the Issuing Lender, the F/X Bank and the Borrower have agreed to amend and restate the terms of the Existing Credit Agreement upon the terms and subject to the conditions contained in this Agreement.

NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which is hereby conclusively acknowledged by each of the parties, the parties hereto agree that the Existing Credit Agreement is hereby amended and restated to read as follows:

ARTICLE 1 DEFINITIONS

1.1 <u>Defined Terms</u>. As used in this Agreement, the following terms have the meanings specified below:

"Acceptance Fee" means a fee payable by the Borrower to the Agent for the account of a Lender in Canadian Dollars with respect to the acceptance of a Bankers Acceptance or the making of a B/A Equivalent Loan, calculated on the face amount of the Bankers Acceptance or the B/A Equivalent Loan at a rate per annum equal to the Applicable Margin from time to time in effect on the basis of the actual number of days in the applicable Contract Period (including the date of acceptance and excluding the date of maturity) and a year of 365 days, (it being agreed that the Applicable Margin in respect of a B/A Equivalent Loan is equivalent to the Applicable Margin otherwise applicable to the B/A Borrowing which has been replaced by the making of such B/A Equivalent Loan pursuant to Section 2.11(h)).

"Accounts" means, in respect of each Credit Party, all of such Credit Party's now existing and future: (a) accounts (as defined in the PPSA), and any and all other receivables (whether or not specifically listed on schedules furnished to the Agent), including all accounts created by, or arising from, all of such Credit Party's sales, leases, loans, rentals of goods or renditions of services to its customers, including those accounts arising under any of such Credit Party's trade names or styles, or through any of such Credit Party's divisions; (b) any and all instruments, documents, chattel paper (including electronic chattel paper) (all as defined in the PPSA); (c) unpaid seller's or lessor's rights (including rescission, replevin, reclamation, repossession and stoppage in transit) relating to the foregoing or arising therefrom; (d) rights to any goods represented by any of the foregoing, including rights to returned, reclaimed or repossessed goods; (e) reserves and credit balances arising in connection with or pursuant hereto; (f) guarantees, indemnification rights, supporting obligations, payment intangibles, tax refunds and letter of credit rights; (g) insurance policies or rights relating to any of the foregoing; (h) intangibles pertaining to any and all of the foregoing (including all rights to payment, including those arising in connection with bank and non-bank credit cards), and including books and records and any electronic media and software relating thereto; (i) notes, deposits or property of borrowers or other account debtors securing the obligations of any such borrowers or other account debtors to such Credit Party; (j) cash and non cash proceeds (as defined in the PPSA) of any and all of the foregoing; and (k) all monies and claims for monies now or hereafter due and payable in connection with any and all of the foregoing or otherwise.

"Acquisition" means any transaction, or any series of related transactions, consummated after the Original Effective Date, by which any Credit Party, directly or indirectly, by means of a take-over bid, tender offer, amalgamation, merger, purchase of assets or otherwise (a) acquires any business or all or substantially all of the assets of any Person engaged in any business, (b) acquires control of securities of a Person engaged in a business representing more than 50% of the ordinary voting power for the election of directors or other governing position if the business affairs of such Person are managed by a board of directors or other governing body, (c) acquires control of more than 50% of the ownership interest in any Person engaged in any business that is not managed by a board of directors or other governing body, or (d) otherwise acquires Control of a Person engaged in a business.

"Adjusted Average Excess Availability" means, for each calendar month, an amount equal to (a) the aggregate sum of Excess Availability at the end of each day occurring during the immediately preceding calendar month, divided by (b) the number of days in the immediately preceding calendar month.

"Administrative Questionnaire" means an administrative questionnaire in a form supplied by the Agent.

"Affiliate" means, (a) any Person which, directly or indirectly, Controls, is Controlled by or is under common Control with any other Person; (b) any Person which beneficially owns or holds, directly or indirectly, 20% or more of any class of voting stock or equity interest (including partnership interests) of any other Person; (c) any Person, 20% or more of any class of the voting stock (or if such Person is not a corporation, 20% or more of the equity interest, including partnership interests) of which is beneficially owned or held, directly or indirectly, by any other Person; or (d) any Person related within the meaning of the ITA to any such Person and includes any "Affiliate" within the meaning specified in the *Canada Business Corporations Act* on the date hereof. The term control (including the terms "controlled by" and "under common control with"), means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of the Person in question.

"Agent" means Canadian Imperial Bank of Commerce, in its capacity as Agent for the Lenders hereunder, or any successor Agent appointed pursuant to Section 8.9.

"Agreement" means this Third Amended and Restated Credit Agreement, as it may be amended, modified, supplemented or restated from time to time.

"Anti-Corruption Laws" means all laws, rules, and regulations of any jurisdiction applicable to the Credit Parties and their Affiliates from time to time concerning or relating to bribery or corruption, including the Corruption of Foreign Public Officials Act (Canada).

"Applicable Law" means all federal, provincial, municipal, foreign and international statutes, acts, codes, ordinances, decrees, treaties, rules, regulations, municipal by-laws, judicial or arbitral or administrative or ministerial or departmental or regulatory judgments, orders, decisions, rulings or awards or any provisions of the foregoing, including general principles of common and civil law and equity, and all policies, practices and guidelines of any Governmental Authority binding on or affecting the Person referred to in the context in which such word is used (including, in the case of tax matters, any accepted practice or application or official interpretation of any relevant taxation authority).

"Applicable Margin" means, with respect to any Loan, the applicable rate per annum, expressed as a percentage, set forth in the relevant column of the table below:

	B/A Borrowing or LIBO	Canadian Prime Loan	
Adjusted Average	Rate Loan Applicable	or Base Rate Loan	Letter of Credit
Excess Availability	Margin	Applicable Margin	Exposure
Equal to or greater than 25%	1.25%	125%	1.00%
Less than 25%	1.50%	Nil	1.25%

The Applicable Margin shall be adjusted on the first day of each month based on Adjusted Average Excess Availability for the immediately preceding month.

"Applicable Percentage" means with respect to any Lender, the percentage of the total Commitments represented by such Lender's Commitment. If any Commitments have terminated or expired, the Applicable Percentages in respect of the terminated or expired Commitments shall be determined based upon the relevant Commitments most recently in effect (*i.e.*, prior to their termination or expiry), giving effect to any assignments.

"Assignment and Transfer" means an assignment and transfer entered into by a Lender and an assignee (with the consent of any party whose consent is required by Section 9.4), and accepted by the Agent, in the form of Exhibit E or any other form approved by the Agent.

"Availability Reserves" means, as of any date of determination, such amounts as the Agent may from time to time establish and revise in its reasonable business judgment (from the perspective of an asset-based lender) reducing the Borrowing Base which would otherwise be available to the Borrower under the lending formulas provided for herein (a) to reflect criteria, events, conditions, contingencies or risks which, as determined by the Agent, do or may affect either (i) any component of the Borrowing Base or its value, (ii) the assets, business, operations, industry, financial performance, financial condition of the Borrower, or (iii) the security interests and other rights of the Agent in the Collateral (including the enforceability, perfection and priority thereof), or (b) to reflect the Agent's customary lending practice or its reasonable belief that any collateral report or financial information furnished by or on behalf of the Borrower to the Agent is or may have been incomplete, inaccurate or misleading, or (c) in respect of any state of facts which the Agent determines constitutes a Default or an Event of Default. Without limiting the foregoing, the Agent, in its reasonable business judgment, may establish and/or increase Availability Reserves in respect of: (a) (i) three months rental payments or similar charges for any of the Borrower's leased premises or other collateral locations for which the Borrower has not delivered to the Agent a landlord's waiver or bailee's letter substantially in the form attached hereto as Exhibits C and D, respectively, plus (ii) three months estimated payments plus any other fees or charges owing by the Borrower to any applicable warehousemen or third party processor (as determined by the Agent in its reasonable business judgement), provided that any of the foregoing amounts shall be adjusted from time to time hereafter upon (x) delivery to the Agent of any such acceptable waiver, (y) the opening or closing of a collateral location and/or (z) any change in the amount of rental, storage or processor payments or similar charges; (b) any reserve established by the Agent on account of statutory claims, deemed trusts, or inventory subject to rights of suppliers under Section 81.1 of the BIA (generally known as the "30-day goods" rule), any amendments to the BIA to the extent such become Applicable Law and which have the effect of implementing any aspect of Bill C-55 as of the date hereof, or any other Applicable Law; (c) liabilities of any Credit Party under any Blocked Account Agreement, Swap Agreement or other swap, cap, floor, collar, futures contract or option designed to hedge against fluctuations in commodity prices, securities prices or other financial market conditions, (d) employee or employee benefit related liabilities, (e) any other claims which may have priority over the claims of the Agent and the Lenders, including Priority Payables; and (f) such other reserves as the Agent may at any time or times deem necessary in its reasonable business judgment as a result of (x) negative forecasts and/or trends in the Borrower's business, operations, industry, profits, operations or financial condition or assets or (y) other issues, circumstances or facts that could otherwise negatively impact the Borrower, its business, operations, industry, prospects, profits, operations or financial condition or assets.

"Authorization" means, with respect to any Person, any authorization, order, permit, approval, grant, licence, consent, right, franchise, privilege, certificate, judgment, writ, injunction, award, determination, direction, decree, by-law, rule or regulation of any Governmental Authority having jurisdiction over such Person, whether or not having the force of Law.

"Bank Act Security" has the meaning given to it in Section 4.1(k)(iv).

"B/A Borrowing" means a Borrowing comprised of one or more Bankers Acceptances or B/A Equivalent Loans. For greater certainty, unless the context requires otherwise, all provisions of this Agreement which are applicable to Bankers Acceptances are also applicable, *mutatis mutandis*, to B/A Equivalent Loans.

"B/A Equivalent Loan" has the meaning given to it in Section 2.11(h).

"B/A Rate" means, for any day and relative to a B/A Borrowing having any specified term, the simple average of the annual rates applicable to a Canadian Dollar bankers' acceptances displayed and identified as such on the display referred to as the "CDOR Page" (or any display substituted therefore) of Reuter Monitor Money Rates Service as at approximately 10:00 a.m. (Toronto time) on such day (or, if such day is not a Business Day, as of 10:00 a.m. on the immediate preceding Business Day), provided that if such rates do not appear on the CDOR Page at such time on such date, the rate for such date will be the annual discount rate (rounded upward to the nearest whole basis point) as of 10:00 a.m. on such day at which the Agent is then offering to purchase Canadian Dollar bankers' acceptances accepted by it having such specified term (or a term as closely as possible comparable to such specified term).

"Bankers Acceptance" and "B/A" mean an instrument denominated in Canadian Dollars, drawn by the Borrower and accepted by a Lender in accordance with this Agreement, and includes a "depository note" within the meaning of the *Depository Bills and Notes Act* (Canada) and a bill of exchange within the meaning of the *Bills of Exchange Act* (Canada).

"Base Rate" means, on any day, the annual rate of interest equal to the annual rate of interest announced from time to time by the Agent and in effect as its base rate at its principal office in Toronto, Ontario on such day for determining interest rates on U.S. Dollar-denominated commercial loans made in Canada. The Base Rate is a rate set by the Agent based upon various factors including the Agent's cost and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans; however, the Agent may price loans at, above or below such announced rate.

"Base Rate Borrowing" means a Borrowing comprised of one or more Base Rate Loans.

"Base Rate Loan" means a Loan denominated in U.S. Dollars made by the Lenders to the Borrower hereunder pursuant to a drawdown, rollover or conversion of a Loan on which interest is payable upon the Base Rate.

"Benefit Plans" means any plan, fund, program, or policy, whether oral or written, formal or informal, funded or unfunded, insured or uninsured, providing material employee benefits, including medical, hospital care, dental, sickness, accident, disability, life insurance, pension, retirement or savings benefits, under which any Credit Party has any liability with respect to any employee or former employee, but excluding any Pension Plans.

"BIA" means the Bankruptcy and Insolvency Act (Canada).

"Blocked Account Agreement" has the meaning set out in Section 2.17(c).

"Blocked Accounts" has the meaning set out in Section 2.17(c).

"Borrower" means Zellstoff Celgar Limited Partnership, a British Columbia limited partnership, its successors and permitted assigns.

"Borrowing" means any availment of the Credit, which includes a Loan and the issuance of a Letter of Credit Guarantee in accordance with Section 2.18, the entry into an F/X Contract in accordance with Section 2.19, and a Borrowing includes a rollover or conversion of any outstanding Loan and the provision of any Loan as required for the Agent to honour any obligations pursuant to any Letter of Credit Guarantee.

"Borrowing Base" means, at any time, an amount (which may not be less than zero) equal to the sum, without duplication, of (i) 85% of the aggregate amount of all Eligible Accounts (other than Insured Accounts and Investment Grade Accounts) and 90% of the aggregate amount of all Eligible Accounts which are Insured Accounts or Investment Grade Accounts, plus (ii) the lesser of (a) 75% of all Eligible Inventory consisting of finished goods (valued at the lower of cost (on a first in, first out basis and excluding any component of cost representing intercompany profit in the case of Inventory acquired from an Affiliate) or market basis in accordance with GAAP) and (b) 90% of the appraised net orderly liquidation value of all Eligible Inventory consisting of finished goods, plus (iii) the lesser of (a) 75% of all Eligible Inventory consisting of raw materials (valued at the lower of cost (on a first in, first out basis and excluding any component of cost representing intercompany profit in the case of Inventory acquired from an Affiliate) or market basis in accordance with GAAP) and (b) 90% of the appraised net orderly liquidation value of all Eligible Inventory consisting of raw materials, plus (iv) the lesser of (a) 70% of all Eligible Inventory consisting of work-in-progress (valued at the lower cost (on a first in, first out basis and excluding any component of cost representing intercompany profit in the case of Inventory acquired from an Affiliate) or market basis in accordance with GAAP), (b) 90% of the appraised net orderly liquidation value of all Eligible Inventory consisting of work-in-progress, and (c) \$300,000; minus (v) an amount equal to all Priority Payables, and minus (vi) an amount equal to all other Availability Reserves.

"Borrowing Base Report" means the report of the Borrower concerning the amount of the Borrowing Base, to be delivered pursuant to Section 5.1, substantially in the form attached as Exhibit A.

"Borrowing Request" means a request by the Borrower for a Borrowing substantially in the form of Exhibit B.

"Business Day" means any day that is not (i) a Saturday, Sunday or other day on which commercial banks in Toronto, Ontario are authorized or required by Applicable Law to remain closed, or (ii) in the case of any U.S. Dollar-denominated Borrowing, any other day on which commercial banks in New York, New York are authorized or required by Applicable Law to remain closed, or (iii) in the case of any LIBO Rate Loan any other day on which commercial banks in London, England are authorized or required by Applicable Law to remain closed.

"Canadian Dollars" and "Cdn.\$" refer to lawful money of Canada.

"Canadian \$ Equivalent" means, on any day, the amount of Canadian Dollars that the Agent could purchase, in accordance with its normal practice, with a specified amount of either U.S. Dollars or RMB based on the spot rate at which Canadian Dollars are offered at the start of such day by the Bank of China (Canada) in Vancouver, British Columbia.

"Canadian Prime Borrowing" means a Borrowing comprised of one or more Canadian Prime Loans.

"Canadian Prime Loan" means a Loan denominated in Canadian Dollars made by the Lenders to the Borrower hereunder pursuant to a drawdown, rollover or conversion of a Loan which bears interest at a rate based upon the Canadian Prime Rate.

"Canadian Prime Rate" means, the rate of interest publicly announced from time to time by the Agent as its reference rate of interest for loans made in Canadian Dollars to Canadian customers and designated as its "prime" rate. It is a rate set by the Agent based upon various factors including the Agent's costs and desired return, general economic conditions and other factors and is used as a reference point for pricing some loans. However, the Agent may price loans at, above or below such announced rate.

"Canadian Resident Lender" means, in respect of a particular Loan, (i) a Lender which holds such Loan and which is resident in Canada for the purposes of the ITA, and (ii) a Lender which is an "authorized foreign bank", as defined in section 2 of the *Bank Act* (Canada) and in section 248(1) of the ITA, and which holds the Loan as part of its "Canadian banking business", as defined in subsection 248(1) of the ITA.

"Capital Expenditures" means all payments due or accruing due (whether or not paid) during a Fiscal Year in respect of the cost (including expenditures on materials, contract labour and direct labour, but excluding expenditures properly chargeable to repairs and maintenance in accordance with GAAP) of any fixed asset or improvement, or replacement, substitution, or addition thereto, which have a useful life of more than one (1) year, including, without limitation, those arising in connection with the direct or indirect acquisition of such assets by way of increased product or service charges or offset items or in connection with Capital Leases.

"Capital Expenditure Plan" means the annual plan in respect of Capital Expenditures for the Borrower and each Restricted Subsidiary for a Fiscal Year of the Borrower. Each such plan shall be in comparative and consistent form to preceding Fiscal Years and Capital Expenditure Plans previously submitted to the Lenders and shall be comprised of (i) an itemized listing of Capital Expenditures made in the immediately preceding Fiscal Year, and (ii) proposed Capital Expenditures for such Fiscal Year. This program shall be reviewed by the Lenders and such other consultants as the Lenders may elect, such review being satisfactory to the Lenders.

"Capital Lease Obligations" of any Person means the obligations of such Person to pay rent or other amounts under any lease of (or other arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases on a balance sheet of such Person under GAAP, and the amount of such obligations shall be the capitalized amount thereof determined in accordance with GAAP.

"CDOR Rate" means, on any day and for any period, an annual rate of interest equal to the average rate applicable to Canadian Dollar bankers' acceptances for the applicable period appearing on the "Reuters Screen CDOR Page" (as defined in the International Swaps and Derivatives Association, Inc. 2000 definitions, as modified and amended from time to time), rounded to the nearest 1/100th of 1% (with .005% being rounded up), at approximately 10:00 a.m., Toronto time, on such day, or if such day is not a Business Day, then on the immediately preceding Business Day, provided that (i) if such rate does not appear on the Reuters Screen CDOR Page on such day as contemplated, then the CDOR Rate on such day shall be calculated as the rate for such period applicable to Canadian Dollar bankers' acceptances quoted by the Agent as of 10:00 a.m., Toronto time, on such day or, if such day is not a Business Day, then on the immediately preceding Business Day, and (ii) if such rate is less than zero, it shall be deemed to be zero.

"Change in Law" means (i) the adoption of any new Applicable Law after the date of this Agreement, (ii) any change in any existing Applicable Law or in the interpretation or application thereof by any Governmental Authority after the date of this Agreement, or (iii) compliance by any Lender (or, for purposes of Section 2.12(b), by any lending office of such Lender or by such Lender's holding company, if any) with any request, guideline or directive (whether or not having the force of law, but in the case of a request, guideline or directive not having the force of law, being a request, guideline or directive with which persons customarily comply) of any Governmental Authority made or issued after the date of this Agreement.

"CIBC" means Canadian Imperial Bank of Commerce and its successors.

"Collateral" means the property described in and subject to the Liens, privileges, priorities and security interests purported to be created by any Security Document.

"Commitment" means, with respect to each Lender, the commitment(s) of such Lender to make Loans hereunder as such commitment may be reduced from time to time pursuant to Sections 2.6 and/or 2.9, and as such commitments may be reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 9.4. The initial amount(s) of each Lender's Commitment(s) are set forth on Schedule A, or in the Assignment and Transfer pursuant to which such Lender shall have assumed its Commitment(s), as applicable. The initial aggregate amount of the Commitments is Cdn.\$40,000,000.

"Consolidated Net Income" means, for any period, the net income on a consolidated basis of the Borrower and its consolidated Subsidiaries; provided, however, that Consolidated Net Income shall not include or take into account:

- (i) any net income (or loss) of any Unrestricted Subsidiary, except that (subject to the exclusions contained in clauses (iii) and (iv) below), the Borrower's equity in the net income of any such Person for such period shall be included in such Consolidated Net Income up to the aggregate amount of cash actually distributed by such Person during such period to the Borrower or a Restricted Subsidiary as a dividend or other distribution (subject, in the case of a dividend or other distribution paid to a Restricted Subsidiary, to the limitations contained in clause (ii) below);
- (ii) any net income of any Restricted Subsidiary which is subject to restrictions, directly or indirectly, on the payment of dividends or the making of distributions, directly or indirectly, to the Borrower, except that (A) subject to the exclusion contained in clauses (iii) and (iv) below, the Borrower's equity in the net income of any such Restricted Subsidiary for such period shall be included in such Consolidated Net Income up to the aggregate amount of cash that could have been distributed by such Restricted Subsidiary consistent with such restriction during such period to the Borrower or another Restricted Subsidiary as a dividend or other distribution (subject, in the case of a dividend or other distribution paid to another Restricted Subsidiary, to the limitation contained in this clause), and (B) the Borrower's equity in a net loss of any such Restricted Subsidiary for such period shall be included in determining such Consolidated Net Income;
- (iii) any gain (or loss) realized upon the sale or other disposition of any assets of the Borrower or any Subsidiary (including pursuant to any sale-and-leaseback arrangement) which is not sold or otherwise disposed of in the ordinary course of business and any gain (or loss) realized upon the sale or other disposition of any capital stock of any Person;
- (iv) extraordinary or nonrecurring gains or non-cash losses; and
- (v) the effect of a change in GAAP.

"Contract Period" means the term of any B/A Borrowing selected by the Borrower in accordance with Section 2.3(a)(iv) commencing on the date of such B/A Borrowing and expiring on a Business Day which shall be either one month, two months, or three months thereafter (or such other terms as may be requested by the Borrower and approved unanimously by the Lenders); provided that (i) subject to subparagraph (ii) below, each such period shall be subject to such extensions or reductions as may be determined by the Agent to ensure that each Contract Period will expire on a Business Day, and (ii) no Contract Period shall extend beyond the Maturity Date.

"Control" means, in respect of a particular Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through the ability to exercise voting power, by contract or otherwise. "Controlling" and "Controlled" have meanings correlative thereto.

"Cover" means the aggregate amount of Bankers Acceptances and Letter of Credit Exposure and F/X Exposure at such time and such amount shall be paid by the Borrower to the Agent and retained by the Agent in a collateral account maintained by the Agent at its Payment Office and collaterally assigned to the Agent as security until such time as the applicable Bankers Acceptances, Letters of Credit and F/X Contracts shall have expired or matured and Reimbursement Obligations, if any, with respect thereto shall have been fully satisfied; provided that if any such Reimbursement Obligations are not satisfied when due hereunder, the Agent may apply any or all amounts in such collateral account in satisfaction of any or all such Reimbursement Obligations.

"Credit" means the Cdn.\$40,000,000 revolving credit facility established pursuant to the Commitments of the Lenders, as such revolving credit facility may be increased pursuant to Section 2.1 or reduced pursuant to Section 2.6.

"Credit Party" means the Borrower, the General Partner, each Restricted Subsidiary and any other Person which is a party to a Loan Document (other than the Agent, the Parent and the Lenders).

"DBRS" shall mean Dominion Bond Rating Service Limited, or its successor.

"Default" means any event or condition which constitutes an Event of Default or which, upon notice, lapse of time or both, would, unless cured or waived, become an Event of Default.

"Defined Benefits Plan" has the meaning given to it in Section 3.11.

"Defined Benefits Plan Liabilities" means any liabilities owed by the Borrower or any other Credit Party under any Defined Benefits Plan.

"Disclosed Matters" means the actions, suits and proceedings and the environmental matters disclosed in Schedule B.

"Discount Proceeds" means, for any Bankers Acceptance (or, as applicable, any B/A Equivalent Loan), an amount (rounded to the nearest whole cent, and with one-half of one cent being rounded up) calculated on the applicable date of Borrowing by multiplying:

- (i) the face amount of the Bankers Acceptance (or, as applicable, the undiscounted amount of the B/A Equivalent Loan); by
- (ii) the quotient of one divided by the sum of one plus the product of:
 - (a) the Discount Rate (expressed as a decimal) applicable to such Bankers Acceptance (or as applicable, such B/A Equivalent Loan), multiplied by

(b) a fraction, the numerator of which is the Contract Period of the Bankers Acceptance (or, as applicable, the B/A Equivalent Loan) and the denominator of which is 365,

with such quotient being rounded up or down to the nearest fifth decimal place, and with .000005 being rounded up.

"Discount Rate" means, with respect to either a Bankers Acceptance for a particular Contract Period being purchased by a Lender on any day or a B/A Equivalent Loan being made by a Lender on any day, (i) for any Lender which is a Schedule I chartered bank under the *Bank Act* (Canada), the CDOR Rate on such day for such Contract Period; and (ii) for any other Lender, the lesser of:

- (i) the CDOR Rate on such day for such Contract Period, plus 0.10%, and
- (ii) the percentage discount rate quoted by such Lender as the percentage discount rate at which such Lender would, in accordance with its normal practices, at or about 10:00 a.m. on such date, be prepared to purchase bankers' acceptances or make B/A Equivalent Loans having a face amount and term comparable to the face amount and term of such Bankers Acceptance or such B/A Equivalent Loan, as applicable.

"Early Termination Fee" means the fee due and payable by the Borrower to the Agent on behalf of the Lenders in the event of termination of this Agreement or reduction of the Commitment on a date prior to the Maturity Date by the Borrower or, upon the occurrence of an Event of Default, by the Agent, as determined by multiplying (a) \$40,000,000, in the case of termination of this Agreement or the Commitment; or (b) the amount by which the Commitment is reduced, in the case of the reduction of the Commitment, by (i) 0.50% if the Commitment is cancelled or reduced at any time on or prior to June 30, 2019, (ii) 0.25% if the Commitment is cancelled or reduced at any time following June 30, 2019 but on or prior to June 30, 2020, and (iii) 0.0% if the Commitment is cancelled or reduced at any time following June 30, 2020.

"EBITDA" means, for the Credit Parties on a consolidated basis and for any period, without duplication, an amount equal to the Consolidated Net Income for such period less any non-cash income included in Consolidated Net Income less non-cash write downs to Inventory, plus to the extent deducted from Consolidated Net Income, Interest Expense, depreciation, depletion and impairment, amortization expense and income tax expenses. For greater certainty, EBITDA for any period shall be determined after the payment of all management and employee bonuses and non-arm's length consulting fees for such period.

"Effective Date" means the date on which all of the conditions specified in Section 4.1 are satisfied or waived in accordance with Section 9.2.

- "Eligible Account" means, at any time, the invoice amount (which shall be the Canadian \$ Equivalent at such time of any amount denominated in U.S.\$ or RMB) owing on each Account of a Credit Party (net of any credit balance, returns, trade discounts, unapplied cash, unbilled amounts or retention or finance charges) which meet such standards of eligibility as the Agent shall establish from time to time in its reasonable discretion (from the perspective of an asset-based lender); provided that, in any event, no Account shall be deemed an Eligible Account unless each of the following statements is accurate and complete (and by including such Account in any computation of the applicable Borrowing Base, the Borrower shall be deemed to represent and warrant to the Agent, each Issuing Bank and the Lenders the accuracy and completeness of such statements and the compliance of each such Account with each such other eligibility standard established by the Agent):
 - (1) Such Account is a binding and valid obligation of the obligor thereon and is in full force and effect;
- (2) Such Account is evidenced by an invoice and is payable in Canadian Dollars, U.S. Dollars or RMB. The total aggregate amount of Eligible Accounts evidenced by an invoice and payable in RMB under this Agreement shall be no more than Cdn.\$15,000,000;
- (3) Such Account is genuine as appearing on its face or as represented in the books and records of the Borrower and the applicable Credit Party;
- (4) Such Account is free from claims regarding rescission, cancellation or avoidance, whether by operation of Applicable Law or otherwise;
- (5) Payment of such Account is less than 90 days past the original invoice date thereof and less than 60 days past the original due date thereof;
- (6) Such Account is net of concessions, offset, deduction, contras, returns, chargebacks or understandings with the obligor thereon that in any way could reasonably be expected to adversely affect the payment of, or the amount of, such Account;
- (7) The Agent on behalf of the Lenders, has a first-priority perfected Lien covering such Account and such Account is, and at all times will be, free and clear of all other Liens or claims (including any claim by the issuer of any performance bond, surety bond, appeal bond, completion guarantee or like instrument arising as a result of any failure of performance by a Credit Party);
 - (8) The obligor on such Account is not an Affiliate or a director, officer or employee of any Credit Party;
- (9) Such Account arose in the ordinary course of business of the Credit Party out of the sale of goods or services by the Credit Party;
- (10) Such Account is not payable by an obligor in respect of which 50% or more (by amount) of the total aggregate Accounts owed to the Credit Party by such obligor or any of its Affiliates are more than 90 days past the original invoice date thereof or more than 60 days past the original due date thereof;
- (11) All consents, licenses, approvals or authorizations of, or registrations or declarations with, any Governmental Authority required to be obtained, effected or given in connection with the execution, delivery and performance of such Account by each party obligated thereunder, or in connection with the enforcement and collection thereof by the Agent, have been duly obtained, effected or given and are in full force and effect;
- (12) The obligor on such Account is not an individual, and is not the subject of any bankruptcy or insolvency proceeding, does not have a trustee or receiver appointed for all or a substantial part of its property, has not made an assignment for the benefit of creditors, admitted its inability to pay its debts as they mature, suspended its business or initiated negotiations regarding a compromise of its debt with its creditors, and the Agent, in its reasonable discretion, is otherwise satisfied with the credit standing of such obligor;

- (13) The chief executive office of the obligor of such Account is located in the United States of America or Canada and the obligor of such Account is organized and existing under the laws of the United States of America or a state thereof or the federal laws of Canada, a province or territory thereof, or if the obligor is not so organized and existing, such Account is covered under letters of credit or export/import insurance provided by the Export Development Corporation, AIG Global Trade & Political Risk, or Euler Hermes (including their respective successors or permitted assigns) (or such other credit insurance as acceptable to the Agent and Lenders) on terms and in a manner reasonably satisfactory to the Agent;
- (14) The obligor of such Account is not a Governmental Authority, if the enforceability or effectiveness against such Governmental Authority of an assignment of such Account is subject to any precondition which has not been fulfilled;
- (15) In the case of the sale of goods, the subject goods have been completed, sold and shipped, on a true sale basis on open account, or subject to contract, and not on consignment, on approval, on a "sale or return" basis, or on a "bill and hold" or "pre-sale" basis or subject to any other repurchase or return agreement; no material part of the subject goods has been returned, rejected, lost or damaged; and such Account is not evidenced by chattel paper or a promissory note or an instrument of any kind, unless such chattel paper, promissory note or other instrument has been delivered to the Agent and is subject to a Lien under the Security Documents;
- (16) Each of the representations and warranties set forth herein and in the Loan Documents with respect to such Account is true and correct on such date;
- (17) A cheque, promissory note, draft, trade acceptance or other instrument has not been received with respect to such Account (or with respect to any other account due from the same account debtor), presented for payment and returned uncollected for any reason;
 - (18) Such Account is not in respect of a volume rebate;
- (19) The Agent does not believe, in the exercise of its reasonable discretion, that the prospect of collection of such Account is impaired or that the Account may not be paid because of the account debtor's inability to pay;
 - (20) Such Account is not a pre-billed account or an account arising from progress billing;
- (21) The assignment (whether absolutely or by way of security) of such Account is not limited or restricted by the terms of the contract evidencing or relating to such Account or, if assignment of such Account is so restricted, such limitation or restriction has been complied with and the laws of the jurisdiction(s) governing the validity of such assignment do not provide that such limitation or restriction is ineffective as against the secured creditor with a security interest therein; and
- (22) Such Account is not an Account which the Agent, in the exercise of its good faith credit discretion, determines to be ineligible for any other reasons deemed necessary by Agent in its reasonable business judgment, including those which are customary either in the commercial lending industry or in the lending practices of the Agent.

"Eligible Assignee" means (a) another Lender, (b) with respect to any Lender, any Affiliate of that Lender, (c) any commercial bank having total assets of \$25,000,000,000 or more, (d) any (i) trust company, savings bank, savings and loan association or similar financial institution, or (ii) insurance company engaged in the business of writing insurance which, in either case (A) has total assets of \$25,000,000,000 or more, (B) is engaged in the business of lending money and extending credit under credit facilities substantially similar to those extended under this Agreement, (C) is operationally and procedurally able to meet the obligations of a Lender hereunder to the same degree as a commercial bank, and (e) any other financial institution (including a mutual fund or other fund) having total assets of \$25,000,000,000,000 or more which meets the requirements set forth in subclauses (B) and (C) of clause (d) above.

"Eligible Inventory" means, at any time with respect to a Credit Party, all raw materials, finished goods and pulp log inventory of such Credit Party valued in Canadian Dollars on a lower of cost (on a first-in, first out basis and excluding any component of cost representing intercompany profit in the case of Inventory acquired from an Affiliate) or market basis in accordance with GAAP, with detailed calculations of lower of cost or market to occur on at least a monthly basis, which meets such standards of eligibility as the Agent shall establish from time to time in its reasonable discretion (from the perspective of an asset-based lender); provided that, in any event, no Inventory shall be deemed Eligible Inventory unless each of the following statements is accurate and complete (and by including such Inventory in any computation of the applicable Borrowing Base, the Borrower shall be deemed to represent and warrant to the Agent, each Issuing Bank and the Lenders the accuracy and completeness of such statements and the compliance of such Inventory with each such other eligibility standard established by the Agent):

(1) Such Inventory is in good condition, merchantable, meets all standards imposed by any Governmental Authority having regulatory authority over it or its use and/or sale and is not obsolete and is either currently usable or currently saleable in the normal course of business of a Credit Party;

(2) Such Inventory is

- (a) in possession of such Credit Party and located on real property owned or leased by such Credit Party within the United States of America or Canada (provided that if such Inventory is located on real property leased by such Credit Party, the landlord of such real property shall have executed and delivered to the Agent a landlord waiver substantially in the form attached hereto as Exhibit C), or
- (b) in the possession of a bailee within Canada and such bailee shall have executed and delivered to the Agent, a bailee letter substantially in the form attached hereto as Exhibit D, or
- (c) in transit within the United States of America or Canada (provided that the jurisdictions through which such Inventory is in transit are jurisdictions where the Liens in such inventory under the Security Documents are validly perfected first-priority Liens, subject to Permitted Liens) and between Credit Parties, and upon arrival at its destination, will comply with either paragraph (a) or (b) above until title to such Inventory passes to purchaser;

- (3) Each of the representations and warranties set forth in the Loan Documents with respect to such Inventory is true and correct on such date;
- (4) The Agent on behalf of the Lenders, has a first-priority perfected Lien covering such Inventory, and such Inventory is, and at all times will be, free and clear of all Liens other than Permitted Liens;
- (5) Such Inventory does not include goods (i) that are not owned by such Credit Party, (ii) that are held by such Credit Party pursuant to a consignment agreement, (iii) which have been sold by such Credit Party on a bill and hold basis, or (iv) that are special order goods or discontinued goods;
- (6) Such Inventory is not subject to repossession under the BIA except to the extent the applicable vendor has entered into an agreement with the Agent, in form and substance reasonably satisfactory to the Agent, waiving its right to repossession;
- (7) Such Inventory does not consist of store room materials, supplies, parts, samples, prototypes, or packing and shipping materials;
- (8) Such Inventory does not consist of goods that are discontinued, obsolete, slow-moving or returned or repossessed or used goods taken in trade;
- (9) Any portion of the value of such Inventory which results from a profit or gain resulting from an inter-company sale or other disposition of such inventory shall be excluded;
 - (10) Such Inventory is not evidenced by negotiable documents of title unless delivered to the Agent with endorsements;
 - (11) Such Inventory does not constitute Hazardous Materials;
 - (12) Such Inventory is covered by casualty insurance;
- (13) Such Inventory is not Inventory which the Agent has determined in the exercise of its reasonable discretion that the Agent may not sell or otherwise dispose of in accordance with the terms of the applicable Security Documents without infringing upon the rights of another Person or violating any contract with any other Person;
- (14) Such Inventory is not Inventory which the Agent, in the exercise of its good faith credit discretion, determines to be not acceptable for any other reasons deemed necessary by Agent in its reasonable business judgment, including those which are customary either in the commercial lending industry or in the lending practices of the Agent; and
- (15) Such Inventory is located on real property where there is Inventory of such Credit Party in the aggregate amount of at least Cdn.\$100,000.

"Environmental Laws" means all federal, provincial, local or foreign laws, rules, regulations, codes, ordinances, orders, decrees, judgements, injunctions, notices or binding agreements issued, promulgated or entered into by any Governmental Authority, relating in any way to the environment, preservation or reclamation of natural resources, the generation, use, handling, collection, treatment, storage, transportation, recovery, recycling, release, threatened release or disposal of any Hazardous Material, or to health and safety matters.

"Environmental Liability" means any liability, contingent or otherwise (including any liability for damages, costs of environmental remediation, fines, penalties or indemnities), of any Credit Party directly or indirectly resulting from or based upon (a) violation of any Environmental Laws, (b) the generation, use, handling, collection, treatment, storage, transportation, recovery, recycling or disposal of any Hazardous Materials, (c) exposure to any Hazardous Materials, (d) the Release or threatened Release of any Hazardous Materials into the environment, or (e) any contract, agreement or other consensual arrangement pursuant to which liability is assumed or imposed with respect to any of the foregoing.

"Equity Securities" means, with respect to any Person, any and all shares, interests, participations, rights in, or other equivalents (however designated and whether voting and non-voting) of, such Person's capital, whether outstanding on the date hereof or issued after the date hereof, including any interest in a partnership, limited partnership or other similar Person and any beneficial interest in a trust, and any and all rights, warrants, debt securities, options or other rights exchangeable for or convertible into any of the foregoing.

"ETA" means Part IX of the Excise Tax Act (Canada).

"Event of Default" has the meaning set out in Section 7.1.

"Excess Availability" means, as of any date, the remainder of (a) the Borrowing Base as of such date, less, without duplication, (b) the aggregate Exposure as of such date. Excess Availability and Adjusted Average Excess Availability shall always be determined on the basis that all debts and obligations shall be current, and all accounts payable shall be handled in the normal course of the Borrower's business consistent with its past practices.

"Excluded Taxes" means, with respect to the Agent, any Lender or any other recipient of any payment to be made by or on account of any obligation of the Borrower hereunder, income or franchise Taxes imposed on (or measured by) its taxable income, in each case by Canada, or by the jurisdiction under the Applicable Laws of which such recipient is organized or in which its principal office is located.

"Exposure" means, with respect to any Lender at any time, the sum of the outstanding principal amount of such Lender's Revolving Loans and, without duplication, its LC Exposure and F/X Exposure at such time.

"Fee Letter" means the letter agreement dated as of the date hereof between the Borrower and the Agent relating to the payment of certain fees.

"Financial Officer" means the chief financial officer, principal accounting officer, treasurer, controller or such other officer of the Borrower acceptable to the Agent.

"Fiscal Quarter" means, in relation to the Borrower, the Parent or any Subsidiary thereof, any fiscal quarter of such Person.

"Fiscal Year" means, in relation to the Borrower, the Parent or any Subsidiary thereof, any fiscal year of such Person.

"Fixed Charge Coverage Ratio" means, as of the last day of any calendar month, the ratio of (a) without duplication EBITDA for the Rolling Period ended on that date minus non-finance Capital Expenditures made by the Borrower during such Rolling Period to (b) the sum of (i) Interest Expense of the Borrower for such Rolling Period plus (ii) the aggregate of all dividends, distributions and principal payments on Indebtedness made by the Borrower during such Rolling Period plus (iii) income taxes paid in cash or cash equivalents by the Borrower during such Rolling Period.

"Foreign Lender" means any Lender that is not a Canadian Resident Lender.

"F/X Bank" means Canadian Imperial Bank of Commerce.

"F/X Contract" means a currency exchange transaction or agreement or any option with respect to any such transaction now existing or hereafter entered into between the Borrower and the F/X Bank in accordance with Section 2.19.

"F/X Contract Sub-Line" means an aggregate amount of credit of up to but not exceeding U.S.\$10,000,000 (or the Canadian \$ Equivalent thereof) to assist the Borrower in obtaining F/X Contracts from the F/X Bank pursuant to Section 2.19.

"F/X Exposure" means, at any time, and subject to the F/X Contract Sub-Line, the sum of: (a) the amount determined by the Agent (acting reasonably with consideration given to any determinations provided to the Agent by the F/X Bank) to be the credit risk associated with all outstanding F/X Contracts, plus (b) the aggregate amount of all Reimbursement Obligations in respect of all F/X Contracts at such time. The F/X Exposure of all Lenders shall not exceed the F/X Contract Sub-Line. Any F/X Exposure denominated in any currency other than Canadian Dollars shall be the Canadian \$ Equivalent thereof.

"GAAP" means at any particular time with respect to any Credit Party, generally accepted accounting principles as in effect at such time in the United States, consistently applied.

"General Partner" means Zellstoff Celgar Limited, a British Columbia corporation, its successors and permitted assigns.

"Governmental Authority" means the Government of Canada, any other nation or any political subdivision thereof, whether provincial, state, territorial or local, and any agency, authority, instrumentality, regulatory body, court, central bank, fiscal or monetary authority or other authority regulating financial institutions, and any other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government, including the Bank Committee on Banking Regulation and Supervisory Practices of the Bank of International Settlements.

"GST" means all amounts payable under the ETA or any similar legislation in any other jurisdiction of Canada, including QST and HST.

"Guarantee" of or by any Person (in this definition, the "guarantor") means any obligation, contingent or otherwise, of the guarantor guaranteeing or having the economic effect of guaranteeing any Indebtedness or other obligation of any other Person (in this definition, the "primary credit party") in any manner, whether directly or indirectly, and including any obligation of the guarantor, direct or indirect, (a) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation or to purchase (or to advance or supply funds for the purchase of) any security for the payment thereof (whether in the form of a loan, advance, stock purchase, capital contribution or otherwise), (b) to purchase or lease property, securities or services for the purpose of assuring the owner of such Indebtedness or other obligation of the payment thereof, (c) to maintain working capital, equity capital solvency, or any other balance sheet, income statement or other financial statement condition or liquidity of the primary credit party so as to enable the primary credit party to pay such Indebtedness or other obligation, (d) as an account party in respect of any letter of credit or letter of guarantee issued to support such Indebtedness or other obligation, or (e) to purchase, sell or lease (as lessor or lessee) property, or to purchase or sell services, primarily for the purpose of enabling the debtor to make payment of such Indebtedness or to assure the holder of such Indebtedness against loss.

"Hazardous Materials" means any substance, product, liquid, waste, pollutant, chemical, contaminant, insecticide, pesticide, gaseous or solid matter, organic or inorganic matter, fuel, micro-organism, ray, odour, radiation, energy, vector, plasma, constituent or material which (a) is or becomes listed, regulated or addressed under any Environmental Laws, or (b) is, or is deemed to be, alone or in any combination, hazardous waste, toxic, a pollutant, a deleterious substance, a contaminant or a source of pollution or contamination under any Environmental Laws, including, asbestos, petroleum and polychlorinated biphenyls, including petroleum or petroleum distillates, asbestos or asbestos containing materials, polychlorinated biphenyls, radon gas, infectious or medical wastes and all other substances or wastes of any nature regulated pursuant to any Environmental Laws.

"HST" means all amounts payable as harmonised sales tax in the Provinces of Ontario, Nova Scotia, Newfoundland and New Brunswick under the ETA.

"Indebtedness" of any Person means, without duplication, (a) all obligations of such Person for borrowed money or with respect to deposits or advances of any kind, (b) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (c) all obligations of such Person upon which interest charges are customarily paid, (d) all obligations of such Person under conditional sale or other title retention agreements relating to property acquired by such Person, (e) all obligations of such Person in respect of the deferred purchase price of property or services (excluding current accounts payable incurred in the ordinary course of business), (f) all Indebtedness of others secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any Lien on property owned or acquired by such Person, whether or not the Indebtedness secured thereby has been assumed, (g) all Guarantees by such Person of Indebtedness of others, (h) all Capital Lease Obligations of such Person, (i) all obligations, contingent or otherwise, of such Person as an account party in respect of letters of credit and letters of guarantee, (j) all obligations, contingent or otherwise, of such Person in respect of bankers' acceptances, (k) the net amount of obligations of such Person (determined on a mark-to-market basis) under Swap Agreements, and (1) all obligations of such Person to purchase, redeem, retire, defease or otherwise acquire for value (other than for other Equity Securities) any Equity Securities of such Person, valued, in the case of redeemable Equity Securities, at the greater of voluntary or involuntary liquidation preference, plus accrued and unpaid dividends. The Indebtedness of any Person shall include the Indebtedness of any other entity (including any partnership in which such Person is a general or limited partner) to the extent such Person is liable therefor as a result of such Person's ownership interest in or other relationship with such entity, except to the extent the terms of such Indebtedness provide that such Person is not liable therefor.

"Indemnified Taxes" means all Taxes other than Excluded Taxes and for greater certainty, includes, without limitation, capital Taxes imposed on (or measured by) the taxable capital of the Lenders as a result of the Borrower's status as a limited partnership.

"Indemnitee" has the meaning set out in Section 9.3(b).

"Insured Account" means an Account which is fully insured by export/import insurance provided by Export Development Corporation (Canada), AIG Global Trade & Political Risk or Euler Hermes (including their respective successors and permitted assigns) or such other credit insurance as acceptance to the Agent and Lenders.

"Interest Expense" shall mean, for any period, the total interest expense of the Borrower and its Restricted Subsidiaries on a consolidated basis, plus, to the extent not included in such total interest expense, and to the extent incurred by the Borrower or any of its Restricted Subsidiaries, (i) interest expense attributable to Capital Lease Obligations of the Borrower or its Restricted Subsidiaries, (ii) amortization of debt discount or financing fees, (iii) capitalized interest, (iv) non-cash interest expense, (v) commissions, discounts and other fees and charges owed with respect to letters of credit and bankers' acceptance financing, (vi) net costs associated with Swap Agreements (including amortization of fees), (vii) standby fees, (viii) preferred stock dividends in respect of all preferred stock issued by the Borrower or a Restricted Subsidiary and held by Persons other than the Borrower or a Restricted Subsidiary, and (ix) interest actually paid by the Borrower or any Restricted Subsidiary on any Indebtedness of any other Person.

"Interest Payment Date" means, (a) in the case of any Loan other than a LIBO Rate Loan or a B/A Borrowing, the first Business Day of each month, (b) in the case of a LIBO Rate Loan, the last day of each Interest Period relating to such LIBO Rate Loan, and (c) in the case of a B/A Borrowing, on the last day of each Contract Period relating to such B/A Borrowing.

"Interest Period" means, with respect to a LIBO Rate Borrowing, the period commencing on the date of such Borrowing and ending on the numerically corresponding day in the calendar month that is 30, 60 or 90 days thereafter, as the Borrower may elect; provided that (i) if any Interest Period would end on a day other than a Business Day, such Interest Period shall be extended to the immediately succeeding Business Day unless such next succeeding Business Day would fall in the next calendar month, in which case such Interest Period shall end on the next preceding Business Day, (ii) any Interest Period that commences on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the last calendar month of such Interest Period) shall end on the last Business Day of the last calendar month of such Interest Period, (iii) no Interest Period shall extend beyond any date that any principal payment or prepayment is scheduled to be due unless the aggregate principal amount of (A) Canadian Prime Borrowings and Base Rate Borrowings and (B) B/A Borrowings and LIBO Rate Borrowings which have Interest Periods or Contract Periods which will expire on or before such date, less the aggregate amount of any other principal payments or prepayments due during such Interest Period, is equal to or in excess of the amount of such principal payment or prepayment, and (iv) no Interest Period shall extend beyond the Maturity Date. For purposes hereof, the date of a Borrowing initially shall be the date on which such Borrowing is made and, in the case of a converted or continued Borrowing, thereafter shall be the effective date of the most recent conversion or continuation of such Borrowing.

"Inventory" means, in respect of each Credit Party, all of such Credit Party's present and hereafter acquired inventory (as defined in the PPSA) and including all merchandise, inventory and goods, and all additions, substitutions and replacements thereof, wherever located, together with all goods and materials used or usable in manufacturing, processing, packaging or shipping same in all stages of production from raw materials through work in process to finished goods, and all "stores" inventory or "operating and maintenance supplies" inventory, and all proceeds of any thereof (of whatever sort).

"Investment" means, as applied to any Person (the "investor"), any direct or indirect purchase or other acquisition by the investor of, or a beneficial interest in, Equity Securities of any other Person, including any exchange of Equity Securities for Indebtedness, or any direct or indirect loan, advance (other than advances to employees for moving and travel expenses, drawing accounts and similar expenditures in the ordinary course of business) or capital contribution by the investor to any other Person, including all Indebtedness and Accounts owing to the investor from such other Person that did not arise from sales or services rendered to such other Person in the ordinary course of the investor's business, or any direct or indirect purchase or other acquisition of bonds, notes, debentures or other debt securities of, any other Person. The amount of any Investment shall be the original cost of such Investment plus the cost of all additions thereto, without any adjustments for increases or decreases in value, or write-ups, write-downs or write-offs with respect to such Investment minus any amounts (a) realized upon the disposition of assets comprising an Investment (including the value of any liabilities assumed by any Person other than the Borrower or any Restricted Subsidiary in connection with such disposition), (b) constituting repayments of Investments that are loans or advances or (c) constituting cash returns of principal or capital thereon (including any dividend, redemption or repurchase of equity that is accounted for, in accordance with GAAP, as a return of principal or capital).

"Investment Grade Account" means an Eligible Account owing to a Credit Party by an obligor that is (i) rated as "investment grade" or higher rating from any nationally recognized rating agency, with "investment grade" being (a) BBB- for S&P, (b) BBB(low) for DBRS, and (c) Baa3 for Moody's, or (ii) otherwise designated in writing as "investment grade" by the Agent. As of the date hereof until such time as the Agent provides written notice to the Borrower, an Eligible Account owing to a Credit Party by Central National-Gottesman Inc. is designated as "investment grade".

"Issuing Bank" means the bank issuing Letters of Credit for the Borrower with the assistance of the Agent in accordance with Section 2.18.

"ITA" means the Income Tax Act (Canada).

"Lender" means any Lender having a Commitment hereunder and/or a Revolving Loan outstanding hereunder.

"Lender Affiliate" means, with respect to any Lender, an Affiliate of such Lender.

"Lenders" means the Persons listed as lenders on Schedule A and any other Person that shall have become a party hereto pursuant to an Assignment and Transfer, other than any such Person that ceases to be a party hereto pursuant to an Assignment and Transfer.

"Letter of Credit" means a letter of credit issued by the Issuing Bank for or on behalf of the Borrower with the assistance of the Agent in accordance with Section 2.18.

"Letter of Credit Exposure" means, at any time and subject to the Letter of Credit Sub-Line, the sum of: (a) the aggregate face amount of all outstanding Letters of Credit at such time, plus (b) the aggregate amount of all Reimbursement Obligations in respect of all Letter of Guarantees at such time. The Letter of Credit Exposure of any Lender at any time shall be its Applicable Percentage of the total Letter of Credit Exposure at such time with the total of all such Letter of Credit Exposure of all Lenders not to exceed the Letter of Credit Sub-Line. Any Letter of Credit Exposure denominated in U.S. Dollars shall be the Cdn.\$ Equivalent thereof.

"Letter of Credit Guarantee" means the agreement for the substitution of applicants or such other form of guarantee or indemnity agreement which is acceptable to the Issuing Bank and the Agent, supporting the issuance of Letters of Credit by the Issuing Bank provided the aggregate amount of all such Letters of Credit issued and to be issued shall not exceed the Letter of Credit Sub-Line.

"Letter of Credit Sub-Line" means the amount of the commitment by the Agent and the Lenders hereunder, in an aggregate amount up to but not exceeding Cdn.\$3,000,000, to assist the Borrower in obtaining Letters of Credit.

"LIBO Rate" means, for any Interest Period, the rate for U.S. Dollar borrowings appearing on Page 3750 of the Telerate Service or Page LIBOR01 of the Reuters Service (or on any successor or substitute page of such Service, or any successor to or substitute for such Service providing rate quotations comparable to those currently provided on such page of such Service, as determined by the Agent from time to time for purposes of providing quotations of interest rates applicable to U.S. Dollar deposits in the London interbank market) at approximately 11:00 a.m., London time, two Business Days prior to the commencement of such Interest Period, as the rate for U.S. Dollar deposits with a maturity comparable to such Interest Period. In the event that (i) such rate is not available at such time for any reason, then the "LIBO Rate" for such Interest Period shall be the rate at which U.S. Dollar deposits of \$5,000,000 and for a maturity comparable to such Interest Period are offered by the principal London office of the Agent in immediately available funds in the London interbank market at approximately 11:00 a.m., London time, two Business Days prior to the commencement of such Interest Period, and (ii) such rate determined as hereinabove contemplated is less than zero, it shall be deemed to be zero.

"LIBO Rate Borrowing" means a Borrowing comprised of one or more LIBO Rate Loans.

"LIBO Rate Loan" means a Loan denominated in U.S. Dollars which bears interest at a rate based upon the LIBO Rate.

"Lien" means, (a) with respect to any asset, any mortgage, deed of trust, lien, pledge, hypothec, hypothecation, encumbrance, charge, security interest, royalty interest, adverse claim, defect of title or right of set off in, on or of such asset, (b) the interest of a vendor or a lessor under any conditional sale agreement, capital lease, title retention agreement or consignment agreement (or any financing lease having substantially the same economic effect as any of the foregoing) relating to any asset, (c) in the case of securities, any purchase option, call or similar right of a third party with respect to such securities, (d) any netting arrangement, defeasance arrangement or reciprocal fee arrangement, and (e) any other arrangement having the effect of providing security.

"Limited Partnership Agreement" means the limited partnership agreement dated January 10, 2006 among the General Partner, as general partner and the Parent, as limited partner, constituting Zellstoff Celgar Limited Partnership, a limited partnership organized under the laws of British Columbia.

"Loan" means any loan made by the Lenders to the Borrower pursuant to this Agreement.

"Loan Documents" means this Agreement, the Security Documents, the Subordination Agreement, the Blocked Account Agreement, the Borrowing Requests and the Borrowing Base Reports, together with any other document, instrument or agreement (other than participation, agency or similar agreements among the Lenders or between any Lender and any other bank or creditor with respect to any indebtedness or obligations of any Credit Party hereunder or thereunder) now or hereafter entered into in connection with this Agreement (including any Swap Agreements and any F/X Contracts), as such documents, instruments or agreements may be amended, modified or supplemented from time to time.

"Material Adverse Change" means any event, development or circumstance that has had or could in the opinion of the Required Lenders reasonably be expected to have a Material Adverse Effect.

"Material Adverse Effect" means a material adverse effect on (a) the business, assets, operations or condition, financial or otherwise, of the Credit Parties taken as a whole, or (b) the validity or enforceability of any of the Loan Documents, the priority of the Liens created thereby or the rights and remedies of the Agent and the Lenders thereunder or (c) any Material Contract, or (d) the amount which the Lenders would be likely to receive (after giving effect to delays in payment and costs of enforcement) upon the liquidation of the Collateral.

"Material Contract" means (a) the contracts, licences and agreements listed and described on Schedule C, and (b) any other contract, licence or agreement (i) to which any Credit Party is a party or bound, (ii) which is material to, or necessary in, the operation of the business of any Credit Party, and (iii) which a Credit Party cannot promptly replace by an alternative and comparable contract with comparable commercial terms.

"Material Indebtedness" means (a) the Parent Subordinated Debt, and (b) any Indebtedness (other than the Loans and the Parent Subordinated Debt) of any one or more of the Credit Parties in an aggregate principal amount exceeding Cdn.\$4,000,000.

"Maturity Date" means July 16, 2023.

"Moody's" means Moody's Investors Service, Inc.

"Obligations" means all obligations, liabilities and indebtedness of the Borrower to the Agent, the Lenders or a Lender with respect to the principal of and interest on the Loans and the payment or performance of all other obligations, liabilities and indebtedness of the Borrower to the Agent, the Lenders or a Lender hereunder or arising under or pursuant to any one or more of the other Loan Documents or with respect to the Loans, including, without limitation, all reimbursement and indemnity obligations of the Borrower to the Agent, the Lenders or a Lender hereunder or in connection with any Letter of Credit Guarantee, F/X Contract or otherwise.

"Operating Account" means the bank account maintained by the Borrower at a financial institution acceptable to the Agent, acting reasonably.

"Original Effective Date" means the date on which all of the conditions specified in Section 4.1 were satisfied or waived in accordance with Section 9.2, as confirmed in a written notice from the Agent to the Borrower pursuant to the credit agreement dated as of May 19, 2006 among the Borrower, the Lenders and the Agent.

"Out-of-Pocket Expenses" means all of the Agent's present and future expenses incurred relative to this Agreement or any other Loan Documents, whether incurred heretofore or hereafter, which expenses shall include, without being limited to: the reasonable cost of retaining external legal counsel, record searches, all costs and expenses incurred by the Agent in opening bank accounts, depositing cheques, receiving and transferring funds, and wire transfer charges, any charges imposed on the Agent due to returned items and "insufficient funds" of deposited cheques and the Agent's standard fees relating thereto, any amounts paid by, incurred by or charged to, the Agent by the Issuing Bank under a Letter of Credit Guarantee, by the F/X Bank under an F/X Contract or the reimbursement agreements related thereto, applications for Letters of Credit, F/X Contracts or other like document which pertain either directly or indirectly to such Letters of Credit or F/X Contracts, and the Agent's standard fees relating to the Letters of Credit, F/X Contracts and any drafts thereunder, reasonable travel, lodging and similar expenses of the Agent's personnel (or any of its agents) in connection with inspecting and monitoring the Collateral from time to time at reasonable intervals hereunder, any applicable reasonable counsel fees and disbursements, fees and taxes relative to the filing of financing statements, and all expenses, costs and fees set forth incurred by or imposed on the Agent by reason of the exercise of any of its rights and remedies under this Agreement or any of the other Loan Documents.

"Parent" means Mercer International Inc. and its successors and permitted assigns.

"Parent Credit Agreement" means any credit or financing agreement now or hereafter entered into by the Borrower in favour of the Parent evidencing the Parent Subordinated Debt, at all times subject to the Subordination Agreement.

"Parent Guarantee" means a guarantee dated May 19, 2006 made by the Parent to and in favour of the Agent.

"Parent Pulp Agency Agreement" means the pulp agency agreement to be entered into between the Parent and the Borrower after the Original Effective Date pursuant to which the Parent will agree to market pulp on behalf of the Borrower on commercially reasonable terms provided that the commission rate payable by the Borrower to the Parent thereunder will not exceed 2% based on net product sales.

"Parent Subordinated Debt" means unsecured Indebtedness of the Borrower, up to an aggregate amount of U.S.\$15,000,000, owing to the Parent from time to time, which shall at all times be subordinated to the Loans.

"Participant" has the meaning set out in Section 9.4.

"Payment Office" means the Agent's office located at 199 Bay Street, 4th Floor, Toronto, Ontario, M5L 1A2, Attention: Senior Director, Portfolio Management, Asset-Based Lending Group (or such other office or individual as the Agent may hereafter designate in writing to the other parties hereto).

"Pension Plan" means any pension or other employee benefit plan (including any plan subject to the *Pension Benefits Act* (British Columbia), as amended from time to time (or any successor statute) in respect of which (i) is maintained by any Credit Party, (ii) any Credit Party makes or has made, has made or is required to make (at any time during the five (5) calendar years preceding the date of this Agreement) contributions in respect of its employees, or (iii) any other plan with respect to which any Credit Party has incurred or may incur liability, including contingent liability either to such plan or to any Person, administration or Governmental Authority in respect of which any Credit Party makes or has made contributions in respect of its employees.

"Permitted Investments" means:

- (a) direct obligations of, or obligations the principal of and interest on which are unconditionally guaranteed by, the Government of Canada or of any Canadian province (or by any agency thereof to the extent such obligations are backed by the full faith and credit of the Government of Canada or of such Canadian province), in each case maturing within one year from the date of acquisition thereof;
- (b) investments in commercial paper maturing within 270 days from the date of acquisition thereof and having, at such date of acquisition, the highest credit rating obtainable from any of Moody's, S&P or DBRS;
- (c) investments in certificates of deposit, bankers' acceptances and time deposits maturing within 180 days from the date of acquisition thereof issued or guaranteed by or placed with, and money market deposit accounts issued or offered by, any Schedule I bank under the Bank Act (Canada).

"Permitted Liens" means:

- (a) Liens in favour of the Lenders for the obligations of the Borrower or any other Credit Party under or pursuant to the Loan Documents;
- (b) Liens granted by a Credit Party in favour of another Credit Party in order to secure any of its indebtedness to such other Credit Party, provided that such Liens are subject to assignment and postponement arrangements satisfactory to the Agent;
- (c) Purchase Money Liens securing Indebtedness and Liens to secure Capital Lease Obligations, in each case only to the extent permitted by Section 6.1(f);
- (d) Liens imposed by any Governmental Authority for Taxes not yet due and delinquent or which are being contested in good faith in compliance with Section 5.3, and, during such period during which such Liens are being so contested, such Liens shall not be executed on or enforced against any of the assets of any Credit Party;
- (e) carrier's, warehousemen's, mechanics', materialmen's, repairmen's, construction and other like Liens arising by operation of Applicable Law, arising in the ordinary course of business, which are not overdue for a period of more than 30 days or which are being contested in good faith and by appropriate proceedings, and, during such period during which such Liens are being so contested, such Liens shall not be executed on or enforced against any of the assets of the Credit Party, provided that the Credit Party shall have set aside on its books reserves deemed adequate therefor and not resulting in qualification by auditors;

- (f) statutory Liens incurred or pledges or deposits made under worker's compensation, unemployment insurance and other social security legislation;
- (g) Liens or deposits to secure the performance of bids, tenders, trade contracts, leases, statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature (other than for borrowed money) incurred in the ordinary course of business
- (h) servitudes, easements, rights-of-way, restrictions and other similar encumbrances on real property imposed by Applicable Law or incurred in the ordinary course of business and encumbrances consisting of zoning or building restrictions, easements, licenses, restrictions on the use of property or minor imperfections in title thereto which, in the aggregate, are not material, and which do not in any case materially detract from the value of the property subject thereto or interfere with the ordinary conduct of the business of the Credit Parties;
- (i) Liens of or resulting from any judgement or award, the time for the appeal or petition for rehearing of which shall not have expired, or in respect of which the Credit Parties shall at any time in good faith be prosecuting an appeal or proceeding for review, so long as the such judgements or awards do not at any time exceed \$4,000,000 in the aggregate;
- (j) undetermined or inchoate Liens and charges arising or potentially arising under statutory provisions which have not at the time been filed or registered in accordance with Applicable Law or of which written notice has not been duly given in accordance with Applicable Law or which although filed or registered, relate to obligations not due or delinquent;
- (k) the rights reserved to or vested in Governmental Authorities by statutory provisions or by the terms of leases, licenses, franchises, grants or permits, which affect any land, to terminate the leases, licenses, franchises, grants or permits or to require annual or other periodic payments as a condition of the continuance thereof;
- (l) securities to public utilities or to any municipalities or Governmental Authorities or other public authority when required by the utility, municipality or Governmental Authorities or other public authority in connection with the supply of services or utilities to a Credit Party;
- (m) Liens or covenants restricting or prohibiting access to or from lands abutting on controlled access highways or covenants affecting the use to which lands may be put; provided that, in the case of a Credit Party such Liens or covenants do not materially and adversely affect the use of the lands by the Credit Party;
- (n) Liens consisting of royalties payable with respect to any asset or property of a Credit Party existing as of the Original Effective Date; provided that the existence of any such Lien on any material property or asset of a Credit Party shall have been disclosed in writing to the Lenders prior to the Original Effective Date;

- (o) Liens securing reimbursement obligations relating to letters of credit issued pursuant to this Agreement, provided that the value of the collateral subject to any such Lien does not exceed the amount of the related reimbursement obligation;
- (p) statutory Liens incurred or pledges or deposits made in favour of a Governmental Authority to secure the performance of obligations of a Credit Party under Environmental Laws to which any assets of such Credit Party are subject, provided that no Default or Event of Default shall have occurred and be continuing;
- (q) a Lien granted by a Credit Party to a landlord to secure the payment of arrears of rent in respect of leased properties in the Province of Quebec leased from such landlord, provided that such Lien is limited to the assets located at or about such leased properties;
- (r) any Lien on any property or asset of a Credit Party existing on the date hereof and set forth in Schedule 3.10; provided that (i) such Lien shall not apply to any other property or asset of such Credit Party, and (ii) such Lien shall secure only those obligations which it secures on the date hereof;
- (s) any Lien existing on any property or asset prior to the acquisition thereof by a Credit Party or existing on any property or asset of any Person that becomes a Credit Party after the date hereof prior to the time such Person becomes a Credit Party; provided that (i) such Lien is not created in contemplation of or in connection with such acquisition or such Person becoming a Credit Party, as the case may be, (ii) such Lien shall not apply to any other property or assets of such Credit Party, and (iii) such Lien shall secure only those obligations which it secures on the date of such acquisition or the date such Person becomes a Credit Party, as the case may be;
- (t) Liens securing Indebtedness to the extent permitted by Section 6.1(l); and
- (u) any extension, renewal or replacement of any of the foregoing; provided, however, that the Liens permitted hereunder shall not be extended to cover any additional Indebtedness of the Credit Parties or their property (other than a substitution of like property), except Liens in respect of Capital Lease Obligations and Purchase Money Liens as permitted by (c) above.

"Person" includes any natural person, corporation, company, limited liability company, trust, joint venture, association, incorporated organization, partnership, Governmental Authority or other entity.

"Phase I Environmental Report" means the existing phase I environmental assessment report prepared by URS Canada Inc. dated November, 2004.

"PPSA" means the *Personal Property Security Act* (British Columbia), as amended from time to time.

"Priority Payables" means, with respect to any Person, any amount payable by such Person which is secured by a Lien in favour of a Governmental Authority which ranks or is capable of ranking prior to or pari passu with the Liens created by the Security Documents in respect of any Eligible Accounts or Eligible Inventory, including amounts owing for wages, vacation pay, severance pay, employee deductions, sales tax, excise tax, Tax payable pursuant to Part IX of the Excise Tax Act (Canada) (net of GST input credits), income tax, workers compensation, government royalties, pension fund obligations, overdue rents or Taxes, and other statutory or other claims that have or may have priority over, or rank pari passu with, such Liens created by the Security Documents.

"Purchase Money Lien" means a Lien taken or reserved in personal property to secure payment of all or part of its purchase price, <u>provided</u> that such Lien (i) secures an amount not exceeding the purchase price of such personal property, (ii) extends only to such personal property and its proceeds, and (iii) is granted prior to or within 30 days after the purchase of such personal property.

"QST" means the Quebec sales tax imposed pursuant to an Act respecting the Québec sales tax.

"Register" has the meaning set out in Section 9.4(c).

"Reimbursement Obligations" means, at any date, the sum of the outstanding obligations of the Borrower to reimburse the Agent at such time to the extent that the Agent is obligated to reimburse (a) the Issuing Bank at such time pursuant to any Letter of Credit Guarantee and (b) the F/X Bank at such time pursuant to any F/X Contract.

"Related Parties" means, with respect to any Person, such Person's Affiliates and the respective directors, officers, employees, agents and advisors of such Person and of such Person's Affiliates.

"Release" is to be broadly interpreted and shall include an actual or potential discharge, deposit, spill, leak, pumping, pouring, emission, emptying, injection, escape, leaching, seepage or disposal of a Hazardous Materials which is or may be in breach of any Environmental Laws.

"Required Lenders" means, at any time, Lenders having Exposure and unused Commitments representing more than 50% of the sum of the total Exposure and unused Commitments at such time.

"Responsible Officer" means, with respect to any Person, the chairman, the president, any vice president, the chief executive officer, the chief operating officer, or such other officer acceptable to the Agent and, in respect of financial or accounting matters, any Financial Officer of such Person; unless otherwise specified, all references herein to a Responsible Officer mean a Responsible Officer of the Borrower.

"Restricted Payment" shall mean, with respect to any Person, any payment by such Person (i) of any dividends on any of its Equity Securities, (ii) on account of, or for the purpose of setting apart any property for a sinking or other analogous fund for, the purchase, redemption, retirement or other acquisition of any of its Equity Securities or any warrants, options or rights to acquire any such shares, or the making by such Person of any other distribution in respect of any of its Equity Securities, (iii) of any principal of or interest or premium on or of any amount in respect of a sinking or analogous fund or defeasance fund for any Indebtedness of such Person to a shareholder of such Person or to an Affiliate of a shareholder of such Person, (v) in respect of an Investment, other than Permitted Investments, or (vi) of any management, consulting or similar fee or any bonus payment or comparable payment, or by way of gift or other gratuity, to any Affiliate of such Person or to any director or officer thereof.

- "Restricted Subsidiary" means each Subsidiary of the Borrower which is not an Unrestricted Subsidiary.
- "Revolving Loan" has the meaning set out in Section 2.1.
- "RMB" means the People's Republic of China Renminbi.
- "Rolling Period" means, as at the end of any calendar month such calendar month taken together with the eleven immediately preceding calendar months.
 - "S&P" means Standard & Poor's Ratings Services, a division of the McGraw-Hill Companies, Inc.
- "Sale/Leaseback Obligations" of any Person means the obligations of such Person to pay rent or other amounts under a Sale/Leaseback Transaction.
- "Sale/Leaseback Transaction" means any arrangement between a Credit Party or any Restricted Subsidiary with any Person whereby the Credit Party or any such Restricted Subsidiary shall sell or transfer any property, whether now owned or hereafter acquired, and whereby the Credit Party or any such Restricted Subsidiary shall then or thereafter rent or lease as lessee such property or any part thereof or other property which the Credit Party or any such Restricted Subsidiary intends to use for substantially the same purpose or purposes as the property sold or transferred.
- "Sanctions" means, with respect to any Person at any time, economic or financial sanctions or trade embargoes imposed, administered or enforced by any Sanctioning Authority that are applicable to such Person at such time.
- "Sanctioned Country" means, at any time, a country, region or territory (or subdivision thereof) which is itself the subject or target of any Sanctions.
- "Sanctioned Person" means, at any time, any Person with whom any Party is prohibited or restricted from transacting or otherwise dealing under any Sanction, whether by reason of designation under such Sanction or otherwise.
- "Sanctioning Authority" means a Governmental Authority in a country where the Borrower or its Subsidiaries conduct business operations or have a presence and/or owns assets that is authorized to impose, administer and enforce Sanctions.
- "Security Documents" means the agreements, documents or instruments described or referred to in Section 4.1 and Section 5.11 (including, to the extent such Section describes an amendment, the agreement, document or instrument amended thereby) and any and all other agreements, documents or instruments now or hereafter executed and delivered by any Credit Party or any other Person as security for the payment or performance of all or part of the obligations of the Borrower (or such Credit Party or other Person) hereunder or under any other Loan Documents, as any of the foregoing may have been, or may hereafter be, amended, modified or supplemented.

"Settlement Date" means the date, which shall be weekly, or more frequently at the discretion of the Agent upon the occurrence of an Event of Default or a continuing decline or increase of the Loans, that the Agent and the Lenders shall settle among themselves so that (a) the Agent shall not at any time have, as the agent for the Lenders, any money at risk, and (b) on such Settlement Date each Lender shall be responsible for its pro rata amount of the Revolving Loan, calculated on the basis of each of their Applicable Percentages in respect of the outstanding Exposure as at such date, provided that each Settlement Date shall be a Business Day.

"Subordination Agreement" means the subordination agreement between the Agent and the Parent in respect of the Parent Subordinated Debt.

"Subsidiary" means, with respect to any Person (in this definition, the "parent") at any date, any other Person the accounts of which would be consolidated with those of the parent in the parent's consolidated financial statements if such financial statements were prepared in accordance with GAAP as of such date, as well as any other Person (a) of which securities or other ownership interests representing more than 50% of the equity or more than 50% of the ordinary voting power or, in the case of a partnership, more than 50% of the general partnership interests are, as of such date, owned, controlled or held, or (b) that is, as of such date, otherwise Controlled, by the parent or one or more Subsidiaries of the parent or by the parent and one or more Subsidiaries of the parent.

"Swap Agreement" means any agreement with respect to any swap, forward, future or derivative transaction or option or similar agreement involving, or settled by reference to, one or more rates, currencies, commodities, equity or debt instruments or securities, or economic, financial or pricing indices or measures of economic, financial or pricing risk or value or any similar transaction or any combination of these transactions.

"Taxes" means all taxes, charges, fees, levies, imposts and other assessments, including all income, sales, use, goods and services, value added, capital, capital gains, alternative, net worth, transfer, profits, withholding, payroll, employer health, excise, real property and personal property taxes, and any other taxes, customs duties, fees, assessments, or similar charges in the nature of a tax, including Canada Pension Plan and provincial pension plan contributions, unemployment insurance payments and workers' compensation premiums, together with any instalments with respect thereto, and any interest, fines and penalties with respect thereto, imposed by any Governmental Authority (including federal, state, provincial, municipal and foreign Governmental Authorities), and whether disputed or not.

"Threshold Amount" has the meaning given to it in Section 5.12.

"**Transactions**" means the execution, delivery and performance by the Borrower of this Agreement and the other Loan Documents, the borrowing of Loans, the use of the proceeds thereof, the issuance of Letters of Credit hereunder and the entering into of F/X Contracts as contemplated herein.

"Type", when used in reference to any Loan or Borrowing, refers to whether the rate of interest on such Loan, or on the Loans comprising such Borrowing, is determined by reference to the Canadian Prime Rate, the CDOR Rate, the Base Rate, the LIBO Rate, or is an LC.

- "Unrestricted Subsidiary" means any Subsidiary of the Borrower which, together with its Subsidiaries, represents less than 5% of the consolidated assets or consolidated gross revenues of the Borrower.
 - "U.S. Dollars" and "U.S.\$" refer to lawful money of the United States of America.
- "U.S.\$ Equivalent" means, on any day, the amount of U.S. Dollars that the Agent could purchase, in accordance with its normal practice, with a specified amount of Canadian Dollars based on the spot rate at which U.S. Dollars are offered at the start of such day by the Agent in Toronto, Ontario.
- **1.2** <u>Classification of Loans and Borrowings</u>. For purposes of this Agreement, Loans may be classified and referred to by Type (e.g., a "Canadian Prime Rate Loan") and Borrowings also may be classified and referred to by Type (e.g., a "Canadian Prime Rate Borrowing").
- 1.3 Terms Generally. The definitions of terms herein shall apply equally to the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "include", "includes" and "including" shall be deemed to be followed by the phrase "without limitation". The word "will" shall be construed to have the same meaning and effect as the word "shall". The word "or" is disjunctive; the word "and" is conjunctive. The word "shall" is mandatory; the word "may" is permissive. The words "to the knowledge of" means, when modifying a representation, warranty or other statement of any Person, that the fact or situation described therein is known by the Person (or, in the case or a Person other than a natural Person, known by the Responsible Officer of that Person) making the representation, warranty or other statement, or with the exercise of reasonable due diligence under the circumstances (in accordance with the standard of what a reasonable Person in similar circumstances would have done) would have been known by the Person (or, in the case of a Person other than a natural Person, would have been known by such Responsible Officer of that Person). Unless the context requires otherwise (a) any definition of or reference to any agreement, instrument or other document herein shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein), (b) any reference herein to any statute or any section thereof shall, unless otherwise expressly stated, be deemed to be a reference to such statute or section as amended, restated or re-enacted from time to time, (c) any reference herein to any Person shall be construed to include such Person's successors and permitted assigns, (d) the words "herein", "hereof" and "hereunder", and words of similar import, shall be construed to refer to this Agreement in its entirety and not to any particular provision hereof, (e) all references herein to Articles, Sections, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Exhibits and Schedules to, this Agreement, and (f) the words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract rights.
- **1.4** Accounting Terms; GAAP. Except as otherwise expressly provided herein, all terms of an accounting or financial nature shall be construed in accordance with GAAP. All calculations for the purposes of determining compliance with the financial ratios and financial covenants contained herein shall be made on a basis consistent with GAAP in existence from time to time and used in the preparation of the financial statements of the Borrower referred to in Section 5.1(a). Upon the adoption by the Borrower of International Financial Reporting Standards, or in the event of a change in GAAP, the Borrower and the Agent shall negotiate in good faith to revise (if appropriate) such ratios and covenants to give effect to the intention of the parties under this Agreement as at the Effective Date, and any new ratio or covenant shall be subject to approval by the Required Lenders. In the event that such negotiation is unsuccessful, all calculations thereafter made for the purpose of determining compliance with the financial ratios and financial covenants contained herein shall be made on a basis consistent with GAAP in existence as at the Effective Date.

- **1.5** <u>Time</u>. All time references herein shall, unless otherwise specified, be references to local time in Toronto, Ontario. Time is of the essence of this Agreement and the other Loan Documents.
- **1.6** <u>Permitted Liens</u>. Any reference in any of the Loan Documents to a Permitted Lien is not intended to subordinate or postpone, and shall not be interpreted as subordinating or postponing, or as any agreement to subordinate or postpone, any Lien created by any of the Loan Documents to any Permitted Lien.
- **1.7** <u>Joint and Several</u>. For greater certainty, the liabilities of the Borrower and the General Partner hereunder and under the Security Documents are joint and several.
- 1.8 Effect of Amendment and Restatement of Existing Credit Agreement. The Existing Credit Agreement shall be amended and restated in its entirety as of the Effective Date upon satisfaction of the conditions precedent set forth in Section 4.1. The parties hereto acknowledge and agree that (a) this Agreement and the other agreements, documents and instruments executed and delivered in connection herewith do not constitute a novation, payment and reborrowing, or termination of the Obligations, (b) the Obligations are in all respects continuing (as amended and restated hereby), (c) the Liens as granted under the Loan Documents securing payment of Obligations are in all respect continuing and in full force and effect and secure the payment of Obligations (including, for greater certainty, those Obligations which any Credit Party has from time to time incurred or may incur or be under to the Agent or any Lender under, in connection with or with respect to this Agreement and the other Loan Documents to which such Credit Party is a party and any ultimate unpaid balance thereof), (d) upon the effectiveness of this Agreement, all (i) Loans (as defined in the Existing Credit Agreement) outstanding under the Existing Credit Agreement will be deemed to have been issued as Letter of Credit hereunder on the terms and conditions set forth in this Agreement, and any F/X exposure outstanding under the Existing Credit Agreement immediately before the effectiveness of this Agreement will continue as F/X Exposure made hereunder.
- **1.9** Reaffirmation. The Borrower has previously executed the Security Documents and (a) reaffirms and agrees that the Security Documents are and shall remain in full force and effect; (b) acknowledges and reaffirms all obligations owing by it to the Administrative Agent and the Lenders under any Security Document; (c) reaffirms and agrees that nothing in the Security Documents obligates the Administrative Agent or the Lenders to seek reaffirmation of the Security Documents in connection with similar matters in the future; and (d) reaffirms and agrees that no requirement to so notify the Borrower or to seek the Borrower's reaffirmation in connection with similar matters in the future shall be implied by the execution of this reaffirmation.

ARTICLE 2 THE CREDITS

2.1 Commitments. Subject to the terms and conditions set forth herein, each Lender commits to make Loans (each such Loan made under this Section 2.1, a "Revolving Loan") to the Borrower from time to time during the period commencing on the Original Effective Date and ending on the Maturity Date (each such commitment, a "Commitment") in an aggregate principal amount equal to the amount set forth beside such Lender's name in Schedule A under the heading "Commitment", provided that any Revolving Loans made by any Lender as requested by the Borrower will not result in (i) such Lender's Exposure exceeding such Lender's Commitment, or (ii) the sum of the total Exposure exceeding either the total Commitment or the Borrowing Base. Within the foregoing limits and subject to the terms and conditions set forth herein, the Borrower may borrow, repay and reborrow Revolving Loans. In addition, the Borrower may request from time to time that the Commitment be increased in an aggregate principal amount of up to \$10,000,000, provided that (i) the Agent and any Lender providing any portion of any such increase must consent in writing thereto (it being agreed that no Lender is required to provide any such Commitment increase), (ii) no increase in the Commitments shall be made if a Default or an Event of Default shall have occurred and be continuing or would result after giving effect to such increase, (iii) each such increase shall be in a minimum principal amount of \$2,500,000, (iv) the Borrower shall pay to the Agent, for the account of the Lenders, a one-time fee in an amount equal to 0.2% of the amount of each such Commitment increase, (v) the aggregate principal amount of all such Commitment increases shall not exceed \$10,000,000. The pro rata share of each Lender's Commitment hereunder shall automatically increase as a result of any permitted increase in the Commitment hereunder, and Schedule A shall be amended to reflect any such permitted increase.

2.2 Loans and Borrowings.

- (a) Each Revolving Loan shall be made as part of a Borrowing consisting of Revolving Loans made by the Lenders rateably in accordance with their respective Commitments. The failure of any Lender to make any Loan required to be made by it shall not relieve any other Lender of its obligations hereunder; provided that the Commitments of the Lenders are several and no Lender shall be responsible for any other Lender's failure to make Loans as required.
- (b) Subject to the Letter of Credit Sub-Line limitation, the F/X Contract Sub-Line limitation, the Borrowing Base limitations and the other limitations on Loans and Borrowings as provided in this Agreement, each Borrowing shall be comprised entirely of Canadian Prime Loans, Bankers Acceptances, B/A Equivalent Loans, Base Rate Loans, LIBO Rate Loans and/or the delivery of Letters of Credit Guarantees or the entry into F/X Contracts as the Borrower may request in accordance herewith. For greater certainty, the Agent's assistance in obtaining the F/X Contracts and the F/X Bank's agreement to provide the F/X Contracts shall at all times and in all respects be in the Agent's and F/X Bank's sole discretion.
- (c) Each Lender may at its option make any LIBO Rate Loan by causing any domestic or foreign branch or Affiliate of such Lender to make such Loan; provided that any exercise of such option shall not result in any increased costs for the Borrower or affect the obligation of the Borrower to repay such Loan in accordance with the terms of this Agreement. At the commencement of each Interest Period for any LIBO Rate Borrowing, such Borrowing shall be in an aggregate amount that is an integral multiple of Cdn.\$500,000 and not less than Cdn.\$1,000,000. At the commencement of each Contract Period for any B/A Borrowing, such Borrowing shall be in an aggregate amount that is an integral multiple of Cdn.\$500,000 and not less than Cdn.\$1,000,000. Borrowings of more than one Type and Class may be outstanding at the same time; provided that there shall not at any time be more than a total of five B/A Borrowings or five LIBO Rate Borrowings outstanding.

2.3 Requests for Borrowings.

- (a) To request a Borrowing, the Borrower shall notify the Agent of such request by written Borrowing Request (i) in the case of a LIBO Rate Borrowing, not later than 1:00 p.m., Toronto time, three Business Days before the date of the proposed Borrowing, (ii) in the case of a B/A Borrowing, not later than 1:00 p.m., Toronto time, two Business Days before the date of the proposed Borrowing, or (iii) in the case of a Canadian Prime Borrowing or a Base Rate Borrowing, not later than 1:00 p.m., Toronto time, one Business Day before the date of the proposed Borrowing; provided that any such notice of a Canadian Prime Borrowing or a Base Rate Borrowing to finance the reimbursement of an LC Disbursement may be given not later than 1:00 p.m., Toronto time, on the date of the proposed Borrowing, or (iv) in the case of the issuance of a Letter of Credit Guarantee in accordance with Section 2.18, not later than 1:00 p.m., Toronto time, five (5) Business Days before the date of the proposed Borrowing. The Agent and each Lender are entitled to rely and act upon any written Borrowing Request given or purportedly given by the Borrower, and the Borrower hereby waives the right to dispute the authenticity and validity of any such request or resulting transaction once the Agent or any Lender has advanced funds or made a B/A Borrowing or issued a Letter a Credit Guarantee based on such written Borrowing Request. Each such written Borrowing Request shall be substantially in the form of Exhibit B and shall specify the following information:
 - (i) the aggregate amount of each requested Borrowing and the Class thereof;
 - (ii) the date of such Borrowing, which shall be a Business Day;
 - (iii) whether such Borrowing is to be a Canadian Prime Borrowing, a B/A Borrowing, a Base Rate Borrowing, a LIBO Rate Borrowing, the issuance of a Letter of Credit Guarantee in accordance with Section 2.18;
 - (iv) in the case of a LIBO Rate Borrowing, the initial Interest Period to be applicable thereto, which shall be a period contemplated by the definition of the term "Interest Period", and in the case of a B/A Borrowing, the initial Contract Period to be applicable thereto, which shall be a period contemplated by the definition of the term "Contract Period"; and
 - (v) the location and number of the Borrower's account to which funds are to be disbursed, which shall comply with the requirements of this Agreement.
- (b) If no election as to the Type of Borrowing is specified, then the requested Borrowing shall be a Canadian Prime Borrowing (if denominated in Canadian Dollars) or a Base Rate Borrowing (if denominated in U.S. Dollars). If no currency is specified, the Borrowing shall be denominated in Canadian Dollars. If no Interest Period is specified with respect to any requested LIBO Rate Borrowing, then the Borrower shall be deemed to have selected an Interest Period of a one month duration. If no Contract Period is specified with respect to any requested B/A Borrowing, then the Borrower shall be deemed to have selected a Contract Period of a one month duration.

- (c) Each Borrowing initially shall be of the Type specified in the applicable Borrowing Request. Thereafter, the Borrower may elect to convert a Borrowing to a different Type or to continue such Borrowing and, in the case of (i) a LIBO Rate Borrowing, may elect a new Interest Period therefor, or (ii) a B/A Borrowing, may elect a new Contract Period therefor, all as provided in this Section 2.3(c). The Borrower may elect different options with respect to different portions of the affected Borrowing, in which case each such portion shall be allocated ratably among the Lenders holding the Loans comprising such Borrowing in accordance with their Applicable Percentage, and the Loans comprising each such portion shall be considered a separate Borrowing. To make an election pursuant to this Section 2.3(c), the Borrower shall notify the Agent of such election in the manner and by the time that a Borrowing Request would be required under Section 2.3(a) if the Borrower were requesting a Borrowing of the Type resulting from such election to be made on the effective date of such election. In addition to the information specified in Section 2.3(a), each Borrowing Request shall specify the Borrowing to which such request applies and, if different options are being elected with respect to different portions thereof, the portions thereof to be allocated to each resulting Borrowing.
- (d) In the absence of a timely and proper election with regard to (i) LIBO Rate Borrowings, the Borrower shall be deemed to have elected to convert such LIBO Rate Borrowings to Base Rate Borrowings on the last day of the Interest Period of the relevant LIBO Rate Borrowings, and (ii) B/A Borrowings, the Borrower shall be deemed to have elected to convert such B/A Borrowings to Canadian Prime Borrowings on the last day of the Contract Period of the relevant B/A Borrowings.
- (e) The Agent shall not incur any liability to the Borrower as a result of acting in accordance with any notice or request referred to in this Section 2.3, which notice or request the Agent believes in good faith to have been given by an officer duly authorized by the Borrower to request Loans on its behalf or for otherwise acting in good faith under this Section 2.3, and the crediting of Loans to the Borrower's disbursement accounts, or transmittal to such Person or other bank account as the Borrower shall direct, shall conclusively establish the obligation of the Borrower to repay such Loans as provided herein.
- (f) Except to the extent otherwise permitted to the contrary hereunder, any Borrowing Request made pursuant to in this Section 2.3 shall be irrevocable and the Borrower shall be bound to borrow the funds requested therein in accordance therewith.

2.4 Funding of Borrowings.

(a) Each Lender shall make each Loan to be made by it hereunder on the proposed date thereof by wire transfer of immediately available funds by 2:00 p.m., Toronto time, to the account of the Agent most recently designated by it for such purpose by notice to the Lenders. The Agent will make such Loans available to the Borrower by promptly crediting the amounts so received, in like funds, to an account of the Borrower and designated by the Borrower in the applicable Borrowing Request. The Borrower shall satisfy Reimbursement Obligations promptly as they arise by way of a request for a Loan and all Loans made hereunder to satisfy Reimbursement Obligations: (i) in respect of any Letter of Credit Guarantee shall be remitted by the Agent to the Issuing Bank in accordance with such Letter of Credit Guarantee (unless the Issuing Bank has already been fully reimbursed directly by the Borrower in respect of drawings under the Letter of Credit which is the subject of such Letter of Credit Guarantee), and (ii) in respect of any F/X Contract shall be remitted by the Agent to the F/X Bank in accordance with such F/X Contract (unless the F/X Bank has already been fully reimbursed directly by the Borrower in respect of all such losses in respect of the F/X Contract).

- (b) The Agent may, upon notice given by the Agent no later than 2:00 p.m., Toronto time on any Settlement Date, request each Lender to make, and each Lender hereby agrees to make, a Revolving Loan in an amount equal to such Lender's Applicable Percentage (calculated with respect to the aggregate Commitments then outstanding) of the aggregate amount of the Revolving Loans made by the Agent from the preceding Settlement Date to the date of such notice. Each Lender's obligation to make the Revolving Loans and to make the settlements pursuant to this Section 2.4 shall not be affected by any circumstance, including (i) any set-off, counterclaim, recoupment, defence or other right which any such Lender or the Borrower may have against the Agent, the Borrower, any Lender or any other Person for any reason whatsoever; (ii) any adverse change in the condition (financial or otherwise) of the Borrower; or (iii) any other circumstance, happening or event whatsoever, whether or not similar to any of the foregoing. Without limiting the liability and obligation of each Lender to make such advances, the Borrower authorizes the Agent to charge the Borrower's Operating Account to the extent amounts received from the Lenders are not sufficient to repay in full the amount of any such deficiency. To the extent that any Lender has failed to fund all such payments and Revolving Loans, the Agent shall be entitled to set off the funding short-fall against that Lender's pro rata share of all payments received from the Borrower.
- (c) The Agent, for the account of the Lenders, shall disburse all amounts to the Borrower and shall handle all collections. It is understood that for purposes of advances to the Borrower and for purposes of this Section 2.4, the Agent is using the funds of the Agent.
- (d) Unless the Agent shall have been notified in writing by any Lender prior to any advance to the Borrower that such Lender will not make the amount which would constitute its share of the Borrowing on such date available to the Agent, the Agent may assume that such Lender shall make such amount available to the Agent on a Settlement Date, and the Agent may, in reliance upon such assumption, make available to the Borrower a corresponding amount. A certificate of the Agent submitted to any Lender with respect to any amount owing under this Section 2.4 shall be conclusive, absent manifest error. If such Lender's share of such Borrowing is not in fact made available to the Agent by such Lender on the Settlement Date, the Agent shall be entitled to recover such amount with interest thereon at the rate per annum applicable to Revolving Loans hereunder, on demand, from the Borrower without prejudice to any rights which the Agent may have against such Lender hereunder. Nothing contained in this shall relieve any Lender which has failed to make available its Applicable Percentage of any borrowing hereunder from its obligation to do so in accordance with the terms hereof. Nothing contained herein shall be deemed to obligate the Agent to make available to the Borrower the full amount of a requested advance when the Agent has any notice (written or otherwise) that any of the Lenders will not advance its Applicable Percentage thereof.
- (e) On the Settlement Date, the Agent and the Lenders shall each remit to the other, in immediately available funds, all amounts necessary so as to ensure that, as of the Settlement Date, the Lenders shall have their Applicable Percentage share of all outstanding Obligations other than in respect of F/X Contracts, which shall remain with the F/X Bank.
- (f) The Agent shall forward to each Lender, at the end of each calendar month, a copy of the account statement rendered by the Agent to the Borrower.

(g) The Agent shall, after receipt of any interest and fees earned under this Agreement, promptly remit to the Lenders their Applicable Percentage of any (i) fees they are entitled to receive, and (ii) interest computed at the rate and as provided for in this Agreement on all outstanding amounts advanced by the Lenders on each Settlement Date, prior to adjustment, that are subsequent to the last remittance by the Agent to the Lenders of such interest amounts.

2.5 Interest.

- (a) The Loans comprising each Canadian Prime Borrowing shall bear interest (computed on the basis of the actual number of days elapsed over a year of 365 or 366 days, as the case may be) at a rate per annum equal to the Canadian Prime Rate plus (or, at any time when Adjusted Average Excess Availability is equal to or greater than 25%, minus) the Applicable Margin from time to time in effect. The Loans comprising each Base Rate Borrowing shall bear interest (computed on the basis of the actual number of days elapsed over a year of 360 days) at a rate per annum equal to the Base Rate plus (or, at any time when Adjusted Average Excess Availability is equal to or greater than 25%, minus) the Applicable Margin from time to time in effect. The Loans comprising each LIBO Rate Borrowing shall bear interest (computed on the basis of the actual number of days in the relevant Interest Period over a year of 360 days) at the LIBO Rate for the Interest Period in effect for such LIBO Rate Borrowing plus the Applicable Margin in effect on the first day of the relevant Interest Period. The Loans comprising each B/A Borrowing shall be subject to the Acceptance Fee which shall be payable as set out in Section 2.11.
- (b) If there is a Default or an Event of Default has occurred and is continuing, all amounts outstanding hereunder (including all Loans and all Letter of Credit Exposure and F/X Exposure) shall bear interest, after as well as before judgment, at a rate per annum equal to 2% plus the rate otherwise applicable to such Loan or, in the case of any amount not constituting principal or interest on a Loan, at a rate equal to 2% plus the rate otherwise applicable to, in the case of Canadian Dollar amounts, Canadian Prime Loans, or in the case of U.S. Dollar amounts, Base Rate Loans.
- (c) Accrued interest on each Loan shall be payable in arrears on the earlier of (i) each applicable Interest Payment Date, and (ii) the date of termination of the Commitments. In addition, in the event of any repayment or prepayment of any Loan, accrued interest on the principal amount repaid or prepaid shall be payable on the date of such repayment or prepayment.
- (d) All interest hereunder shall be payable for the actual number of days elapsed (including the first day but excluding the last day). Any Loan that is repaid on the same day on which it is made shall bear interest for one day. The applicable Canadian Prime Rate, Base Rate, LIBO Rate or B/A Rate shall be determined by the Agent, and such determination shall be conclusive absent manifest error.
- (e) For the purposes of the *Interest Act* (Canada) and disclosure thereunder, whenever any interest or any fee to be paid hereunder or in connection herewith is to be calculated on the basis of a 360-day, 365-day or 366-day year, the yearly rate of interest to which the rate used in such calculation is equivalent is the rate so used multiplied by the actual number of days in the calendar year in which the same is to be ascertained and divided by 360, 365 or 366, as applicable. The rates of interest under this Agreement are nominal rates, and not effective rates or yields. The principle of deemed reinvestment of interest does not apply to any interest calculation under this Agreement.

- (f) The Borrower acknowledges and confirms that:
 - (i) Section 2.5(e) above satisfies the requirements of Section 4 of the *Interest Act* (Canada) to the extent it applies to the expression or statement of any interest payable under any Loan Document; and
 - (ii) each Credit Party is able to calculate the yearly rate or percentage of interest payable under any Loan Document based upon the methodology set out in Section 2.5(e) above.
- (g) The Borrower agrees not to, and to cause each Credit Party not to, plead or assert, whether by way of defence or otherwise, in any proceeding relating to the Loan Documents, that the interest payable thereunder and the calculation thereof has not been adequately disclosed to any Credit Party, whether pursuant to Section 4 of the *Interest Act* (Canada) or any other Applicable Law or legal principle.
- (h) If any provision of this Agreement would oblige the Borrower to make any payment of interest or other amount payable to any Lender in an amount or calculated at a rate which would be prohibited by any Applicable Law or would result in a receipt by that Lender of "interest" at a "criminal rate" (as such terms are construed under the *Criminal Code* (Canada)), then, notwithstanding such provision, such amount or rate shall be deemed to have been adjusted with retroactive effect to the maximum amount or rate of interest, as the case may be, as would not be so prohibited by Applicable Law or so result in a receipt by that Lender of "interest" at a "criminal rate", such adjustment to be effected, to the extent necessary (but only to the extent necessary), as follows:
 - (i) first, by reducing the amount or rate of interest required to be paid to the affected Lender under Section 2.5; and
 - (ii) thereafter, by reducing any fees, commissions, costs, expenses, premiums and other amounts required to be paid to the affected Lender which would constitute interest for purposes of section 347 of the *Criminal Code* (Canada).
- (i) Any Loan that is outstanding on the Effective Date under the Existing Credit Agreement that is comprised of a Canadian Prime Borrowing shall, as of the Effective Date, bear interest in accordance with Section 2.5(a) hereof.
- (j) Any Loan that is outstanding on the Effective Date under the Existing Credit Agreement that is comprised of a B/A Borrowing or a LIBO Rate Borrowing shall continue to be subject to any fees and interest calculated in accordance with the Existing Credit Agreement until the expiration date of the applicable Contract Period, or Interest Period, as applicable.
- (k) Any Letter of Credit issued under the Existing Credit Agreement which remains outstanding on the Effective Date shall continue to be subject to participation fees and fronting fees determined in accordance with the Existing Credit Agreement until the expiration date of such Letter of Credit. If any such Letter of Credit is extended, such Letter of Credit shall be subject to participation fees and fronting fees under this Agreement.

(l) Notwithstanding anything to the contrary contained in this Agreement, if, as a result of any restatement or other adjustment to the financial statements delivered under this Agreement (including any adjustment to unaudited financial statements as a result of subsequent audited financial statements) or for any other reason (including without limitation an adjustment on any Borrowing Base Report delivered hereunder), the Borrower, the Agent or the Lenders determine that Adjusted Average Excess Availability as of any applicable date was inaccurate and, as a result of such inaccuracy, the Applicable Margin applicable to any Loans or any fees for any period was lower than would otherwise be the case had such inaccuracy not occurred, then the Borrower shall immediately and retroactively be obligated to pay to the Agent for the account of the applicable Lenders, promptly on demand by the Agent (or, if an Event of Default pursuant to any of Sections 7.1(i), (j) or (k) shall have occurred and be continuing, automatically and without further action by the Agent), an amount equal to the excess of the amount of interest and fees that should have been paid by the Borrower for such period over the amount of interest and fees actually paid by the Borrower for such period, plus interest on such amount at the rate otherwise applicable herein.

2.6 Termination and Reduction of Commitments.

- (a) Unless previously terminated, the Commitments shall terminate on the Maturity Date.
- (b) The Borrower may, upon five Business Days prior written notice to the Agent, permanently cancel any unused portion of the Commitments, provided, however, that the Borrower shall pay to the Agent, on the date on which such cancellation becomes effective, the Early Termination Fee in respect of the cancelled portion of the Commitments. Notwithstanding the foregoing, no Early Termination Fee shall be payable if the Borrower enters into financing arrangements with CIBC concurrently with the effectiveness of such cancellation. The Agent shall promptly notify each Lender of the receipt by the Agent of any such notice. Any such cancellation shall be applied rateably in respect of the Commitments of each Lender. Each notice delivered by the Borrower pursuant to this Section 2.6(b) shall be irrevocable.
- (c) Subject to the other terms and conditions of this Agreement and unless the Commitments have been earlier terminated, the Commitments shall be available hereunder until the Maturity Date.
- **2.7** <u>Repayment of Loans</u>. The Borrower hereby unconditionally promises to pay to the Agent for the account of each Lender the then unpaid principal amount of each Revolving Loan on the earlier of the Maturity Date and the date that the Commitment is terminated pursuant to Section 2.6(b) or Section 7.1.

2.8 Evidence of Debt.

- (a) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the Indebtedness of the Borrower to such Lender resulting from each Borrowing made by such Lender hereunder, including the amounts of principal and interest payable and paid to such Lender from time to time hereunder.
- (b) The Agent shall maintain accounts in which it shall record (i) the amount of each Borrowing made hereunder, the Type thereof and, in the cases of B/A Borrowings and LIBO Rate Loans, the relevant Contract Period or Interest Period, applicable thereto, (ii) the amount of any principal or interest due and payable or to become due and payable from the Borrower to each Lender hereunder, and (iii) the amount of any sum received by the Agent hereunder for the account of the Lenders and each Lender's share thereof.

- (c) The entries made in the accounts maintained pursuant to Sections 2.8(a) and (b) shall be conclusive evidence (absent manifest error) of the existence and amounts of the obligations recorded therein; provided that the failure of any Lender or the Agent to maintain such accounts or any error therein shall not in any manner affect the obligation of the Borrower to repay the Borrowings in accordance with the terms of this Agreement. In the event of a conflict between the records maintained by the Agent and any Lender, the records maintained by the Agent shall govern.
- (d) Any Lender may request that Loans (other than B/A Borrowings) made by it be evidenced by a promissory note. In such event, the Borrower shall prepare, execute and deliver to such Lender a promissory note payable to the order of such Lender (or, if requested by such Lender, to such Lender and its registered assigns) and in a form approved by the Agent. Thereafter, the Loans evidenced by such promissory note and interest thereon shall at all times be represented by one or more promissory notes in such form payable to the order of the payee named therein (or, if such promissory note is a registered note, to such payee and its registered assigns).

2.9 Prepayments.

- (a) <u>Mandatory Borrowing Base Prepayments</u>. If at any time the aggregate Exposure of all Lenders is in excess of the Borrowing Base, the Borrower shall immediately pay to the Agent, for the account of the Lenders, the amount of such excess to be applied (i) first, in satisfaction of all Reimbursement Obligations, if any, outstanding at such time, (ii) second, as a prepayment of the Revolving Loans, and (iii) third, as Cover for any remaining Letter of Credit Exposure, F/X Exposure and Bankers Acceptances in an amount of such remaining excess.
- (b) <u>Currency Fluctuations</u>. If, at any time, the Canadian \$ Equivalent of the Loans made by any Lender to the Borrower under any Credit exceeds the Commitment of such Lender under such Credit (any such excess being referred to in this Section as an "Excess Amount"), then the Borrower will repay to the Agent, for the account of each applicable Lender, an amount equal to the Excess Amount with respect to such Lender. If the amount of any Excess Amount with respect to any Lender is equal to or greater than 3% of the Commitment of such Lender for the applicable Credit, then the repayment of the Excess Amount to such Lender shall be made by the Borrower within one Business Day after the Agent requests such repayment. If the amount of any Excess Amount with respect to any Lender is less than 3% of the Commitment of such Lender for such credit, then the repayment of the Excess Amount to such Lender shall be made on the next Quarterly Date. The Agent shall request repayment of any Excess Amount forthwith upon request therefor by any Lender, but the Agent is not otherwise required to monitor Excess Amount levels or to request repayment thereof.
- (c) <u>Notice by Borrower</u>. Each prepayment notice provided by the Borrower hereunder in respect of any permanent repayment or prepayment hereunder shall be in a form acceptable to the Agent and shall be irrevocable.
- (d) <u>Notice by Agent</u>. Upon receipt of a notice of prepayment pursuant to this Section 2.9, the Agent shall promptly notify each applicable Lender of the contents thereof and of such Lender's Applicable Percentage of such prepayment and the corresponding Early Termination Fee, if any.

2.10 Fees.

- (a) The Borrower shall pay to the Agent for the account of and distribution to each Lender rateably in accordance with each such Lender's Applicable Percentage a standby fee for the period commencing on the date of this Agreement to and including the Maturity Date (or such earlier date as the Commitments shall have been terminated entirely) computed at a rate of 0.25% per annum on the average daily excess amount of the aggregate Commitments over the aggregate Exposure (but excluding, solely for the purpose of this Section 2.10, any F/X Exposure). The standby fees on the Commitments shall be calculated monthly in arrears on the last Business Day of each calendar month (and on the date on which the Commitments terminate) and each such calculated amount shall be payable on the first Business Day of the immediately following calendar month (or on the date on which the Commitments terminate, as the case may be). All standby fees shall be computed on the basis of a year of 365 or 366 days, as the case may be, and shall be payable for the actual number of days elapsed (including the first day but excluding the last day).
- (b) The Borrower agrees to pay to the Agent for the account of each Lender rateably in accordance with each Lender's Applicable Percentage, a guarantee fee (an "LC Guarantee Fee") with respect to the provision of Letter of Credit Guarantees, at the rate per annum equal to the Applicable Margin on the average daily amount of the Letter of Credit Exposure during the period from and including the date of this Agreement (or the date on which any Letter of Credit Exposure first exists to but excluding the latter of: (i) the date of termination of the Commitments and (ii) the date on which there ceases to be any Letter of Credit Exposure. All such LC Guarantee Fees (the "LC Fees") shall be calculated monthly in arrears on the last Business Day of each calendar month (and on the date on which the Commitments terminate) and each such calculated amount shall be payable on the first Business Day of the immediately following calendar month (or on the date on which the Commitments terminate, as the case may be); provided that all LC Fees, together with all Standard LC Fees (as defined below), accruing after the date on which the Commitments terminate shall be payable on demand. All LC Fees payable pursuant to this Section 2.10(b) shall be computed on the basis of a year of 365 or 366 days, as the case may be, and shall be payable for the actual number of days elapsed (including the first day but excluding the last day). The Borrower also agrees to pay to the Agent, for the account of the Issuing Bank, the Issuing Bank's standard fees (the "Standard LC Fees") with respect to the issuing, administration, handling, amendment, renewal or extension of any Letter of Credit or processing of drawings thereunder. Such standard fees shall be payable within 10 days after demand by the Agent or the Issuing Bank.
- (c) The Borrower agrees to pay to the Agent, for its own account, on the date of this Agreement and on the first Business Day of each calendar month thereafter an administrative management fee in an amount equal to Cdn.\$1,000, which the Borrower acknowledges and agrees shall be fully earned when paid.
- (d) The Borrower agrees to pay to the Agent, for its own account, the Agent's standard charges, fees, costs and expenses for its field examinations, verifications and audits in an amount equal to \$1,200 per person per day plus such field examiner's and auditor's reasonable out-of-pocket expenses.
- (e) The Borrower agrees to pay to the Agent, for its own account, fees payable in the amounts and at the times separately agreed upon in writing between the Borrower and the Agent.

(f) All fees payable hereunder shall be paid on the dates due, in immediately available funds, to the Agent, for its own account or for distribution to the Lenders. Fees paid shall not be refundable except in the case of manifest error in the calculation of any fee payment.

2.11 B/A Borrowings.

- (a) Subject to the terms and conditions of this Agreement, the Borrower may request a Borrowing by presenting drafts for acceptance and purchase as Bankers Acceptances by the Lenders.
- (b) No Contract Period with respect to a B/A Borrowing shall extend beyond the Maturity Date. The Borrower shall not be entitled to obtain or roll over any B/A Borrowings at any time that a Default or an Event of Default has occurred and is continuing.
- (c) To facilitate availment of B/A Borrowings, the Borrower hereby appoints each Lender as its attorney to sign and endorse on its behalf (in accordance with a Borrowing Request relating to a B/A Borrowing), in handwriting or by facsimile or mechanical signature as and when deemed necessary by such Lender, blank forms of Bankers Acceptances in the form requested by such Lender. In this respect, it is each Lender's responsibility to maintain an adequate supply of blank forms of Bankers Acceptances for acceptance under this Agreement. The Borrower recognizes and agrees that all Bankers Acceptances signed and/or endorsed by a Lender on behalf of the Borrower shall bind the Borrower as fully and effectually as if signed in the handwriting of and duly issued by the proper signing officers of the Borrower. Each Lender is hereby authorized (in accordance with a Borrowing Request relating to a B/A Borrowing) to issue such Bankers Acceptances endorsed in blank in such face amounts as may be determined by such Lender; provided that the aggregate amount thereof is equal to the aggregate amount of Bankers Acceptances required to be accepted and purchased by such Lender. No Lender shall be liable for any damage, loss or other claim arising by reason of any loss or improper use of any such instrument. Each Lender shall maintain a record with respect to Bankers Acceptances (i) received by it in blank hereunder, (ii) voided by it for any reason, (iii) accepted and purchased by it hereunder, and (iv) cancelled at their respective maturities. On request by or on behalf of the Borrower, a Lender shall cancel all forms of Bankers Acceptances which have been pre-signed or pre-endorsed on behalf of the Borrower and which are held by such Lender and are not required to be issued in accordance with the Borrower's irrevocable notice. Alternatively, the Borrower agrees that, at the request of the Agent, the Borrower shall deliver to the Agent a "depository note" which complies with the requirements of the Depository Bills and Notes Act (Canada), and consents to the deposit of any such depository note in the book-based debt clearance system maintained by the Canadian Depository for Securities.
- (d) Drafts of the Borrower to be accepted as Bankers Acceptances hereunder shall be signed as set out in this Section 2.11. Notwithstanding that any person whose signature appears on any Bankers Acceptances may no longer be an authorized signatory for any Lender or the Borrower at the date of issuance of a Bankers Acceptances, such signature shall nevertheless be valid and sufficient for all purposes as if such authority had remained in force at the time of such issuance and any such Bankers Acceptances so signed shall be binding on the Borrower.

- (e) Promptly following receipt of a Borrowing Request specifying a Borrowing by way of Bankers Acceptances, the Agent shall so advise the Lenders and shall advise each Lender of the aggregate face amount of the Bankers Acceptances to be accepted by it and the applicable Contract Period (which shall be identical for all Lenders). The aggregate face amount of the Bankers Acceptances to be accepted by the Lenders shall be in a minimum aggregate amount of Cdn.\$1,000,000 and shall be a whole multiple of Cdn.\$500,000, and such face amount shall be in the Lenders' pro rata portions of such Borrowing, provided that the Agent may in its sole discretion increase or reduce any Lender's portion of such B/A Borrowing to the nearest Cdn.\$100,000 without reducing the overall Commitments.
- (f) Upon acceptance of a Bankers Acceptance by a Lender, such Lender shall purchase, or arrange for the purchase of, each Bankers Acceptance from the Borrower at the Discount Rate for such Lender applicable to such Bankers Acceptance accepted by it and provide to the Agent the Discount Proceeds therefor for the account of the Borrower. The Acceptance Fee payable by the Borrower to a Lender under Section 2.5 in respect of each Bankers Acceptance accepted by such Lender shall be set off against the Discount Proceeds payable by such Lender under this Section 2.11.
- (g) Each Lender may at any time and from time to time hold, sell, rediscount or otherwise dispose of any or all Bankers Acceptances accepted and purchased by it.
- (h) If a Lender is not a chartered bank under the *Bank Act* (Canada) or if a Lender notifies the Agent in writing that it is otherwise unable to accept Bankers Acceptances, such Lender will, instead of accepting and purchasing Bankers Acceptances, make a Loan (a "B/A Equivalent Loan") to the Borrower in the amount and for the same term as the draft which such Lender would otherwise have been required to accept and purchase hereunder. Each such Lender will provide to the Agent the Discount Proceeds of such B/A Equivalent Loan for the account of the Borrower. Each such B/A Equivalent Loan will bear interest at the same rate which would result if such Lender had accepted (and been paid an Acceptance Fee) and purchased (on a discounted basis) a Bankers Acceptance for the relevant Contract Period (it being the intention of the parties that each such B/A Equivalent Loan shall have the same economic consequences for the Lenders and the Borrower as the Bankers Acceptance which such B/A Equivalent Loan replaces). All such interest shall be paid in advance on the date such B/A Equivalent Loan is made, and will be deducted from the principal amount of such B/A Equivalent Loan in the same manner in which the Discount Proceeds of a Bankers Acceptance would be deducted from the face amount of the Bankers Acceptance. Subject to repayment requirements, on the last day of the relevant Contract Period for such B/A Equivalent Loan, the Borrower shall be entitled to convert each such B/A Equivalent Loan into another type of Loan, or to roll over each such B/A Equivalent Loan into another B/A Equivalent Loan, all in accordance with the applicable provisions of this Agreement.
- (i) With respect to each B/A Borrowing, at or before 10:00 a.m. two Business Days before the last day of the Contract Period of such B/A Borrowings, the Borrower shall notify the Agent in writing, if the Borrower intends to issue Bankers Acceptances on such last day of the Contract Period to provide for payment of such maturing B/A Borrowing. If the Borrower fails to notify the Agent of its intention to issue Bankers Acceptances on such last day of the Contract Period, the Borrower shall provide payment to the Agent on behalf of the Lenders of an amount equal to the aggregate face amount of such B/A Borrowing on the last day of the Contract Period thereof. If the Borrower fails to make such payment, such maturing B/A Borrowing shall be deemed to have been converted on the last day of the Contract Period into a Canadian Prime Loan in an amount equal to the aggregate amount of such B/A Borrowing.

- (j) The Borrower waives presentment for payment and any other defence to payment of any amounts due to a Lender in respect of a Bankers Acceptances accepted and purchased by it pursuant to this Agreement which might exist solely by reason of such Bankers Acceptances being held, at the maturity thereof, by such Lender in its own right, and the Borrower agrees not to claim any days of grace if such Lender, as holder, sues the Borrower on the Bankers Acceptances for payment of the amount payable by the Borrower thereunder. On the last day of the Contract Period of a Bankers Acceptances, or such earlier date as may be required or permitted pursuant to the provisions of this Agreement, the Borrower shall pay the Lender that has accepted and purchased such Bankers Acceptances the full face amount of such Bankers Acceptances and, after such payment, the Borrower shall have no further liability in respect of such Bankers Acceptances and such Lender shall be entitled to all benefits of, and be responsible for all payments due to third parties under, such Bankers Acceptances.
- (k) If a Lender grants a participation in a portion of its rights under this Agreement to a Participant under Section 9.4(e), then, in respect of any B/A Borrowing, a portion thereof may, at the option of such Lender, be by way of Bankers Acceptance accepted by such Participant. In such event, the Borrower shall upon request of the Agent or the Lender granting the participation execute and deliver a form of Bankers Acceptances undertaking in favour of such Participant for delivery to such Participant.
- (l) Except as required by any Lender upon the occurrence of an Event of Default, no B/A Borrowing may be repaid by the Borrower prior to the expiry date of the Contract Period applicable to such B/A Borrowing; provided, however, that the Borrower may defease any B/A Borrowing by depositing with the Agent an amount that is sufficient to repay such B/A Borrowing on the expiry date of the Contract Period applicable to such B/A Borrowing.

2.12 Increased Costs; Illegality.

- (a) If any Change in Law shall:
 - (i) impose, modify or deem applicable any reserve, special deposit or similar requirement against assets of, deposits with or for the account of, or credit extended by, any Lender; or
 - (ii) impose on any Lender or the Issuing Bank or the London interbank market any other condition affecting this Agreement (including the imposition on any Lender of, or any change to, any Indemnified Tax or other charge with respect to its LIBO Rate Loans or any Letter of Credit or participation therein, or its obligation to make LIBO Rate Loans or any Letter of Credit);

and the result of any of the foregoing shall be to increase the cost to such Lender of making or maintaining any Loan (or of maintaining its obligation to make any such Loan) or to increase the cost to such Lender of participating in, issuing or maintaining any Letter of Credit or any Loan or to reduce the amount of any sum received or receivable by such Lender hereunder (whether of principal, interest or otherwise), then the Borrower will pay to such Lender, such additional amount or amounts as will compensate such Lender, for such additional costs incurred or reduction suffered.

- (b) If any Lender determines that any Change in Law regarding capital or liquidity requirements has or would have the effect of reducing the rate of return on such Lender's capital or liquidity as a consequence of this Agreement or the Loans made by, or participations in Letters of Credit held by such Lender, or the Letters of Credit issued by the Issuing Bank, to a level below that which such Lender or such Lender's holding company could have achieved but for such Change in Law (taking into consideration such Lender's policies and the policies of such Lender's holding company with respect to capital adequacy or liquidity) and such Lender's desired return on capital, then from time to time the Borrower will pay to such Lender, such additional amount or amounts as will compensate such Lender or such Lender's holding company for any such reduction suffered. Notwithstanding anything herein to the contrary, (a) all requests, rules, guidelines, requirements and directives promulgated by the Bank for International Settlements, and Basel Committee on Banking Supervision (or any successor or similar authority) or by United States, Canadian or foreign regulatory authorities, in each case pursuant to Basel III, and (b) the Dodd-Frank Wall Street Reform and Consumer Protection Act (United States) and all requests, rules, guidelines, requirements and directives thereunder or issued in connection therewith or in implementation thereof, shall in each case be deemed to be a Change in Law for purposes of this Section 2.12(b) regardless of the date enacted, adopted, issued or implemented.
- (c) A certificate of a Lender setting forth amount or amounts necessary to compensate such Lender as specified in Sections 2.12(a) or (b), together with a brief description of the Change of Law, shall be delivered to the Borrower, and shall be conclusive absent manifest error. In preparing any such certificate, a Lender shall be entitled to use averages and to make reasonable estimates, and shall not be required to "match contracts" or to isolate particular transactions. The Borrower shall pay such Lender the amount shown as due on any such certificate within 10 days after receipt thereof.
- (d) Failure or delay on the part of any Lender to demand compensation pursuant to this Section 2.12 shall not constitute a waiver of such Lender's right to demand such compensation.
- (e) In the event that any Lender shall have determined (which determination shall be reasonably exercised and shall, absent manifest error, be final, conclusive and binding upon all parties) at any time that the current or reasonably expected foreign currency markets are unusually unstable or that the making or continuance of any Loan denominated in a currency other than Canadian Dollars has become unlawful or materially restricted as a result of compliance by such Lender in good faith with any Applicable Law, or by any applicable guideline or order (whether or not having the force of law and whether or not failure to comply therewith would be unlawful), then, in any such event, such Lender shall give prompt notice (by telephone and confirmed in writing) to the Borrower and to the Agent of such determination (which notice the Agent shall promptly transmit to the other Lenders). Upon the giving of the notice to the Borrower referred to in this Section 2.12(e), the Borrower's right to request (by continuation, conversion or otherwise), and such Lender's obligation to make, Loans denominated in a currency other than Canadian Dollars shall be immediately suspended, and thereafter any requested Borrowing of Loans denominated in a currency other than Canadian Dollars shall, as to such Lender only, be deemed to be a request for a Canadian Prime Loan, and if the affected Loan or Loans are then outstanding, the Borrower shall immediately, or if permitted by Applicable Law, no later than the date permitted thereby, upon at least one Business Day prior written notice to the Agent and the affected Lender, convert each such Loan denominated in a currency other than Canadian Dollars into a Canadian Prime Loan, provided that if more than one Lender is affected at any time, then all affected Lenders must be treated the same pursuant to this Section 2.12(e).

- (f) If prior to the commencement of any Interest Period for a LIBO Rate Borrowing:
 - (i) the Agent determines (which determination shall be conclusive absent manifest error) that adequate and reasonable means do not exist for ascertaining the LIBO Rate for such Interest Period; or
 - (ii) the Agent is advised by the Required Lenders that the LIBO Rate for such Interest Period will not adequately and fairly reflect the cost to such Lenders of making or maintaining their Loans included in such Borrowing for such Interest Period;

then the Agent shall give notice thereof to the Borrower and the Lenders by telephone or telecopy as promptly as practicable thereafter and, until the Agent notifies the Borrower and the Lenders that the circumstances giving rise to such notice no longer exist, (i) any Interest Election Request that requests the conversion of any Borrowing to, or continuation of any Borrowing as, a LIBO Rate Borrowing shall be ineffective, and (ii) if any Borrowing Request requests a LIBO Rate Borrowing, such Borrowing shall be made as a Base Rate Borrowing; provided that (A) if the circumstances giving rise to such notice do not affect all the Lenders, then requests by the Borrower for LIBO Rate Borrowings may be made to Lenders that are not affected thereby, and (B) if the circumstances giving rise to such notice affect only one Type of Borrowings, then the other Type of Borrowings shall be permitted.

2.13 Break Funding Payments. In the event of (a) the failure by the Borrower to borrow, convert, continue or prepay any Loan on the date specified in any notice delivered by the Borrower pursuant hereto, or (b) the payment or conversion of any principal of any B/A Borrowing or LIBO Rate Loan other than on the last day of a Contract Period or, as applicable, Interest Period applicable thereto (including as a result of an Event of Default), or (c) the prepayment or conversion of any B/A Borrowing or LIBO Rate Loan other than on the last day of the Interest Period applicable thereto, then, in any such event, the Borrower shall compensate each Lender for the loss, cost and expense attributable to such event. A certificate of any Lender setting forth any amount or amounts that such Lender is entitled to receive pursuant to this Section 2.13 shall be delivered to the Borrower and shall be conclusive absent manifest error. The Borrower shall pay such Lender the amount shown as due on any such certificate within 10 days after receipt thereof.

2.14 Taxes.

(a) Any and all payments by or on account of any obligation of the Borrower hereunder shall be made free and clear of and without deduction or withholding for any Indemnified Taxes; provided that if the Borrower shall be required to deduct or withhold any Indemnified Taxes from such payments, then (i) the sum payable shall be increased as necessary so that, after making all required deductions or withholdings (including deductions or withholdings applicable to additional sums payable under this Section 2.14), the Agent, or Lender (as the case may be) receives an amount equal to the sum it would have received had no such deduction or withholding been made, (ii) the Borrower shall make such deduction or withholding, and (iii) the Borrower shall pay to the relevant Governmental Authority in accordance with Applicable Law the full amount deducted or withheld.

- (b) In addition to the payments by the Borrower required by Section 2.14(a), the Borrower shall pay any and all present or future stamp or documentary Taxes or any other excise or property Taxes, charges or similar levies arising from any payment made hereunder or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement to the relevant Governmental Authority in accordance with Applicable Law.
- (c) The Borrower shall indemnify the Agent, and each Lender, within 10 days after written demand therefor, for the full amount of any Indemnified Taxes paid by the Agent, such Lender, as the case may be, on or with respect to any payment by or on account of any obligation of the Borrower hereunder (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section 2.14) and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the Borrower by a Lender, or by the Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.
- (d) As soon as practicable after any payment of Indemnified Taxes by the Borrower to a Governmental Authority, the Borrower shall deliver to the Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Agent.
- (e) If, following the imposition of any Taxes on any payment by the Borrower to any Lender in respect of which the Borrower is required to make an additional payment pursuant to this Section 2.14, any Lender receives or is granted a credit against or remission for or deduction from or in respect of any Taxes paid by it or shall obtain any other relief which, in such Lender's opinion, is both reasonably identifiable and quantifiable by it without involving it in an unacceptable administrative burden (any of the foregoing being a "saving"), such Lender will reimburse the Borrower with such amount as such Lender shall have concluded, in its absolute discretion but in good faith, to be the amount or value of the relevant saving. Nothing herein contained shall interfere with the right of any Lender to arrange its affairs in whatever manner it thinks fit and, in particular, no Lender shall be under any obligation to claim relief for tax purposes on its corporate profits or otherwise, or to claim such relief in priority to any other claims, reliefs, credits or deductions available to it or to disclose details of its affairs. Each Lender will notify the Borrower promptly of the receipt by such Lender of any such saving and of such Lender's opinion as to the amount or value thereof, and any reimbursement to be made by such Lender will be made promptly on the date of receipt of such saving by such Lender or, if later, on the last date on which the applicable taxation authority would be able in accordance with applicable law to reclaim or reduce such saving.

2.15 Payments Generally; Pro Rata Treatment; Sharing of Set-offs.

(a) The Borrower shall make each payment required to be made by it hereunder (whether of principal, interest, fees or amounts payable in respect of Reimbursement Obligations, amounts payable under any of Sections 2.12, 2.13 or 2.14, or amounts otherwise payable hereunder) prior to 1:00 p.m., Toronto time, on the date when due, in immediately available funds, without set-off or counterclaim. Any amounts received after such time on any date may, in the discretion of the Agent, be deemed to have been received on the next succeeding Business Day for purposes of calculating interest thereon. All such payments shall be made to the Agent at the Payment Office, except that payments pursuant to any indemnities contained herein shall be made directly to the Persons entitled thereto. The Agent shall distribute any such payments received by it for the account of any other Person to the appropriate recipient promptly following receipt thereof. If any payment hereunder shall be due on a day that is not a Business Day, the date for payment shall be extended to the next succeeding Business Day, and, in the case of any payment accruing interest, interest thereon shall be payable for the period of such extension, provided that, in the case of any payment with respect to a LIBO Rate Loan, the date for payment shall be advanced to the next preceding Business Day if the next succeeding Business Day is in a subsequent calendar month. All payments under this Section 2.15 in respect of LIBO Rate Loans and Base Rate Loans shall be made in U.S. Dollars. All other payments under this Section 2.15 shall be made in Canadian Dollars. The Borrower hereby authorizes the Agent to debit the Operating Account to effect any payment due to the Lenders or the Agent pursuant to this Agreement. Any resulting overdraft in such account shall be payable by the Borrower to the Agent in same day funds.

- (b) Unless an Event of Default has occurred and is continuing (in which case, Section 7.2(d) shall apply), if at any time insufficient funds are received by and available to the Agent to pay fully all amounts of principal, interest, fees, amounts payable in respect of Repayment Obligations, amounts payable under any of Sections 2.12, 2.13 or 2.14 and other amounts payable hereunder, any available funds shall be applied (i) first, towards payment of interest and fees then due hereunder, rateably among the parties entitled thereto in accordance with the amounts of interest and fees then due to such parties, (ii) second, towards payment of principal and in satisfaction of Reimbursement Obligations then due hereunder, rateably among the parties entitled thereto in accordance with the amounts of principal and Reimbursement Obligations, and (iii) third, towards payment of amounts payable under any of Sections 2.12, 2.13 or 2.14 and other amounts otherwise payable hereunder.
- (c) If any Lender shall, by exercising any right of set-off or counterclaim or otherwise, obtain payment in respect of any principal of or interest on or fees in respect of any of its Revolving Loans or its share of Reimbursement Obligations resulting in such Lender receiving payment of a greater proportion of the aggregate amount of any principal of or interest on or fees in respect of any of its Revolving Loans or participations in Reimbursement Obligations than the proportion to which it is entitled, then the Lender receiving such greater proportion shall purchase (for cash at face value) participations in the Revolving Loans or participations in Reimbursement Obligations owed to other Lenders (as the case may be) to the extent necessary so that the benefit of all such payments shall be shared by the Lenders rateably taking into account each of the Applicable Percentages in respect of each Lender; provided that (i) if any such participations are purchased and all or any portion of the payment giving rise thereto is recovered, such participations shall be rescinded and the purchase price restored to the extent of such recovery, without interest, and (ii) this Section 2.15(c) shall not apply to any payment made by the Borrower pursuant to and in accordance with the express terms of this Agreement or any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans or participations in Reimbursement Obligations to any assignee or participant, other than to the Borrower or any Subsidiary or Affiliate thereof (as to which the provisions of this paragraph shall apply). The Borrower consents to the foregoing and agrees, to the extent it may effectively do so under Applicable Law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against the Borrower rights of set-off and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of the Borrower in the amount of such participation.
- (d) Unless the Agent shall have received written notice from the Borrower prior to the date on which any payment is due to the Agent for the account of the Lenders hereunder that the Borrower will not make such payment, the Agent may assume that the Borrower has made such payment on such date in accordance herewith and may, in reliance upon such assumption, distribute to the Lenders, the amount due. In such event, if the Borrower has not in fact made such payment, then each of the Lenders, severally agrees to repay to the Agent forthwith on demand the amount so distributed to such Lender with interest thereon, for each day from and including the date such amount is distributed to it to but excluding the date of payment to the Agent, at the applicable rate for Canadian Prime Loans (if such amount is denominated in Canadian Dollars) or the applicable rate for Base Rate Loans (if such amount is denominated in U.S. Dollars).

- (e) If any Lender shall fail to make any payment required to be made by it pursuant to Section 2.15(d), then the Agent may, in its discretion (notwithstanding any contrary provision hereof), apply any amounts thereafter received by the Agent for the account of such Lender to satisfy such Lender's obligations under such Section 2.15(d) until all such unsatisfied obligations are fully paid.
- (f) Nothing in this Agreement shall be deemed to obligate any Lender to obtain the funds for any Loan in any particular place or manner or to constitute a representation by any Lender that it has obtained or will obtain the funds for any Loan in any particular place or manner.
- 2.16 Currency Indemnity. If, for the purposes of obtaining judgment in any court in any jurisdiction with respect to this Agreement or any other Loan Document, it becomes necessary to convert into a particular currency (the "Judgment Currency") any amount due under this Agreement or under any other Loan Document in any currency other than the Judgment Currency (the "Currency Due"), then conversion shall be made at the rate of exchange prevailing on the Business Day before the day on which judgment is given. For this purpose "rate of exchange" means the rate at which the Agent is able, on the relevant date, to purchase the Currency Due with the Judgment Currency in accordance with its normal practice at its head office in Toronto, Ontario. In the event that there is a change in the rate of exchange prevailing between the Business Day before the day on which the judgment is given and the date of receipt by the Agent of the amount due, the Borrower will, on the date of receipt by the Agent, pay such additional amounts, if any, or be entitled to receive reimbursement of such amount, if any, as may be necessary to ensure that the amount received by the Agent on such date is the amount in the Judgment Currency which when converted at the rate of exchange prevailing on the date of receipt by the Agent is the amount then due under this Agreement or such other Loan Document in the Currency Due. If the amount of the Currency Due which the Agent is so able to purchase is less than the amount of the Currency Due originally due to it, the Borrower shall indemnify and save the Agent and the Lenders harmless from and against all loss or damage arising as a result of such deficiency. This indemnity shall constitute an obligation separate and independent from the other obligations contained in this Agreement and the other Loan Documents, shall give rise to a separate and independent cause of action, shall apply irrespective of any indulgence granted by the Agent from time to time and shall continue in full force and effect notwithstanding any judgment or order for a liquidated sum in respect of an amount due under this Agreement or any other Loan Document or under any judgment or order.

2.17 Collection of Accounts.

(a) The Borrower shall, and shall cause each other Credit Party to, at its expense, enforce, collect and receive all amounts owing on its Accounts in the ordinary course of its business and any proceeds it so receives shall be subject to the terms hereof. Any proceeds received by a Credit Party in respect of Accounts, and any cheques, cash, credit card sales and receipts, notes or other instruments or property received by a Credit Party with respect to any Collateral, shall be held by such Credit Party in trust for the Agent, separate from such Credit Party's own property and funds, and promptly turned over to the Agent with proper assignments or endorsements by deposit to the Blocked Accounts.

- (b) The Borrower shall, and shall cause each other Credit Party to at the request of the Agent: (i) indicate on all of its invoices that funds should be delivered to and deposited in a lock box or a Blocked Account; (ii) direct all of its account debtors to deposit any and all proceeds of Collateral into the lock boxes or the Blocked Accounts; (iii) irrevocably authorize and direct any bank which maintains any Credit Party's initial receipt of cash, cheques and other items to promptly wire transfer all available funds to a Blocked Account; and (iv) advise all such banks of the Agent's security interest in such funds. The Borrower shall, and shall cause each other Credit Party to, provide the Agent with prior written notice of any and all deposit accounts opened or to be opened subsequent to the Original Effective Date. All amounts received by the Agent in payment of Accounts will be credited to the Operating Account when the Agent is advised by its bank of its receipt of "collected funds" at the Agent's bank account in Toronto, Ontario on the Business Day of such advise if advised no later than 12:00 noon, Toronto time, or on the next succeeding Business Day if so advised after 12:00 noon, Toronto time. No cheques, drafts or other instrument received by the Agent shall constitute final payment to the Agent unless and until such instruments have actually been collected.
- (c) The Borrower shall, and shall cause each other Credit Party to, establish and maintain, in its own respective name and at its expense, deposit accounts and lock boxes with such banks as are acceptable to the Agent (the "Blocked Accounts") into which the Borrower shall promptly cause to be deposited: (i) all proceeds of Collateral received by any Credit Party, including all amounts payable to any Credit Party from credit card issuers and credit card processors, and (ii) all amounts on deposit in deposit accounts used by any Credit Party at each of its locations, all as further provided in Section 2.17(b). The banks at which the Blocked Accounts are established and the applicable Credit Parties shall enter into three-party agreements, in form and substance satisfactory to the Agent (the "Blocked Account Agreements"), providing that, among other things, all cash, cheques and items received or deposited in the Blocked Accounts are subject to Liens in favour of the Agent, that the depository bank has no Lien upon, or right of set off against, the Blocked Accounts and any cash, cheques, items, wires or other funds from time to time on deposit therein, except as otherwise provided in the Blocked Account Agreements, and that on a daily basis the depository bank will wire, or otherwise transfer, in immediately available funds, all funds received or deposited into the Blocked Accounts to such bank account as the Agent may from time to time designate for such purpose. The Borrower hereby confirms and agrees that all amounts deposited in such Blocked Accounts and any other funds received and collected by the Agent, whether as proceeds of Inventory or other Collateral or otherwise, shall be subject to the Liens in favour of the Agent. The Blocked Accounts Agreements will be "springing dominion agreements" and the Agent will be entitled to exercise full dominion and control over the blocked accounts at any time when Excess Availability is less than the Threshold Amount.
- (d) The parties hereto hereby acknowledge, confirm and agree that the implementation of the cash management arrangements is a contractual right provided to the Agent and the Lenders hereunder in order for the Agent and the Lenders to manage and monitor their collateral position and not a proceeding for enforcement or recovery of a claim, or pursuant to, or an enforcement of, any security or remedies whatsoever, that the cash management arrangements contemplated herein are critical to the structure of the lending arrangements contemplated herein, that the Lenders are relying on the Borrower's acknowledgement, confirmation and agreement with respect to such cash management arrangements in making accommodations of credit available to the Borrower and in particular that any accommodations of credit are being provided by the Lenders to the Borrower strictly on the basis of a borrowing base calculation to fully support and collateralize any such accommodations of credit hereunder.

- **2.18** Letters of Credit. In order to assist the Borrower in establishing Letters of Credit with the Issuing Bank, the Borrower has requested the Agent and the Agent has agreed to execute a Letter of Credit Guarantee subject to the following terms and conditions:
- (a) Within the limits of the Commitments and the Borrowing Base, and the other limitations contained in this Agreement, the Agent shall assist the Borrower in obtaining Letters of Credit, denominated in Canadian Dollars or U.S. Dollars, in an amount not to exceed the outstanding amount of the Letter of Credit Sub-Line. The Agent's assistance for amounts in excess of the limitation set forth herein shall at all times and in all respects be in the Agent's sole discretion. It is understood that the term, form and purpose of each Letter of Credit and all documentation in connection therewith, and any amendments, modifications or extensions thereof, must be mutually acceptable to the Agent, the Issuing Bank and the Borrower and it is further understood the term of each Letter of Credit issued cannot be for a term that extends beyond the Maturity Date. Any and all outstanding Letter of Credit Guarantees shall be reserved dollar for dollar from the Borrowing Base as an Availability Reserve. Upon the expiry of a Letter of Credit Guarantee, amounts reserved as an Availability Reserve in respect of such Letter of Credit Guarantee, as the case may be, shall no longer be reserved from the Borrowing Base as an Availability Reserve.
- (b) The Agent shall have the right, without notice to the Borrower, to charge the Borrower's Operating Account with the amount of any and all indebtedness, liability or obligation of any kind incurred by the Agent under the Letter of Credit Guarantees at the earlier of (a) payment by the Agent under the Letter of Credit Guarantee; or (b) the occurrence and continuance of an Event of Default. Any amount so charged to the Borrower's Operating Account shall be deemed a Canadian Prime Rate Loan or a US Base Rate Loan hereunder, depending on the currency of the Borrower's payment obligation thereunder, and shall incur interest at the rate provided in Section 2.5.
- (c) The Borrower unconditionally indemnifies the Agent and holds the Agent harmless from any and all loss, claim or liability incurred by the Agent arising from any transactions or occurrences relating to Letters of Credit established or opened for the Borrower's account, the collateral relating thereto and any drafts or acceptances thereunder, and all Obligations thereunder, including any such loss or claim due to any errors, omissions, negligence, misconduct or action taken by any Issuing Bank, other than for any such loss, claim or liability arising out of the gross negligence or willful misconduct by the Agent under the Letter of Credit Guarantee. This indemnity shall survive termination of this Agreement. The Borrower agrees that any charges incurred by the Agent for the Borrower's account from the Issuing Bank shall be conclusive upon the Agent and may be charged to the Borrower's Operating Account.
- (d) The Agent shall not be responsible for: (a) the existence, character, quality, quantity, condition, packing, value or delivery of the goods purporting to be represented by any documents; (b) any difference or variation in the character, quality, quantity, condition, packing, value or delivery of the goods from that expressed in the documents; (c) the validity, sufficiency or genuineness of any documents or of any endorsements thereon, even if such documents should in fact prove to be in any or all respects invalid, insufficient, fraudulent or forged; (d) the time, place, manner or order in which shipment is made; partial or incomplete shipment, or failure or omission to ship any or all of the goods referred to in the Letters of Credit or documents; (e) any deviation from instructions; (f) delay, default, or fraud by the shipper and/or anyone else in connection with the goods or the shipping thereof; or (g) any breach of contract between the shipper or vendors and the Borrower.

- (e) The Borrower agrees that any action taken by the Agent, if taken in good faith, or any action taken by any Issuing Bank, under or in connection with the Letters of Credit, the Letter of Credit Guarantees, the drafts or acceptances, or the Collateral, shall be binding on the Borrower and shall not result in any liability whatsoever of the Agent to the Borrower. In furtherance thereof, the Agent shall have the full right and authority to: (a) clear and resolve any questions of non compliance of documents; (b) give any instructions as to acceptance or rejection of any documents or goods; (c) execute any and all steamship or airways guarantees (and applications therefor), indemnities or delivery orders; (d) grant any extensions of the maturity of, time of payment for, or time of presentation of, any drafts, acceptances, or documents; and (e) agree to any amendments, renewals, extensions, modifications, changes or cancellations of any of the terms or conditions of any of the applications, Letters of Credit, drafts or acceptances or Letters of Credit Guarantees; all in the Agent's sole discretion. The Issuing Bank shall be entitled to comply with and honor any and all such documents or instruments executed by or received solely from the Agent, all without any consent from the Borrower. In addition, without the Agent's express consent and endorsement in writing, the Borrower agrees: (a) not to (i) execute any applications for steamship or airway guarantees, indemnities or delivery orders; (ii) grant any extensions of the maturity of, time of payment for, or time of presentation of, any drafts, acceptances or documents; or (iii) agree to any amendments, renewals, extensions, modifications, changes or cancellations of any of the terms or conditions of any of the applications, Letters of Credit, drafts or acceptances or Letters of Credit Guarantees; and (b) upon the occurrence and during the continuance of an Event of Default, not to (i) clear and resolve any questions of non compliance of documents, or (ii) give any instructions as to acceptances or rejection of any documents or goods.
- (f) The Borrower shall, and shall cause each other Credit Party to: (a) procure any necessary import, export or other licenses or certificates for the import or handling of the Collateral; (b) comply with all Applicable Law in regard to the shipment and importation of the Collateral, or the financing thereof; and (c) deliver to the Agent any certificates in that regard that the Agent may at any time request to be furnished. In connection herewith, the Borrower warrants and represents that all shipments made under any such Letters of Credit are in accordance with Applicable Law of the countries in which the shipments originate and terminate, and are not prohibited by any such Applicable Law. The Borrower assumes all risk, liability and responsibility for, and agrees to pay and discharge, all present and future local, provincial, state, federal or foreign Taxes, duties, or levies with respect to such Collateral. Any embargo, restriction, laws, customs or regulations of any country, state, city, or other political subdivision, where the Collateral is or may be located, or wherein payments are to be made, or wherein drafts may be drawn, negotiated, accepted, or paid, shall be solely the Borrower's risk, liability and responsibility.
- (g) Upon any payments made to the Issuing Bank under the Letter of Credit Guarantee, the Agent shall acquire by subrogation, any rights, remedies, duties or obligations granted or undertaken by the Borrower to the Issuing Bank in any application for Letters of Credit, any standing agreement relating to Letters of Credit or otherwise, all of which shall be deemed to have been granted to the Agent and apply in all respects to the Agent and shall be in addition to any rights, remedies, duties or obligations contained herein.

- 2.19 F/X Contracts. Subject to Sections 4.1 and 4.2, the Borrower may request F/X Contracts in accordance with this Section 2.19:
- (a) Within the limits of the Commitments and the Borrowing Base and the other limitations as contained in this Agreement, the Agent may assist the Borrower on an uncommitted basis to obtain F/X Contracts in an amount such that the F/X Exposure does not exceed the outstanding amount of the F/X Contract Sub-Line. The Agent's assistance in obtaining the F/X Contracts and the F/X Bank's agreement to provide the F/X Contracts shall at all times and in all respects be in the Agent's and F/X Bank's sole discretion. The entry into F/X Contracts for amounts in excess of the limitation set forth herein shall at all times and in all respects be in the Agent's sole discretion. Any F/X Contract will be documented by separate documentation in the form required by the F/X Bank. The term, form and purpose of the F/X Contract and all confirmations and other documentation in connection therewith, and any amendments, modifications or extensions thereof, must be mutually acceptable to the F/X Bank and the Borrower.
- (b) The Agent shall have the right, without notice to the Borrower, to charge the Borrower's loan account with the amount of any and all indebtedness, liability or obligation of any kind incurred by the Agent or the F/X Bank under any F/X Contract at such time which is the earlier of (a) payment by the Agent under the F/X Contract; or (b) the occurrence and continuance of an Event of Default, unless the Borrower has provided Cover to the Agent. Any amount charged to Borrower's loan account shall be deemed a Canadian Prime Loan or a Base Rate Loan hereunder, depending on the currency of the Borrower's payment obligation in respect of such F/X Contract, and shall incur interest at the rate provided in Section 2.5.
- (c) Each of the Credit Parties unconditionally agrees to indemnify the Agent and the F/X Bank and holds the Agent and the F/X Bank harmless from any and all loss, claim or liability incurred by the Agent or the F/X Bank arising from any transactions or occurrences relating to F/X Contracts, the collateral relating thereto, and all Obligations thereunder, including any such loss or claim due to any errors, omissions, negligence, misconduct or action taken by the F/X Bank, other than for any such loss, claim or liability arising out of the gross negligence or wilful misconduct of the Agent or the F/X Bank, as applicable. This indemnity shall survive termination of this Agreement. The Borrower agrees that any charges incurred by the Agent or the F/X Bank, as applicable, are for the Borrower's account and may be charged to the Borrower's loan account.
- (d) Each of the Credit Parties agrees that any action taken by the Agent, if taken in good faith, or any action taken by the F/X Bank, under or in connection with the F/X Contracts or the Collateral, shall be binding on the Credit Parties and shall not result in any liability whatsoever of the Agent or any Lender to any Credit Party.
- (e) All rights, remedies, duties and obligations of the Credit Parties in respect of F/X Contracts shall be secured by the Liens arising under the Security Documents.

ARTICLE 3 REPRESENTATIONS AND WARRANTIES

In order to induce the Agent and the Lenders to enter into this Agreement, to make any Loans hereunder and to issue any Letters of Credit hereunder and to permit the Borrower to obtain F/X Contracts, each Credit Party hereby represents and warrants to the Agent and each Lender that each statement set forth in this Article 3 is true and correct on the date hereof, and will be true and correct on the date of each Borrowing, on the date each Letter of Credit is requested hereunder and on the date each Letter of Credit is issued hereunder:

- **3.1** Organization; Powers. The Borrower and each other Credit Party is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization, has all requisite power and authority to carry on its business as now and formerly conducted and, except where the failure to do so, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, is qualified to do business in, and is in good standing in, every jurisdiction where such qualification is required.
- **3.2** <u>Authorization; Enforceability</u>. The Transactions are within each Credit Party's corporate or partnership powers and have been duly authorized by all necessary corporate and, if required, shareholder action. This Agreement and the other Loan Documents have been duly executed and delivered by the Borrower and each other Credit Party party thereto and constitute legal, valid and binding obligations of the Borrower and each other Credit Party party thereto, enforceable in accordance with their terms, subject to applicable bankruptcy, insolvency, reorganisation, moratorium or other Applicable Laws affecting creditors' rights generally and subject to general principles of equity, regardless of whether considered in a proceeding in equity or at law.
- **3.3** Governmental Approvals; No Conflicts. The Transactions (a) do not require any consent or approval of, registration or filing with, or any other action by, any Governmental Authority, except as disclosed in Schedule 3.3, (b) will not violate any Applicable Law or the charter, by-laws or other organizational documents of the Borrower or any other Credit Party or any order of any Governmental Authority, (c) will not violate or result in a default under any indenture, agreement or other instrument binding upon the Borrower or any other Credit Party or their respective assets, or give rise to a right thereunder to require any payment to be made by the Borrower or any other Credit Party, and (d) will not result in the creation or imposition of any Lien on any asset of the Borrower or any other Credit Party, except for any Lien arising in favour of the Agent, for the benefit of the Lenders, under the Loan Documents.

3.4 Financial Condition; No Material Adverse Effect.

- (a) The Borrower has furnished to the Lenders its consolidated balance sheets and statements of income, retained earnings and changes in financial position as of and for the portion of the Fiscal Year ended December 31, 2017. Such financial statements present fairly, in all material respects, the consolidated financial position and results of operations and cash flows of the Borrower as of such dates and for such periods in accordance with GAAP.
- (b) Since December 31, 2017, there has been no event, development or circumstance that has had or could reasonably be expected to have a Material Adverse Effect.

(c) All information (including that disclosed in all financial statements) pertaining to the Borrower and its Subsidiaries (other than projections) (in this Section 3.4(c), the "Information") that has been or will be made available to the Lenders, or the Agent by the Borrower or any representative of the Borrower and the other Credit Parties, taken as a whole, is or will be, when furnished, complete and correct in all material respects and does not or will not, when furnished, contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements contained therein not materially misleading in light of the circumstances under which such statements are made. The projections that have been or will be made available to the Lenders, or the Agent by the Borrower or any representative of the Borrower have been or will be prepared in good faith based upon reasonable assumptions.

3.5 Litigation.

- (a) Except as disclosed in Schedule 3.5, there are no actions, suits, counterclaims or proceedings (including any Tax-related matter) by any Person or investigation by any Governmental Authority pending against or, to the knowledge of the Borrower, threatened against or affecting the Borrower or any of the other Credit Parties (i) as to which there is a reasonable possibility of an adverse determination and that, if adversely determined, could reasonably be expected, individually or in the aggregate, to result in a Material Adverse Effect (other than the Disclosed Matters), or (ii) that involve this Agreement, any other Loan Document, or the Transactions.
- (b) Since the date of this Agreement, there has been no change in the status of the Disclosed Matters that, individually or in the aggregate, has resulted in, or materially increased the likelihood of, a Material Adverse Effect.
- **3.6** Compliance with Applicable Laws and Agreements. The Borrower and each other Credit Party is in compliance with all Applicable Laws applicable to it or its property and all indentures, agreements and other instruments binding upon it or its property, except where the failure to do so, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect. Neither the Borrower nor any other Credit Party has violated or failed to obtain any Authorization necessary to the ownership of any of its property or assets or the conduct of its business, which violation or failure could reasonably be expected to have (in the event that such a violation or failure were asserted by any Person through appropriate action) a Material Adverse Effect.
- **3.7 Ownership**. As at the date of this Agreement, the registered and beneficial holders of all of the Equity Securities of the Borrower are as set out on Schedule 3.7.
- **3.8 Taxes**. Other than as disclosed in Schedule B, the Borrower and each other Credit Party has timely filed or caused to be filed all Tax returns and reports required to have been filed and has paid or caused to be paid all Taxes required to have been paid by it (including all instalments with respect to the current period) and has made adequate provision for Taxes for the current period, except Taxes that are being contested in good faith by appropriate proceedings and for which the Borrower or such other Credit Party, as applicable, has set aside on its books adequate reserves.
- **3.9** <u>Titles to Real Property</u>. The Borrower and each other Credit Party have indefeasible fee simple title to their respective owned real properties (or in Quebec, immoveable properties), and with respect to leased real properties, indefeasible title to the leasehold estate with respect thereto, pursuant to valid and enforceable leases, free and clear of all Liens except Permitted Liens, including the Liens disclosed to the Lenders in Schedule 3.9.

- **3.10** <u>Titles to Personal Property</u>. The Borrower and each other Credit Party have title to their respective owned personal properties (or, in Quebec, moveable properties), and with respect to leased personal properties, title to the leasehold estate with respect thereto, pursuant to valid and enforceable leases, free and clear of all Liens except Permitted Liens, including the Liens disclosed in Schedule 3.10.
- **3.11 Pension Plans.** Schedule 3.11 lists all Pension Plans and Benefit Plans maintained or contributed to by each Credit Party as at December 31, 2017, and indicates, for each Pension Plan, whether such Pension Plan is a defined benefits plan ("Defined Benefits Plan") or a defined contribution plan. Since December 31, 2017, no Credit Party has established any new Defined Benefits Plan. Except for employees of any Credit Party who were, as of December 31, 2008, a beneficiary of a Defined Benefits Plan, no employee of any Credit Party (including, for greater certainty, any new employee of any Credit Party who became employed after such date) has, is or will be entitled to become a beneficiary of a Defined Benefits Plan. Other than as set forth in Schedule 3.11, no Credit Party has converted any Defined Benefits Plan to a defined contribution plan which is continuing. The Pension Plans are duly registered under the ITA and any other Applicable Laws which require registration, have been administered in accordance with the ITA and such other Applicable Laws and no event has occurred which could reasonably be expected to cause the loss of such registered status, except to the extent that any failure to do so could not reasonably be expected to have a Material Adverse Effect. All material obligations of the Borrower and each other Credit Party (including fiduciary, funding, investment and administration obligations) required to be performed in connection with the Pension Plans and the funding agreements therefor have been performed on a timely basis, except to the extent that any failure to do so could not reasonably be expected to have a Material Adverse Effect. There are no inactive plans or outstanding disputes concerning the assets of the Pension Plans or any benefit plans. No promises of benefit improvements under the Pension Plans or any benefit plans have been made except where such improvement could not reasonably be expected to have a Material Adverse Effect. All contributions or premiums required to be made or paid by the Borrower and each other Credit Party to the Pension Plans or any benefit plans have been made on a timely basis in accordance with the terms of such plans and all Applicable Laws. There have been no improper withdrawals or applications of the assets of the Pension Plans or any benefit plans. Except as disclosed in Schedule 3.11, as of the date hereof, each of the Pension Plans is fully funded on a solvency basis and going concern basis (using actuarial methods and assumptions which are consistent with the valuations last filed with the applicable Governmental Authorities and which are consistent with GAAP). Neither the Borrower, nor any Credit Party is subject to the United States Employee Retirement Income Security Act of 1974, as amended. Subject to the matters disclosed in Schedule 3.11:
- (a) for any Pension Plan or fund, and for any other employee benefit plan which is a defined contribution plan requiring the Borrower or each other Credit Party to contribute thereto, or to deduct from payments to any individual and pay such deductions into or to the credit of such Pension Plan or fund, all required employer contributions have been properly withheld by the Borrower or such Subsidiary and fully paid into the funding arrangements for the applicable Pension Plan or fund,
- (b) except as otherwise set forth in Schedule 3.11, for any Defined Benefits Plan, in each case of the Borrower or any other Credit Party: (A) each such Pension Plan or fund or Defined Benefits Plan is fully funded on both a solvency basis and a going concern basis, (B) the most recent actuarial valuations in respect thereof are disclosed in Schedule 3.11, (C) no material changes have occurred since the date of such actuarial valuations which could reasonably be expected to materially adversely affect the conclusions of the actuary concerning the funding of any Defined Benefits Plan, and (D) all payments and contributions required to be remitted or paid to or in respect of each such Pension Plan or fund or Defined Benefits Plan, including special payments and any other payments in respect of any funding deficiencies or shortfalls, have been remitted or paid to or in respect of each such plan in a timely fashion, in accordance in all material respects with the terms of the plan and all Applicable Law, and

- (c) any assessments owed to any government-sponsored pension benefits guarantee fund, or other assessments or payments required under similar legislation in any other jurisdiction, have been paid when due.
- **3.12** <u>Disclosure</u>. The Borrower has disclosed to the Lenders all agreements, instruments and corporate or other restrictions to which it or any other Credit Party is subject, and all other matters known to it, that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect.
- **3.13** <u>Defaults</u>. Neither the Borrower nor any other Credit Party is in default nor has any event or circumstance occurred which, but for the passage of time or the giving of notice, or both, would constitute a default (in any respect that would have a Material Adverse Effect) under any loan or credit agreement, indenture, mortgage, deed of trust, security agreement or other instrument or agreement evidencing or pertaining to any Indebtedness of or Lien against the Borrower or any other Credit Party, or under any material agreement or instrument to which the Borrower or any other Credit Party is a party or by which the Borrower or any other Credit Party is bound, except as disclosed to the Lenders in Schedule 3.13. No Default has occurred and is continuing.
- **3.14** <u>Casualties; Taking of Properties</u>. Since December 31, 2017, neither the business nor the properties of the Borrower or any other Credit Party have been affected in a manner that has had, or could reasonably be expected to have, a Material Adverse Effect as a result of any fire, explosion, earthquake, flood, drought, windstorm, accident, embargo, requisition or taking of property or cancellation of contracts, permits or concessions by any domestic or foreign Governmental Authority, riot, activities of armed forces, or acts of God or of any public enemy.
- **3.15** Subsidiaries. As of the date of this Agreement, Schedule 3.15 correctly sets forth the (i) names, (ii) form of legal entity, (iii) Equity Securities issued and outstanding, (iv) Equity Securities owned by the Borrower or a Subsidiary of the Borrower (and specifying such owner), and (v) jurisdictions of organization of all Subsidiaries of the Borrower. Except as described in Schedule 3.15, as of the date of this Agreement, the Borrower does not own any Equity Securities or debt security which is convertible, or exchangeable, for Equity Securities of any other Person. Unless otherwise indicated in Schedule 3.15, as of the date of this Agreement, all of the outstanding Equity Securities of each Restricted Subsidiary is owned of record and beneficially by the Borrower, there are no outstanding options, warrants or other rights to purchase Equity Securities of any such Restricted Subsidiary, and all such Equity Securities so owned are duly authorized, validly issued, fully paid and non-assessable, and were issued in compliance with all applicable federal, provincial or foreign securities and other Applicable Laws, and are free and clear of all Liens, except for Permitted Liens.

- **3.16** <u>Insurance</u></u>. All policies of fire, liability, workers' compensation, casualty, flood, business interruption and other forms of insurance owned or held by the Borrower or any other Credit Party are (a) sufficient for compliance with all requirements of Applicable Law and of all agreements to which the Borrower or any other Credit Party is a party, (b) are valid, outstanding and enforceable policies, (c) provide adequate insurance coverage in at least such amounts and against at least such risks (but including in any event public liability) as are usually insured against in the same general area by Persons engaged in the same or a similar business to the assets and operations of the Borrower and each other Credit Party, and (d) will not in any way be adversely affected by, or terminate or lapse by reason of, the Transactions. All such material policies are in full force and effect, all premiums with respect thereto have been paid in accordance with their respective terms, and no notice of cancellation or termination has been received with respect to any such policy. Neither the Borrower nor any other Credit Party maintains any formalized self-insurance program with respect to its assets or operations or material risks with respect thereto. The certificate of insurance delivered to the Agent pursuant to Section 4.1(f) contains an accurate and complete description of all material policies of insurance owned or held by the Borrower and each other Credit Party on the date of this Agreement that relate to the Collateral and business interruption.
- **3.17 Solvency**. Neither the Borrower nor any other Credit Party is an "insolvent person" within the meaning of the BIA.
- **3.18** Material Contracts. Schedule C sets out all Material Contracts as of the date of this Agreement. A true and complete copy of each Material Contract has been delivered to the Agent as of the date of this Agreement. Each of the Material Contracts is in full force and effect. Neither the Borrower nor any Restricted Subsidiary is in default under or in breach of any term or condition of any Material Contract that would have, either individually or in the aggregate, a Material Adverse Effect, nor is the Borrower or any Restricted Subsidiary aware of any default under or breach of any term or condition of any Material Contract by any other party thereto that would have a Material Adverse Effect. No contract to which the Borrower or any Restricted Subsidiary is a party contains any material provisions which impose burdensome or onerous obligations on the Borrower or any Restricted Subsidiary which are inconsistent with prudent commercial activity by the Borrower or any Restricted Subsidiary.
- **3.19** Environmental Matters. Except as disclosed to the Lenders in the Disclosed Matters or set out in the Phase I Environmental Report:
- (a) <u>Environmental Laws</u>. Neither any property of the Borrower or its Subsidiaries nor the operations conducted thereon violate any applicable order of any court or Governmental Authority or any Environmental Laws, which violation could reasonably be expected to result in remedial obligations having a Material Adverse Effect, assuming disclosure to the applicable Governmental Authority of all material relevant facts, conditions and circumstances, if any, pertaining to the relevant property.
- (b) <u>Notices and Permits</u>. All notices, permits, licenses or similar authorizations, if any, required to be obtained or filed by the Borrower or its Subsidiaries in connection with the operation or use of any and all property of the Borrower or its Subsidiaries, including but not limited to past or present treatment, transportation, storage, disposal or Release of Hazardous Materials into the environment, have been duly obtained or filed, except to the extent the failure to obtain or file such notices, permits, licenses or similar authorizations could not reasonably be expected to have a Material Adverse Effect, or which could not reasonably be expected to result in remedial obligations having a Material Adverse Effect, assuming disclosure to the applicable Governmental Authority of all material relevant facts, conditions and circumstances, if any, pertaining to the relevant property.

- (c) <u>Hazardous Substances Carriers</u>. All Hazardous Materials generated at any and all property of the Borrower or its Subsidiaries have been treated, transported, stored and disposed of only in accordance with all Environmental Laws applicable to them, except to the extent the failure to have such Hazardous Materials transported, treated or disposed by such carriers could not reasonably be expected to have a Material Adverse Effect, and only at treatment, storage and disposal facilities maintaining valid permits under applicable Environmental Laws, which carriers and facilities have been and are operating in compliance with such permits, except to the extent the failure to have such Hazardous Materials treated, transported, stored or disposed at such facilities, or the failure of such carriers or facilities to so operate, could not reasonably be expected to have a Material Adverse Effect or which could not reasonably be expected to result in remedial obligations having a Material Adverse Effect, assuming disclosure to the applicable Governmental Authority of all material relevant facts, conditions and circumstances, if any, pertaining to the relevant property.
- (d) <u>Hazardous Materials Disposal</u>. The Borrower and its Subsidiaries have taken all reasonable steps necessary to determine and have determined that no Hazardous Materials have been disposed of or otherwise released and there has been no threatened Release of any Hazardous Materials on or to any property of the Borrower or its Subsidiaries other than in compliance with Environmental Laws, except to the extent the failure to do so could not reasonably be expected to have a Material Adverse Effect or which could not reasonably be expected to result in remedial obligations having a Material Adverse Effect, assuming disclosure to the applicable Governmental Authority of all material relevant facts, conditions and circumstances, if any, pertaining to the relevant property.
- (e) No Contingent Liability. The Borrower and its Subsidiaries have no material contingent liability in connection with any Release or threatened Release of any Hazardous Materials into the environment other than such contingent liabilities which could not reasonably be expected to result in remedial obligations having a Material Adverse Effect, assuming disclosure to the applicable Governmental Authority of all relevant facts, conditions and circumstances, if any, pertaining to such Release or threatened Release.

3.20 Employee Matters.

- (a) Except as set forth on Schedule 3.20 as of the date of this Agreement, none of the Borrower or any of the other Credit Parties, nor any of their respective employees, is subject to any collective bargaining agreement. There are no strikes, slowdowns, work stoppages or controversies pending or, to the best knowledge of the Borrower, threatened against the Borrower or any other Credit Party, or their respective employees, which could reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect.
- (b) Each of the Borrower and the other Credit Parties has withheld from each payment to each of their respective officers, directors and employees the amount of all Taxes, including income tax, pension plan, unemployment insurance and other payments and deductions required to be withheld therefrom, and has paid the same to the proper taxation or other receiving authority in accordance with Applicable Law. None of the Borrower nor any other Credit Party is subject to any claim by or liability to any of their respective officers, directors or employees for salary (including vacation pay) or benefits which would rank in whole or in part *pari passu* with or prior to the Liens created by the Security Documents, other than Permitted Liens to the extent reserved for as Priority Payables.

- **3.21 Fiscal Year**. The Fiscal Year of the Borrower ends on December 31 of each calendar year.
- 3.22 Intellectual Property Rights. The Borrower and each Restricted Subsidiary is the registered and beneficial owner of, with good and marketable title, free of all licenses, franchises and Liens other than Permitted Liens, to all patents, patent applications, trade marks, trade mark applications, trade names, service marks, copyrights, industrial designs, integrated circuit topographies, or other rights with respect to the foregoing and other similar property, used in or necessary for the present and planned future conduct of its business, without any conflict with the rights of any other Person, other than as listed on Schedule 3.22, or other than for such conflicts as could not reasonably be expected to have a Material Adverse Effect. All material patents, trade marks, trade names, service marks, copyrights, industrial designs, integrated circuit topographies, and other similar rights owned or licensed by the Borrower or any Restricted Subsidiary, and all rights of the Borrower and each Restricted Subsidiary to the use of any patents, trade marks, trade names, service marks, copyrights, industrial designs, integrated circuit topographies, or other similar rights, the absence of which could have a Material Adverse Effect (and other than "off the shelf" software) are described in Schedule 3.22 (collectively, the "Intellectual Property Rights"). Except as set forth in Schedule 3.22, no material claim has been asserted and is pending by any Person with respect to the use by the Borrower or any Restricted Subsidiary of any intellectual property or challenging or questioning the validity, enforceability or effectiveness of any intellectual property necessary for the conduct of the business of the Borrower or any Restricted Subsidiary. Except as disclosed in Schedule 3.22 or except as could not reasonably be expected to have a Material Adverse Effect, (i) the Borrower and each Restricted Subsidiary has the exclusive right to use the intellectual property which the Borrower (or each Restricted Subsidiary) owns, (ii) all applications and registrations for such intellectual property are current, and (iii) to the knowledge of the Borrower, the conduct of the Borrower's and each Restricted Subsidiary's business does not infringe the intellectual property rights of any other Person.
- **3.23 Residency of Borrower for Tax Purposes**. The Borrower is a resident of Canada for Canadian tax purposes.
- **3.24** <u>Distributions</u>. No Distribution has been declared, paid, or made upon or in respect of Equity Securities in any Credit Party since December 31, 2017, except in compliance with the Existing Credit Agreement.
- **3.25** <u>Debt</u>. None of the Credit Parties nor any of their Subsidiaries has any Indebtedness except (a) the Obligations, (b) the Indebtedness set forth in the most recent financial statements delivered to the Agent, or the notes thereto, (c) Tax obligations (including deferred Taxes), trade payables and other contractual obligations arising in the ordinary course of business as carried on by the Credit Parties and their Subsidiaries since the date of such financial statements, and (d) Indebtedness created in accordance with Section 6.1.
- **3.26** Workers' Compensation. None of the Credit Parties has any unpaid workers' compensation or like obligations except as are being incurred, and paid on a current basis in the ordinary course of business, and there are no proceedings, claims, actions, orders or investigations of any Governmental Authority relating to workers' compensation outstanding, pending or, to their knowledge, threatened relating to them or any of their employees or former employees which could reasonably be expected to have a Material Adverse Effect.

- **3.27 Bank Accounts**. Schedule 3.27 contains as of the date of this Agreement a complete and accurate list of all bank accounts maintained by the Credit Parties with any bank or other financial institution.
- **3.28** Real Property and Leases. Schedule 3.28 hereto is a correct and complete list, of all real property owned by each Credit Party, all leases and subleases of real property or movable or personal property by any Credit Party, as lessee or sublessee, and all leases and subleases of real property or movable or personal property by any Credit Party, as lessor or sublessor.

3.29 Further Real Property Matters.

- (a) Except as advised in writing to the Agent or as could not have a Material Adverse Effect, to the best of the knowledge of the Borrower, (i) no investigation or proceeding of any Governmental Authority is pending in respect of real property owned by any of the Credit Parties; and (ii) no part of any such real property has been condemned, taken or expropriated by any Governmental Authority, federal, state, provincial, municipal or any other competent authority.
- (b) Except as advised in writing to the Agent or as could not have a Material Adverse Effect, all present uses in respect of any real property of the Credit Parties may lawfully be continued and all permitted uses are satisfactory for the Credit Parties' current and intended purposes; and
 - (c) No Inventory is located at any leased real property of the Credit Parties except as indicated in Schedule 3.28.
- **3.30** <u>Jurisdictions of Credit Parties</u>. Schedule 3.30 sets out as at the date set forth therein the various jurisdictions in which the Borrower and each other Credit Party carries on business or has tangible assets having an aggregate value in excess of Cdn.\$250,000.
- **3.31** Anti-Corruption Laws and Sanctions. Each Credit Party has implemented and maintains in effect reasonable policies and procedures designed to ensure compliance by such Credit Party and its directors, officers, employees and agents with Anti-Corruption Laws and Sanctions. Each Credit Party and, to the knowledge of the Borrower, its directors, officers, employees and agents is in compliance with Anti-Corruption Laws and Sanctions. No Credit Party or, to the knowledge of the Borrower, any of its directors, officers or employees or agents is a Sanctioned Person or is engaged in any activity that would reasonably be expected to result in such Credit Party being designated as a Sanctioned Person. No Borrowing or Letter of Credit, use of proceeds or other transaction contemplated by this Agreement will violate Anti-Corruption Laws or Sanctions.

ARTICLE 4 CONDITIONS

- **4.1** Effective Date. The obligations of the Lenders to make Loans or provide a Letter of Credit Guarantee or to permit the Borrower to obtain an F/X Contract hereunder shall not become effective until the date on which each of the following conditions is satisfied (or waived in accordance with Section 9.2):
- (a) <u>Credit Agreement</u>. The Agent (or its counsel), each Lender, and the Issuing Bank shall have received from each party hereto either (i) a counterpart of this Agreement signed on behalf of each party hereto, or (ii) written evidence satisfactory to the Agent (which may include facsimile transmission of a signed signature page of this Agreement) that each such party has signed a counterpart of this Agreement.

(b) <u>Legal Opinions</u>. The Agent shall have received a favourable written opinion of counsel to the Borrower, and covering such matters relating to the Borrower, the Credit Parties, this Agreement, the other Loan Documents, or the Transactions as the Lenders shall reasonably request (together with copies of all factual certificates and legal opinions delivered to such counsel in connection with such opinion upon which counsel has relied). The Agent shall also have received favourable written opinions of such special and local counsel as may be required by the Agent (together with copies of all factual certificates and legal opinions delivered to such counsel in connection with such opinion upon which such counsel has relied). The Borrower hereby requests each such counsel to deliver such opinions and supporting materials. All opinions and certificates referred to in this Section 4.1(b) shall be addressed to the Agent and the Lenders and dated as of the date hereof.

(c) Corporate Certificates. The Agent shall have received:

- (i) certified copies of the resolutions of the Board of Directors of the General Partner, and any other Credit Party which is a party to any Loan Document, dated as of the date hereof, and approving, as appropriate, the Loans, this Agreement and the other Loan Documents, and all other documents, if any, to which the Borrower or such other Credit Party is a party and evidencing corporate authorization with respect to such documents; and
- (ii) a certificate of the Secretary or an Assistant Secretary of the Borrower, and any other Credit Party which is a party to any Loan Document, dated as of the date hereof, and certifying (A) the name, title and true signature of each officer of such Person authorized to execute this Agreement and the other Loan Documents to which it is a party, (B) the name, title and true signature of each officer of such Person authorized to provide the certifications required pursuant to this Agreement, including certifications required pursuant to Section 5.1 and Borrowing Requests, and (C) that attached thereto is a true and complete copy of the Certificate of Limited Partnership and Limited Partnership Agreement of the Borrower, and any other Credit Party which is a party to any Loan Document, as amended to date, and a recent certificate of status, certificate of compliance, good standing certificate or analogous certificate.
- (d) <u>Closing Conditions Certificate</u>. The Agent shall have received a certificate, dated as of the Effective Date and signed by a Responsible Officer of the Borrower, confirming compliance with the financial covenants set forth in Section 5.12 and with the conditions set forth in Section 4.2(a) and (b).
- (e) <u>Fees</u>. The Agent and the Lenders shall have received all fees and other amounts due and payable on or prior to the date of this Agreement, including, to the extent invoiced, reimbursement or payment of all legal fees and other Out-of-Pocket Expenses required to be reimbursed or paid by the Borrower hereunder, under the Fee Letter or under any other Loan Document.

- (f) <u>Insurance</u>. The Agent shall have received a certificate of insurance coverage, naming CIBC, as agent, as first loss payee in respect of the Collateral and in respect of business interruption insurance dated not more than 30 days prior to the date of this Agreement, evidencing that the Borrower and its Restricted Subsidiaries are carrying insurance in accordance with Section 5.9 hereof.
- (g) Inventory Control Systems; Appraisal; Field Audit; Opening Availability. The Agent shall have reviewed and be satisfied with the Collateral, the inventory control systems, the books and records and the reporting capability of the Credit Parties. The Agent shall have received appraisals, completed by a reputable and independent appraisal firm at the expense of the Borrower, determining the value-in-use of the real and personal property of the Credit Parties. In addition, the Agent shall have received the results of an updated field audit, and the Borrowing Base on the Effective Date shall be sufficient in value, as determined by Agent, to provide Borrower with a Borrowing Base, after giving effect to the extensions of credit to be made hereunder on the Effective Date (on a pro forma basis, with trade payables being paid currently, and expenses and liabilities being paid in the ordinary course of business and without acceleration of sales or deterioration of working capital) of at least Cdn.\$5,000,000.
 - (h) Fee Letter. The Agent shall have received the executed Fee Letter.
- (i) No Cessation of Financing Market. There shall have not been occurred and be continuing on the date of this Agreement any general banking moratorium or any practical cessation in the bank or private debt financing markets, and there shall not have been introduced any material governmental restrictions imposed on lending institutions, which materially affect the type of lending transactions contemplated by this Agreement.
- (j) Execution and Delivery of Documentation. The Borrower and any other Credit Party which is a party to any Loan Document shall have duly authorized, executed and delivered all documents, including Loan Documents, required hereunder, all in form and substance satisfactory to the Agent, acting reasonably, and all of the Security Documents shall have been registered in all offices in which, in the opinion of the Agent or its counsel, registration is necessary or of advantage to preserve the priority of the Liens intended to be created thereby, and duplicate copies of such Security Documents bearing or accompanied by appropriate endorsements or certificates of registration shall have been delivered to the Agent. The Agent shall have received and be satisfied with the results of all personal property, pending litigation, judgment, bankruptcy, execution and other searches conducted by the Agent and its counsel with respect to the Borrower and any other Credit Party in all jurisdictions selected by the Agent and its counsel.
 - (k) Security Documents. The Agent shall have received:
 - (i) [Intentionally Deleted];
 - (ii) a security agreement executed by each Credit Party in favour of the Agent, as agent for the Lenders, dated as of the Original Effective Date, constituting a first-priority Lien on all present and after acquired (a) accounts receivable, (b) inventory, (c) intangibles, (d) collateral proceeds accounts and (e) Equity Securities, of each Subsidiary of the Borrower, subject only to Permitted Liens;

- (iii) securities pledge agreements executed by each Credit Party in favour of the Agent, as agent for the Lenders, dated as of the Original Effective Date, together with all stock certificates, instruments and other documents required to be delivered to the Agent pursuant to such securities pledge agreements; and
- (iv) documents constituting security under section 427 of the *Bank Act* (Canada) granted by the Borrower and each other Credit Party entitled to grant such security to any Lender, including (A) a notice of intention to give security under section 427 of the *Bank Act* (Canada), (B) an assignment under section 427 of the *Bank Act* (Canada), and (C) an agreement respecting section 427 security (the "Bank Act Security").

provided that if any of the foregoing documents are not suitable for use in any jurisdiction, the applicable Credit Party shall provide to the Agent alternative document(s) with substantially equivalent substantive effect and which are suitable for use in such jurisdiction.

- (l) <u>Landlord Waivers</u>; <u>Bailee Letters</u>. The Agent shall have received (i) executed copies of a landlord waiver, in form and substance satisfactory to the Agent, acting reasonably, from each landlord of Real Property where any Collateral of any of the Credit Parties is located and (ii) bailee letters, in form and substance reasonably satisfactory to the Agent, from each bailee who is in possession of any Collateral of any of the Credit Parties.
- (m) <u>Regulatory Approval; Consents; Waivers</u>. The Agent and the Lenders shall be satisfied, acting reasonably, that all material Authorizations required in connection with the Transactions contemplated hereby have been obtained and are in full force and effect (including all approvals listed in Schedule 3.3), and that all consents and waivers required to consummate the Transactions have been obtained, to the extent that consummation of the Transactions would otherwise be restricted or prohibited under the terms of any Material Contract to which the Borrower or any other Credit Party is a party, or by which it is bound, in each case without the imposition of any burdensome provisions.
- (n) <u>Delivery of Financial Statements</u>. The Agent and the Lenders shall have received and be satisfied with the unaudited consolidated and consolidating balance sheets, statements of income and retained earnings and statements of changes in financial position of the Borrower and its Subsidiaries for the Fiscal Year ended December 31, 2017.
- (o) No Material Adverse Change. The Agent and the Lenders shall be satisfied that, since December 31, 2017, there has not been a Material Adverse Change.
- (p) Other Documentation. The Agent and the Lenders shall have received such other documents and instruments as are customary for transactions of this type or as they may reasonably request.
- (q) <u>Indebtedness</u>. The Transactions contemplated in this Agreement and the other Loan Documents shall not have caused any event or condition to occur which has resulted, or which will result, in any Material Indebtedness becoming due prior to its scheduled maturity or that permits (with or without the giving of notice, the lapse of time, or both) the holder or holders of any Material Indebtedness or any trustee or agent on its or their behalf to cause any Material Indebtedness to become due, or to require the prepayment, repurchase, redemption or defeasance thereof, prior to its scheduled maturity, or which will result in the creation of any Liens under any Indebtedness.

- (r) <u>Blocked Account/Cash Management Systems</u>. The Agent shall have received evidence satisfactory to the Agent that, as of the Original Effective Date, blocked account and cash management systems complying with Section 2.17 have been established and are currently being maintained in the manner set forth in such Section 2.17, and the Agent shall have received copies of duly executed triparty blocked account and other control agreements satisfactory to the Agent, acting reasonably, with the banks and other Persons as required hereunder or pursuant to Section 5.14.
- (s) <u>Material Contracts</u>. The Agent and the Lenders shall be satisfied with the terms and conditions of each of the Material Contracts.

(t) [Intentionally Deleted].

- (u) <u>Capitalization Arrangement</u>. The Lenders shall be satisfied with the capital structure of the Borrower, that the Borrower is solvent, and that the Borrower has sufficient working capital to pay its debts as they become due.
- (v) "Know Your Customer" Information. The Agent and the Lenders shall have received all documentation and other information required by bank regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including the *Proceeds of Crime (Money Laundering) and Terrorist Financing Act* (Canada).

The obligations of the Lenders to make Loans or provide a Letter of Credit Guarantee hereunder shall not become effective unless each of the foregoing conditions is satisfied (or waived pursuant to Section 9.2) at or prior to 3:00 p.m., Toronto time, on July 31, 2018 (and, in the event such conditions are not so satisfied or waived by such time, the Commitments shall terminate at such time). Upon the satisfaction (or waiver pursuant to Section 9.2) of each of the foregoing conditions, the Parent Guarantee will be released and terminated.

- **4.2** Each Credit Event. The obligation of each Lender to make a Loan or provide a Letter of Credit Guarantee or to permit the Borrower to obtain any F/X Contract on the occasion of any Borrowing, (including on the occasions of the initial Borrowings hereunder), is subject to the satisfaction of the following conditions:
- (a) the representations and warranties of the Borrower set forth in this Agreement shall be true and correct on and as of the date of each such Borrowing (including the date of issuance, amendment, renewal or extension of such Letter of Credit, as applicable) as if made on such date (except where such representation or warranty refers to a different date);
- (b) at the time of and immediately after giving effect to such Borrowing (including the issuance, amendment, renewal or extension of such Letter of Credit, as applicable), no Default shall have occurred and be continuing;
 - (c) the Agent shall have received a Borrowing Request in the manner and within the time period required by Section 2.3; and
- (d) except as may be otherwise agreed to from time to time by the Agent and the Borrower in writing, after giving effect to the extension of credit requested to be made by the Borrower on such date, the aggregate Exposure will not exceed the lesser of (i) the Commitments, or (ii) an amount equal to the Borrowing Base.

Each Borrowing, including each issuance, amendment, renewal or extension of a Letter of Credit, shall be deemed to constitute a representation and warranty by the Borrower on the date thereof as to the accuracy of the matters specified in paragraphs (a) and (b) above. This requirement does not apply on the conversion or rollover of an existing Borrowing provided that the aggregate outstanding Borrowings will not be increased as a consequence thereof.

ARTICLE 5 AFFIRMATIVE COVENANTS

From (and including) the Original Effective Date until the Commitments have expired or been terminated and the principal of and interest on each Loan and all fees payable hereunder shall have been paid in full and all Letters of Credit and Letter of Credit Guarantees and F/X Contracts shall have expired and been terminated and all Reimbursement Obligations have been satisfied by the Borrower, the Borrower and each other Credit Party covenants and agrees with the Lenders that:

5.1 Financial Statements and Other Information. The Borrower will furnish to the Agent and each Lender:

- (a) as soon as available and in any event within 120 days after the end of each Fiscal Year of the Borrower, its audited consolidated balance sheet and related statements of income, retained earnings and changes in financial position as of the end of and for such Fiscal Year, setting forth in each case in comparative form the figures for the previous Fiscal Year, all reported on by independent auditors of recognized national standing (without a "going concern" or like qualification or exception and without any qualification or exception as to the scope of such audit) to the effect that such consolidated financial statements present fairly in all material respects the financial condition and results of operations of the Borrower and its Subsidiaries on a consolidated basis in accordance with GAAP consistently applied;
- (b) as soon as available and in any event within 90 days after the end of each Fiscal Year of the Parent, its audited consolidated balance sheet and related statements of income, retained earnings and changes in financial position as of the end of and for such Fiscal Year, setting forth in each case in comparative form the figures for the previous Fiscal Year, all reported on by independent auditors of recognized national standing (without a "going concern" or like qualification or exception and without any qualification or exception as to the scope of such audit) to the effect that such consolidated financial statements present fairly in all material respects the financial condition and results of operations of the Parent and its Subsidiaries on a consolidated basis in accordance with GAAP consistently applied;
- (c) as soon as available and in any event within: (i) 30 days after the end of each calendar month, except in respect of its first, second and third Fiscal Quarters; and (ii) 45 days after the end of its first, second and third Fiscal Quarters, its unaudited consolidated balance sheet and related statements of income, retained earnings and changes in financial position as of the end of such month and the then elapsed portion of the Fiscal Year which includes such calendar month, setting forth in each case in comparative form the figures for the corresponding period or periods of (or, in the case of the balance sheet, as of the end of) the previous Fiscal Year, all certified by a Responsible Officer as presenting fairly in all material respects the financial condition and results of operations of the Borrower and its Subsidiaries on a consolidated basis in accordance with GAAP consistently applied, subject to normal year-end audit adjustments, together with a management discussion and analysis of results, if any;

- (d) concurrently with the financial statements required pursuant to Sections 5.1(a) and (c) above, a certificate of the Borrower, signed by a Responsible Officer (i) stating that a review of such financial statements during the period covered thereby and of the activities of the Borrower and the Subsidiaries has been made under such Responsible Officer's supervision with a view to determining whether the Borrower and the Subsidiaries have fulfilled all of their obligations under this Agreement and the other Loan Documents, (ii) stating that the Borrower and the Subsidiaries have fulfilled their obligations under this Agreement and the other Loan Documents and that all representations made in this Agreement continue to be true and correct as if made on the date of such certification (or specifying the nature of any change), except where such representation or warranty refers to a different date, or, if there shall be a Default or Event of Default, specifying the nature and status thereof and the Borrower's proposed response thereto, (iii) demonstrating in reasonable detail compliance (including showing all material calculations) as at the end of the most recently completed Fiscal Year or the most recently completed calendar month, as the case may be, with the covenants in Section 5.12, and (iv) containing or accompanied by such financial or other details, information and material as the Agent may reasonably request to evidence such compliance;
- (e) concurrently with the financial statements required pursuant to Section 5.1(a) and (c), a certificate of the Borrower, signed by a Responsible Officer, demonstrating in reasonable detail compliance (including showing all material calculations) as at the end of the most recently completed Fiscal Year or the most recently completed calendar month, as the case may be, with Section 6.3, describing by category (utilizing the same categories as are used by the Borrower in its internal financial reports) any Acquisitions permitted by Section 6.4 and any Capital Expenditures made by the Borrower or any Restricted Subsidiary as of the end of such Fiscal Year;
- (f) copies of each management letter issued to the Borrower by its auditors promptly following consideration or review thereof by the Board of Directors of the Borrower, or any committee thereof (together with any response thereto prepared by the Borrower);
- (g) promptly upon the request of the Agent, and in any event no less frequently than monthly, within 20 days of the last day of each calendar month, (together with a copy of all or any part of the following reports requested by any Lender in writing after the Original Effective Date), each of the following reports and certificates, each of which shall be prepared by the Borrower as of the last day of the immediately preceding calendar month (in a manner satisfactory to the Agent):
 - (i) a certified Borrowing Base Report, accompanied by such supporting detail and documentation as shall be requested by the Agent in its reasonable discretion;
 - (ii) a monthly accounts receivable aging (including both summary and detail format); showing Accounts outstanding aged from invoice date as follows: 1 to 30 days past due, 31 to 60 days past due, 61-90 days past due, and 91 days or more past due, accompanied by such supporting detail and documentation as shall be requested by the Agent in its reasonable discretion, including the ledger for disputed/legal accounts;

- (iii) a monthly accounts payable aging (including both summary and detail format when available);
- (iv) a reconciliation of Accounts aging to the general ledger and to the financial statement as at the month end;
- (v) a calculation of the Accounts which would not meet the criteria of an Eligible Account Receivable;
- (vi) a copy of the internally generated monthly sales, credit memo, debit memo journals and invoice register;
- (vii) a copy of the internally generated month end cash receipts and collections journal;
- (viii) a reconciliation of the cash receipts journal to the blocked depository account;
- (ix) an aged listing of the ten largest customer accounts for the month;
- (x) an aged listing of the ten largest accounts payable for the month;
- (xi) a detailed, monthly, Inventory listing based on a perpetual inventory listing system (as determined by the Agent in its audit process);
- (xii) a summary of Inventory of the Borrower by location and type with a supporting perpetual Inventory report, in each case, accompanied by such supporting detail and documentation as shall be requested by the Agent in its reasonable discretion; such summaries and reports shall include the dollar value thereof (both at cost, determined on a first in, first out basis, and at fair market value) by location and product group;
- (xiii) a reconciliation of the monthly inventory perpetual listing to the general ledger and to the financial statement as at month end;
- (xiv) a monthly general ledger/trial balance; and
- (xv) a calculation and report as to the Inventory which does not meet the definition of Eligible Inventory;
- (h) if, at any time, the Excess Availability is less than 10% of the Borrowing Base, as set out in the most recently delivered Borrowing Base Report in accordance with Section 5.1(g)(i) or Section 5.1(h)(i) or if an Event of Default has occurred and is continuing, promptly, and in any event no less frequently than weekly, on the 4th business day of the week (together with a copy of all or any part of the following reports requested by any Lender in writing after the Original Effective Date), each of the following reports and certificates, each of which shall be prepared by the Borrower as of the last day of the immediately preceding week (in a manner satisfactory to the Agent):
 - (i) a Borrowing Base Report (such weekly Borrowing Base Reports to include Inventory values set out in the most recent month end delivery unless requested otherwise by the Agent in consultation with the Borrower);

- (ii) a weekly accounts receivable aging (including both summary and detail format); showing Accounts outstanding aged from invoice date as follows: 1 to 30 days past due, 31 to 60 days past due, 61-90 days past due, and 91 days or more past due, accompanied by such supporting detail and documentation as shall be requested by the Agent in its reasonable discretion, including the ledger for disputed/legal accounts;
- (iii) a copy of the internally generated weekly sales journal, invoice register;
- (iv) a copy of the internally generated credit memo journal (if not included in the weekly sales journal);
- (v) a copy of the internally generated debit memo journal (if not included in the weekly sales journal);
- (vi) a copy of the internally generated weekly cash receipts and collections journal;
- (i) by the 25th day of each calendar month, as at the 15th day of the current calendar month, (until the Agent advises the Borrower otherwise, the Agent is requesting Inventory information on a twice-monthly basis, however, the Agent may, in its sole discretion, determine that twice-monthly inventory information is unnecessary and amend to monthly) a perpetual inventory listing, by location, which reflects any differences from the month end Borrowing Base submitted as to such Inventory;
- (j) such other reports designating, identifying and describing the Accounts as required by the Agent and on a more frequent basis as the Agent may reasonably request in its reasonable credit discretion;
- (k) such other reports designating, identifying and describing the Inventory as required by the Agent and on a more frequent basis as the Agent may reasonably request in its reasonable credit discretion;
- (l) at the time of delivery of each of the monthly financial statements delivered pursuant to Section 5.1(c), (i) a listing of Accounts the assignment of which would be subject to the *Financial Administration Act* (Canada) or the *Federal Assignment of Claims Act of 1940* of the United States of America or any similar law of any jurisdiction or subject to a limitation or restriction contained in the contract evidencing or relating to such Accounts and (ii) a list of any applications for the registration of any patent, trademark or copyright filed by the Borrower or any Subsidiary with the Canadian Intellectual Property Office or any similar office or agency in the prior calendar month;
- (m) the results of each physical verification, if any, that the Borrower may have made, or caused any other Person to have made on its behalf, of all or any portion of its Inventory, within 10 Business Days of completion of any such physical verification (and, if a Default or an Event of Default has occurred and be continuing, the Borrower shall, upon the request of the Agent, conduct, and deliver the results of, such physical verifications as the Agent may require);

- (n) at the cost of the Borrower, one appraisal every two Fiscal Years if at all times Borrowings are less than Cdn.\$10,000,000, with such appraisal determining the net orderly liquidation value of the Inventory of the Borrower and the Restricted Subsidiaries, such appraisal to be conducted by an appraiser that is acceptable to the Agent, and shall be in scope, form and substance acceptable to the Agent (and every Fiscal Year if at any time Borrowings are equal to or more than Cdn.\$10,000,000 and semi-annually if, at any time during a Fiscal Year, Excess Availability is less than 10% of the Borrowing Base, as set out in the most recently delivered Borrowing Base Report in accordance with Section 5.1(g)(i) or Section 5.1(h)(i), and without limit if an Event of Default shall have occurred and be continuing);
- (o) promptly after the Borrower learns of the receipt or occurrence of any of the following, a certificate of the Borrower, signed by a Responsible Officer, specifying (i) any official notice of any violation, possible violation, non-compliance or possible noncompliance, or claim made by any Governmental Authority pertaining to all or any part of the properties of the Borrower or any of the Subsidiaries which could reasonably be expected to have a Material Adverse Effect, (ii) any event which constitutes a Default or Event of Default, together with a detailed statement specifying the nature thereof and the steps being taken to cure such Default or Event of Default, (iii) the receipt of any notice from, or the taking of any other action by, the holder of any promissory note, debenture or other evidence of Indebtedness of the Borrower or any of the Subsidiaries in an amount in excess of Cdn.\$2,000,000 with respect to an actual or alleged default, together with a detailed statement specifying the notice given or other action taken by such holder and the nature of the claimed default and what action the Borrower or the relevant Subsidiary is taking or proposes to take with respect thereto, (iv) any default or non-compliance of any party to any of the Loan Documents with any of the terms and conditions thereof or any notice of termination or other proceedings or actions which could reasonably be expected to adversely affect any of the Loan Documents, (v) the creation, dissolution, merger or acquisition of any Restricted Subsidiary, (vi) any event or condition not previously disclosed to the Agent, which violates any Environmental Law and which could potentially, in the Borrower's reasonable judgment, have a Material Adverse Effect, (vii) any material amendment to, termination of, or material default under a Material Contract or any execution of, or material amendment to, termination of, or material default under, any material collective bargaining agreement, (viii) any other event, development or condition which may reasonably be expected to have a Material Adverse Effect, and (ix) any jurisdiction not listed in Schedule 3.30 where the Borrower or each Credit Party carries on business or has tangible assets having an aggregate value in excess of Cdn.\$250.000:
- (p) promptly after the occurrence thereof, notice of the institution of or any material adverse development in any action, suit or proceeding or any governmental investigation or any arbitration before any court or arbitrator or any Governmental Authority or official against the Borrower or any Restricted Subsidiary or any material property of any thereof which could reasonably be expected to have a Material Adverse Effect;
- (q) annually, together with its delivery pursuant to Section 5.1(a), copies of each annual and other report (including applicable schedules) with respect to each Pension Plan of the Borrower or any Restricted Subsidiary or any trust created thereunder, and copies of any "check-up" or other updated Pension Plan report prepared for internal purposes;

- (r) at the cost of the Borrower, once every two Fiscal Years if at all times Borrowings are less than Cdn.\$10,000,000 (and every Fiscal Year if at any time Borrowings are equal to or more than Cdn.\$10,000,000 and semi-annually if, at any time during a Fiscal Year, Excess Availability is less than 10% of the Borrowing Base, as set out in the most recently delivered Borrowing Base Report in accordance with Section 5.1(g)(i) or Section 5.1(h)(i), and without limit if a Default or Event of Default shall have occurred and be continuing), a report or reports of an independent collateral field examiner (which collateral field examiner may be the Agent or an Affiliate thereof) approved (i) by the Borrower, whose approval shall not be unreasonably withheld, and (ii) by the Agent with respect to the Eligible Accounts and Eligible Inventory components included in the Borrowing Base. The calculation of the foregoing maximum amount shall not include reasonable out-of-pocket expenses incurred in connection therewith, which shall also be for the account of the Borrower. The Agent may (and, at the direction of the Required Lenders, shall) request such reports or additional reports as it (or the Required Lenders) shall reasonably deem necessary;
- (s) upon request by the Agent, a summary of the insurance coverages of the Borrower and the other Credit Parties, in form and substance reasonably satisfactory to the Agent, and upon renewal of any insurance policy, a copy of an insurance certificate summarizing the terms of such policy, and upon request by the Agent, copies of the applicable policies;
- (t) on or before the earlier of the 10th day after approval by the Board of Directors of the Borrower and the 30th day before each Fiscal Year end, an annual budget of the Borrower and each Restricted Subsidiary, approved by the Board of Directors of the General Partner, setting forth in reasonable detail and on a monthly basis the projected revenues and expenses of the Borrower for the following Fiscal Year, and shall include the Capital Expenditure Plan, it being recognized by the Lenders that projections as to future results are not to be viewed as fact and that the actual results for the period or periods covered by such projections may differ from the projected results;
- (u) concurrently with any delivery of financial statements under Section 5.1 (a) or (b) above, a certificate of a Responsible Officer of the Borrower (i) stating whether any change in GAAP or in the application thereof has occurred since the date of the audited financial statements referred to in Section 5.1(a) and, if any such change has occurred, specifying the effect of such change on the financial statements accompanying such certificate, (ii) identifying all its Subsidiaries existing on the date of such certificate and indicating, for each such Subsidiary, whether such Subsidiary is a Restricted Subsidiary and whether such Subsidiary was formed or acquired since the end of the previous calendar month, (iii) identifying any parcels of real property or improvements thereto that have been acquired by any Credit Party since the end of the previous calendar month, and (iv) identifying any Permitted Acquisitions that have been completed since the end of the previous calendar month, including the date on which each such Permitted Acquisition was completed and the consideration therefor; and
- (v) promptly after receipt or delivery thereof, (i) a copy of any direction, order, notice, ruling or opinion that any Credit Party may receive from any applicable Governmental Authority with respect to any Pension Plan, and (ii) a copy of any material letter or correspondence between any Credit Party and any applicable Governmental Authority with respect to any Pension Plan.
- **5.2** Existence; Conduct of Business. The Borrower will, and will cause each Restricted Subsidiary to, do or cause to be done all things necessary to preserve, renew and keep in full force and effect its legal existence (subject only to Section 6.3), and obtain, preserve, renew and keep in full force and effect any and all rights, licenses, permits, privileges and franchises material to the conduct of its business.

- **5.3** Payment of Obligations. The Borrower will, and will cause each Restricted Subsidiary to, pay its obligations, including Tax liabilities, that, if not paid, could result in a Material Adverse Effect before the same shall become delinquent or in default, except where (a) the validity or amount thereof is being contested in good faith by appropriate proceedings, (b) the Borrower or such Restricted Subsidiary has set aside on its books adequate reserves with respect thereto in accordance with GAAP, and (c) the failure to make payment pending such contest could not reasonably be expected to result in a Material Adverse Effect.
- **5.4** <u>Maintenance of Properties</u>. The Borrower will, and will cause each Restricted Subsidiary to, keep and maintain all property material to the conduct of its business in good working order and condition, ordinary wear and tear excepted, except to the extent that the failure to do so could not reasonably be expected to have a Material Adverse Effect.
- **5.5** <u>Books and Records; Inspection Rights</u>. The Borrower will, and will cause Restricted Subsidiary to, keep proper books of record and account in which full, true and correct entries are made of all dealings and transactions in relation to its business and activities. The Borrower will, and will cause each Restricted Subsidiary to, permit any representatives designated by the Agent or any Lender, upon reasonable prior notice, to visit and inspect its properties, to examine and make extracts from its books and records, and to discuss its affairs, finances and condition with its officers and independent accountants, all at such reasonable times and as often as reasonably requested.
- **5.6** Compliance with Applicable Laws and Material Contracts. The Borrower will, and will cause each Restricted Subsidiary to, comply with all Applicable Laws and orders of any Governmental Authority applicable to it or its property and with all of its material contractual obligations, except where the failure to do so, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect. The Borrower shall not modify, amend or alter its Limited Partnership Agreement other than in a manner which does not adversely affect the rights of the Lenders or the Agent.
- **5.7** <u>Use of Proceeds and Letters of Credit</u>. The proceeds of the Revolving Loans will be used for working capital and other general corporate purposes of the Borrower. Letters of Credit will be issued only to support general corporate purposes of the Borrower.
- **5.8 Further Assurances**. The Borrower will and will cause each other Credit Party to cure promptly any defects in the execution and delivery of the Loan Documents, including this Agreement. Upon request, the Borrower will, at its expense, as promptly as practical, execute and deliver to the Agent, all such other and further documents, agreements and instruments (and cause each other Credit Party to take such action) in compliance with or performance of the covenants and agreements of the Borrower or any other Credit Party in any of the Loan Documents, including this Agreement, or to further evidence and more fully describe the Collateral, or to correct any omissions in any of the Loan Documents, or more fully to state the security obligations set out herein or in any of the Loan Documents, or to perfect, protect or preserve any Liens created pursuant to any of the Loan Documents, or to make any recordings, to file any notices, or obtain any consents, all as may be necessary or appropriate in connection therewith, in the judgment of the Agent, acting reasonably.

5.9 Insurance. The Borrower shall, and shall cause each other Credit Party to, maintain insurance on its property and assets under such policies of insurance, with such insurance companies, in such reasonable amounts and covering such insurable risks as are at all times reasonably satisfactory to the Agent. All such policies are subject to the rights of any holders of Permitted Liens holding claims senior to the Agent, to be made payable to the Agent, to the extent required herein, in case of loss, under a standard non contributory "mortgagee", "lender" or "secured party" clause and are to contain such other provisions as the Agent may require to fully protect the Agent's interest in the property and assets subject to the Liens in favour of the Agent and to any payments to be made under such policies. All original policies or true copies thereof are to be delivered to the Agent, premium prepaid, with a loss payable endorsement in respect of the Collateral and in respect of business interruption insurance in the Agent's favour, and shall provide for not less than thirty (30) days prior written notice to the Agent of the exercise of any right of cancellation. Upon the occurrence and continuance of an Event of Default which is not waived in writing by the Agent, the Agent shall, subject to the rights of any holders of Permitted Liens holding claims senior to the Agent, have the sole right, in the name of the Agent, the Borrower or any other applicable Credit Party, to file claims under any insurance policies in respect of the Collateral and in respect of business interruption insurance, to receipt and give acquittance for any payments that may be payable thereunder, and to execute any and all endorsements, receipts, releases, assignments, reassignments or other documents that may be necessary to effect the collection, compromise or settlement of any such claims under any such insurance policies. Upon the occurrence and during the continuance of an Event of Default, all insurance proceeds in respect of any Collateral shall be paid to the Agent. The Agent may apply such insurance proceeds to the Obligations in such manner as it may deem advisable in its sole discretion. In the event the Borrower fails to provide the Agent with timely evidence, acceptable to the Agent, of the maintenance of insurance coverage required pursuant to this Section 5.9, or in the event that any Credit Party fails to maintain such insurance, the Agent may purchase or otherwise arrange for such insurance, but at the Borrower's expense and without any responsibility on the Agent's part for: (i) obtaining the insurance; (ii) the solvency of the insurance companies; (iii) the adequacy of the coverage; or (iv) the collection of claims. The insurance acquired by the Agent may, but need not, protect the Borrower's or any other Credit Party's interest in the Collateral, and therefore such insurance may not pay claims which the Borrower may have with respect to the Collateral or pay any claim which may be made against the Borrower in connection with the Collateral. In the event the Agent purchases, obtains or acquires insurance covering all or any portion of the Collateral, the Borrower shall be responsible for all of the applicable costs of such insurance, including premiums, interest (at the applicable interest rate for Revolving Loans set forth in Section 2.5), fees and any other charges with respect thereto, until the effective date of the cancellation or the expiration of such insurance. The Agent may charge all of such premiums, fees, costs, interest and other charges to the Borrower's Operating Account. The Borrower hereby acknowledges that the costs of the premiums of any insurance acquired by the Agent may exceed the costs of insurance which the Borrower may be able to purchase on its own. In the event that the Agent purchases such insurance, the Agent will promptly, and in any event within fifteen (15) days, notify the Borrower of said purchase.

5.10 Operation and Maintenance of Property. The Borrower will, and will cause each other Credit Party to, manage and operate its business or cause its business to be managed and operated (i) in accordance with prudent industry practice in all material respects and in compliance in all material respects with the terms and provisions of all applicable licenses, leases, contracts and agreements, and (ii) in compliance with all Applicable Laws of the jurisdiction in which such businesses are carried on, and all Applicable Laws of every other Governmental Authority from time to time constituted to regulate the ownership, management and operation of such businesses, except in each instance where a failure to so manage and operate would not have a Material Adverse Effect.

- **5.11** <u>Additional Subsidiaries</u>; <u>Additional Liens</u>. If, at any time on or after the Original Effective Date, the Borrower or any Restricted Subsidiary creates or acquires an additional Subsidiary or in some other fashion becomes the holder of any Equity Securities of a new Subsidiary:
- (a) the Borrower will, and will cause any relevant Restricted Subsidiary, to immediately execute and deliver to the Agent a securities pledge agreement, in form and substance satisfactory to the Agent, granting a security interest in 100% of the Equity Securities of such new Subsidiary owned by the Borrower or such Restricted Subsidiary; and
- (b) to the extent permitted by Applicable Law, the Borrower will cause such new Subsidiary to immediately execute and deliver to the Agent (i) a Guarantee, and (ii) mortgages, security agreements and other security-related documents covering such new Subsidiary's property, all in form and substance satisfactory to the Agent, acting reasonably;

provided that, if such new Subsidiary does not have, but only for so long as such new Subsidiary does not have, assets of more than Cdn.\$1,000,000, the Borrower shall not be required to comply with this Section with respect to such new Subsidiary. If, at any time after the Original Effective Date, the Borrower or any Restricted Subsidiary makes a permitted acquisition with respect to a Person that is not a Restricted Subsidiary, the Borrower shall, and shall cause such Restricted Subsidiary, to execute and deliver to the Agent at the time of such permitted acquisition, additional pledge agreements granting a security interest in 100% of the Equity Securities of such Person owned by the Borrower or such Restricted Subsidiary, respectively. In connection with the execution and delivery of any guarantee, pledge agreement, mortgage, security agreement or related document pursuant to this Section, the Borrower will, or will cause the relevant Credit Party to, deliver to the Agent such corporate resolutions, certificates, legal opinions and such other related documents as shall be reasonably requested by the Agent and consistent with the relevant forms and types thereof delivered on the Original Effective Date or as shall be otherwise reasonably acceptable to the Agent. Each guarantee, pledge agreement, mortgage, security agreement, Bank Act Security and other document delivered pursuant to this Section shall be deemed to be a Security Document from and after the date of execution thereof.

- **5.12** <u>Financial Covenants</u>. If, at any time, an Event of Default has occurred or the Excess Availability is less than Cdn.\$5,000,000 (the "Threshold Amount"), and until the Excess Availability thereafter becomes greater than or equal to the Threshold Amount for a period of not less than 7 Business Days, the Borrower shall maintain a Fixed Charge Coverage Ratio of not less than 1.10:1.00 for each Rolling Period, provided that, if the Excess Availability becomes greater than or equal to the Threshold Amount for a period of not less than 7 Business Days immediately thereafter, upon each such occurrence, such Excess Availability greater than or equal to the Threshold Amount shall be deemed to have been in effect from the date such Threshold Amount was first re-established, solely for the purpose of allowing the Borrower to comply with Fixed Charge Coverage Ratio hereunder.
- **5.13** <u>Post Closing Undertakings</u>. Borrower will ensure that all post closing undertakings as set forth in Schedule 5.13 (collectively, the "Undertakings") have been satisfied within the time periods set forth therein and any failure to satisfy any of the Undertakings within the applicable time periods shall constitute an Event of Default.

- **5.14** Canadian Pension Plans. The Borrower and each other Credit Party shall administer any Canadian Pension Plans in accordance with the requirements of the applicable pension plan texts, funding agreements, any other documents governing such Canadian Pension Plans, the ITA and applicable federal or provincial pension benefits legislation except for any non-compliance which would not reasonably be expected to have a Material Adverse Effect. The Borrower shall, and shall cause the other Credit Parties and Affiliates to, promptly provide the Agent with any documentation relating to any of the Canadian Pension Plans as the Agent may reasonably request. The Borrower shall, and shall cause the other Credit Parties and Affiliates to, notify the Agent within thirty (30) days of: (i) a material increase in the obligations, liabilities and indebtedness of any of the Canadian Pension Plans; and (ii) commencing payment of contributions to a Canadian Pension Plan to which the Borrower had not previously been contributing.
- **5.15** Application under the CCAA. The Borrower acknowledges that its business and financial relationships with the Agent and the Lenders are unique from its relationship with any other of its creditors. The Borrower agrees that it shall not file any plan of arrangement under the Companies' Creditors Arrangement Act (the "CCAA Plan") which provides for, or would permit, directly or indirectly, the Agent or the Lenders to be classified in the same class with any other creditor of the Credit Parties for purposes of such CCAA Plan.

ARTICLE 6 NEGATIVE COVENANTS

From (and including) the Original Effective Date until the Commitments have expired or been terminated and the principal of and interest on each Loan and all fees payable hereunder shall have been paid in full and all Letters of Credit, F/X Contracts, Letter of Credit Guarantees and Reimbursement Obligations have been satisfied by the Borrower, the Borrower and each Credit Party covenants and agrees with the Lenders that:

- **6.1** <u>Indebtedness.</u> No Credit Party will, and will not permit any Restricted Subsidiary to, create, incur, assume or permit to exist any Indebtedness, except:
 - (a) any Indebtedness created hereunder;
- (b) any Indebtedness existing on the date hereof and set forth in Schedule 6.1 (including, any extensions or renewals but excluding any replacements of any such Indebtedness);
 - (c) the Parent Subordinated Debt;
 - (d) any Indebtedness of one Credit Party to another Credit Party;
 - (e) any Guarantee by a Credit Party of Indebtedness of any other Credit Party;
- (f) any Indebtedness of the Credit Parties incurred under Purchase Money Liens, Sale and Leasebacks or to Capital Lease Obligations in an aggregate amount not exceeding Cdn.\$5,000,000 for all Credit Parties;
- (g) any Indebtedness of any Person that becomes a Credit Party after the date hereof, provided that (i) such Indebtedness exists at the time such Person becomes a Credit Party and is not created in contemplation of or in connection with such Person becoming a Credit Party and is on terms reasonably satisfactory to the Lenders, and (ii) the aggregate principal amount of Indebtedness permitted by this clause (g) shall not exceed Cdn.\$4,000,000 at any time outstanding;

- (h) any Indebtedness in respect of trade letters of credit;
- (i) any Indebtedness in respect of Swap Agreements permitted hereunder, provided that the aggregate notional amounts under all such Swap Agreements shall not exceed Cdn.\$25,000,000;
- (j) annually, if no Default or Event of Default shall have occurred and be continuing and there is an Excess Availability equal to or greater than such amount, any Indebtedness of the Borrower to the Parent incurred on the Business Day immediately prior to the fiscal year end of the Lenders, such Indebtedness being permitted only up to and including the Business Day immediately following the fiscal year end of the Lenders;
- (k) any Indebtedness of the Borrower to the Parent (on terms and conditions satisfactory to the Agent and Lenders) in respect of funds used solely for the purpose of curing any Default occurring as a result of the Borrower being in breach of its Fixed Charge Coverage Ratio; and
- (l) other Indebtedness of the Borrower or a Credit Party, provided that (i) the aggregate amount of such other Indebtedness permitted by this clause (l) shall not exceed Cdn\$25,000,000 at any time outstanding, (ii) at the time of and immediately after giving effect to any incurrence of any such Indebtedness, no Event of Default shall have occurred and be continuing, (iii) such Indebtedness shall mature and be repayable only after the Maturity Date, and (iv) if such Indebtedness is secured, such security shall not constitute a Lien on the Accounts or Inventory of the Borrower or any other Credit Party, and shall be subject to a customary intercreditor agreement satisfactory to the Agent, acting reasonably, which would provide (among other things) that the Liens in favour of the Agent in respect of the assets other than the Accounts or Inventory of the Borrower or any other Credit Party are subordinated to the Liens securing such other Indebtedness, and would provide for customary access rights for the Agent.
- **6.2** <u>Liens</u>. No Credit Party will, and will not permit any Restricted Subsidiary to, create, incur, assume or permit to exist any Lien on any property or asset now owned or hereafter acquired by the Credit Party or any Restricted Subsidiary, or assign or sell any income or revenues (including Receivables) or rights in respect of any thereof, except Permitted Liens.

6.3 Fundamental Changes.

(a) No Credit Party will, and will not permit any Restricted Subsidiary to, merge into or amalgamate or consolidate with any other Person, or permit any other Person to merge into or amalgamate or consolidate with it, or sell, transfer, lease or otherwise dispose of (in one transaction or in a series of transactions) all or substantially all of its assets, or all or any of the Equity Securities of any of its Subsidiaries (in each case, whether now owned or hereafter acquired), or liquidate or dissolve, except that, if at the time thereof and immediately after giving effect thereto no Default shall have occurred and be continuing, (i) any Restricted Subsidiary may amalgamate with the Borrower, (ii) any Restricted Subsidiary may amalgamate with any other Restricted Subsidiary, (iii) any Restricted Subsidiary may sell, transfer, lease or otherwise dispose of its assets to the Borrower or to another Restricted Subsidiary, and (iv) any Restricted Subsidiary may liquidate or dissolve if the Borrower determines in good faith that such liquidation or dissolution is in the best interests of the Borrower and the Agent determines that such liquidation or dissolution is not disadvantageous to the Lenders; provided that any amalgamation pursuant to Sections 6.3(a)(i) or (ii) shall not be permitted unless permitted by Section 6.4 and unless the amalgamated corporation confirms to the Agent in writing that the amalgamated corporation is liable, by operation of law or otherwise, for the obligations of the Borrower under this Agreement.

- (b) No Credit Party will, and will not permit any Restricted Subsidiary to, engage to any material extent in any material business other than businesses of the type conducted by the Credit Party on the date of execution of this Agreement and businesses reasonably related thereto.
- **6.4** <u>Investments, Loans, Advances, Guarantees and Acquisitions</u>. No Credit Party will, and will not permit any Restricted Subsidiary to, purchase, hold or acquire (including pursuant to any amalgamation with any Person that was not a wholly-owned Restricted Subsidiary prior to such amalgamation) any Equity Securities, evidences of indebtedness or other securities (including any option, warrant or other right to acquire any of the foregoing) of, make or permit to exist any loans or advances to, Guarantee any obligations of, or make or permit to exist any investment or any other interest in, any other Person, or purchase or otherwise acquire (in one transaction or a series of transactions) any assets of any other Person or otherwise make an Acquisition, except:
 - (a) investments by a Credit Party in the Equity Securities of any other Credit Party;
 - (b) loans or advances made by one Credit Party to any other Credit Party;
 - (c) Guarantees constituting Indebtedness permitted by Section 6.1;
 - (d) Permitted Investments; and
 - (e) Investments permitted by Section 6.6(h).
- **6.5** Swap Agreements. No Credit Party will, and will not permit any Restricted Subsidiary to, enter into any Swap Agreement, except (i) Swap Agreements entered into to hedge or mitigate risks to which the Borrower or any other Credit Party has actual exposure (other than those in respect of Equity Securities, (ii) Swap Agreements entered into in order to effectively cap, collar or exchange interest rates (from fixed to floating rates, from one floating rate to another floating rate or otherwise) with respect to any interest-bearing liability or investment of the Borrower or any other Credit Party, and (iii) F/X Contracts entered into pursuant to Section 2.19.
- **6.6** <u>Restricted Payments</u>. No Credit Party will, and will not permit any Restricted Subsidiary to, declare, pay or make, or agree to pay or make, directly or indirectly, any Restricted Payment, except that, provided that Excess Availability is at all times equal to or greater than the Threshold Amount, and no Event of Default shall have occurred and be continuing before or immediately after any of the following, the Credit Parties may make the following Restricted Payments:
 - (a) the Borrower may declare and pay dividends with respect to its Equity Securities;

- (b) any Restricted Subsidiary may declare and pay dividends to the Borrower or any wholly-owned Restricted Subsidiary and any wholly-owned Restricted Subsidiary may redeem or repurchase its own Equity Securities;
- (c) the Borrower may make Restricted Payments pursuant to and in accordance with stock option plans, profit sharing plans and/or other benefit plans for management or executives of the Borrower and its Subsidiaries, provided that the aggregate amount of cash payments made by the Borrower and the Subsidiaries in any Fiscal Year pursuant to all such stock option plans, profit sharing plans and other compensation benefit plans shall not exceed Cdn.\$3,500,000;
- (d) the Borrower may pay to the Parent any amounts required to pay income taxes on earnings attributed from the Borrower's business to the Parent;
- (e) the Borrower may make payments of (i) interest under the Parent Subordinated Debt at any time and (ii) principal under the Parent Subordinated Debt;
 - (f) the Borrower may make any payment to the Parent which is permitted by any of Sections 6.7(d), (e) or (f);
 - (g) the Borrower may distribute any amount to the Parent in any Fiscal Year; and
- (h) the Borrower may make Investments in businesses which are related to the Borrower's business in an aggregate amount not exceeding \$10,000,000 in any Fiscal Year; provided that, immediately after giving effect to any such Investment, no Default shall have occurred and be continuing and Excess Availability shall be not less than 25% of the lesser of the Borrowing Base and the Commitment.

If at any time Excess Availability is below the Threshold Amount, the Borrower must maintain a Fixed Charge Coverage Ratio of at least 1.1:1.0 prior to and immediately after making or permitting any Credit Party to make a Restricted Payment.

6.7 Transactions with Affiliates. No Credit Party will, and will not permit any Restricted Subsidiary to, sell, lease or otherwise transfer any property or assets to, or purchase, lease or otherwise acquire any property or assets from, or otherwise engage in any other transactions with, any of its Affiliates, except (a) in the ordinary course of business at prices and on terms and conditions not less favourable to the Credit Party than could be obtained on an arm's-length basis from unrelated third parties, (b) transactions between or among the Credit Parties not involving any other Affiliate, (c) any Restricted Payment permitted by Section 6.6, (d) foreign exchange transactions with the Parent at market rates in the normal course of business and any agreements existing as at the Original Effective Date with the Parent provided no material change shall be permitted thereunder without consent of the Agent, (e) pursuant to the Parent Credit Agreement, and (f) pursuant to the Parent Pulp Agency Agreement. The foregoing restrictions shall not apply to: (i) the payment of reasonable and customary fees to directors of the Credit Party, (ii) any other transaction with any employee, officer or director of a Credit Party pursuant to employee profit sharing and/or benefit plans and compensation and non-competition arrangements in amounts customary for corporations similarly situated to the Credit Party and entered into in the ordinary course of business and approved by the board of directors of the Credit Party, or (iii) any reimbursement of reasonable out-of-pocket costs incurred by an Affiliate of the Credit Party on behalf of or for the account of the Credit Party.

- **6.8** Repayment of Debt. No Credit Party will, and will not permit any Restricted Subsidiary to, repay, prepay, redeem, repurchase, defease or otherwise make any payment on account of any Indebtedness for borrowed money except for (a) payment on account of Indebtedness owing to the Agent or the Lenders under this Agreement, (b) any payment consented to in writing by the Required Lenders, and (c) payment on account of Indebtedness for borrowed money permitted by Section 6.1, the repayment of which is not restricted by Section 6.6.
- **6.9 Restrictive Agreements.** No Credit Party will, and will not permit any Restricted Subsidiary to, directly or indirectly, enter into, incur or permit to exist any agreement or other arrangement that prohibits, restricts or imposes any condition upon (a) the ability of the Borrower or any Restricted Subsidiary to create, incur or permit to exist any Lien upon any of its property or assets, (b) the ability of any Restricted Subsidiary to pay dividends or other distributions with respect to any Equity Securities or with respect to, or measured by, its profits or to make or repay loans or advances to the Borrower or any Restricted Subsidiary or to provide a Guarantee of any Indebtedness of the Borrower or any Restricted Subsidiary, (c) the ability of the Borrower or any Restricted Subsidiary to make any loan or advance to the Borrower or any of the Subsidiaries, or (d) the ability of the Borrower or any Restricted Subsidiary to sell, lease or transfer any of its property to the Borrower or any of the Subsidiaries; provided that (i) the foregoing shall not apply to restrictions and conditions imposed by Applicable Law or by this Agreement, (ii) the foregoing shall not apply to restrictions and conditions identified on Schedule 6.9; (iii) the foregoing shall not apply to customary restrictions and conditions contained in agreements relating to the sale of an Restricted Subsidiary pending such sale, provided such restrictions and conditions apply only to the Restricted Subsidiary that is to be sold and such sale is permitted hereunder, (iv) clause (a) of the foregoing shall not apply to customary provisions in leases and other ordinary course contracts restricting the assignment thereof.
- **6.10 <u>Capital Lease Obligations</u>**; <u>Sale/Leaseback Obligations</u>. No Credit Party will, and will not permit any Restricted Subsidiary to, create, incur, assume or suffer to exist, any Capital Lease Obligations or Sale/Leaseback Obligations, whether directly or as a guarantor, if, after giving effect thereto, the aggregate amount of all payments required to be made by the Credit Parties on a consolidated basis pursuant to Capital Lease Obligations and Sale/Leaseback Obligations would exceed Cdn.\$4,000,000 in any Fiscal Year.
- **6.11 Pension Plan Compliance.** No Credit Party will (a) terminate, or permit any Restricted Subsidiary to terminate, any Pension Plan in a manner, or take any other action with respect to any Pension Plan, which could reasonably be expected to result in any material liability of the Borrower or a Restricted Subsidiary of the Borrower, (b) fail to make, or permit any Restricted Subsidiary to fail to make, full payment when due of all amounts which, under the provisions of any Pension Plan, agreement relating thereto or Applicable Law, the Credit Party or any Restricted Subsidiary is required to pay as contributions thereto, except where the failure to make such payments could not reasonably be expected to have a Material Adverse Effect, (c) permit to exist, or allow any Restricted Subsidiary to permit to exist, any accumulated funding deficiency, whether or not waived, with respect to any Pension Plan in an amount which could reasonably be expected to cause a Material Adverse Effect, (d) contribute to or assume an obligation to contribute to, or permit any Restricted Subsidiary (other than any Restricted Subsidiary acquired as permitted pursuant to Section 5.11) to contribute to or assume an obligation to contribute to, any "multi-employer pension plan" as such term is defined in the Pension Benefits Act (Ontario), (e) acquire, or permit any Restricted Subsidiary to acquire, an interest in any Person if such Person sponsors, maintains or contributes to, or at any time in the six-year period preceding such acquisition has sponsored, maintained, or contributed to any "multi-employer pension plan" as such term is defined in the Pension Benefits Act (Ontario); provided that, the Credit Party or any Restricted Subsidiary may acquire an interest in any such Person if such Person is acquired as a Permitted Acquisition and neither the Credit Party nor any of its other Subsidiaries has any legal liability to perform such Person's obligations or assume such Person's liabilities, and (f) permit, or allow any Restricted Subsidiary to permit, the actuarial present value of the benefit liabilities (computed on an accumulated benefit obligation basis in accordance with GAAP) under all Pension Plans in the aggregate to exceed the current value of the assets of all Pension Plans in the aggregate that are allocable to such benefit liabilities, in each case only to the extent such liabilities and assets relate to benefits to be paid to employees of the Credit Parties, by an amount that could reasonably be expected to cause a Material Adverse Effect.

- **6.12** Sale or Discount of Receivables. No Credit Party will, and will not permit any Restricted Subsidiary to, discount or sell (with or without recourse) any of its or the Restricted Subsidiaries' Accounts, provided the Borrower shall be entitled to compromise, adjust, or extend the time for payment of its Accounts, in the ordinary course of business, consistent with past practice or existing arrangements with customers or agents.
- **6.13** <u>Unconditional Purchase Obligations</u>. No Credit Party will, and will not permit any Restricted Subsidiary to, enter into or be a party to, any material contract for the purchase of materials, supplies or other property or services, if such contract requires that payment be made by it regardless of whether or not delivery of such materials, supplies or other property or services is ever made, provided that this Section 6.13 shall not restrict the ability of any Credit Party or any Restricted Subsidiary to enter into any such contract in the ordinary course of its business to the extent that the materials, supplies or other property or services which are the subject matter of such contract are reasonably expected to be used by the applicable Credit Party in the ordinary course of its business.

6.14 [Intentionally Deleted.]

- **6.15** No Amendments to Material Contracts. No Credit Party will, and will not permit any Restricted Subsidiary to, amend, modify or terminate (or waive any provision of or provide any consent under), any Material Contract in a manner which may reasonably be expected to have a Material Adverse Effect.
- **6.16** Prohibited Use of Proceeds; Sanctions. The Borrower will not request any Borrowing, and the Borrower shall not use, and shall ensure each Credit Party and its or their respective director, officers, employees and agents shall not use, the proceeds of any Borrowing (a) in furtherance of an offer, payment, promise to pay, or authorization of the payment or giving of money, or anything else of value, to any Person in violation of any Anti-Corruption Laws; (b) for the purpose of funding, financing or facilitating any activities, business or transaction of or with any Sanctioned Person, or in any Sanctioned Country in violation of Sanctions; or (c) in any manner that would result in the violation of any Sanctions.

ARTICLE 7 EVENTS OF DEFAULT

- 7.1 Events of Default. It shall constitute an event of default ("Event of Default") if any one or more of the following shall occur:
- (a) the Borrower shall fail to pay any principal of any Loan or any reimbursement obligation in respect of any LC Disbursement or F/X Contract within 2 Business Days of the due date thereof;
- (b) the Borrower shall fail to pay any interest on any Loan or any fee or any other amount (other than an amount referred to in clause (a) above) payable under this Agreement within 2 Business Days of the due date thereof;
- (c) any representation or warranty made or deemed made by or on behalf of any Credit Party in or in connection with any Loan Document or any amendment or modification thereof or waiver thereunder, or in any report, certificate, financial statement or other document furnished pursuant to or in connection with any Loan Document or any amendment or modification thereof or waiver thereunder, shall prove to have been incorrect when made or deemed to be made;
- (d) any Credit Party shall fail to observe or perform any covenant, condition or agreement contained in Section 5.1(o) (notices of Defaults or Events of Default), 5.2 (with respect to the Credit Party's existence), 5.9 or in Article 6 (or in any comparable provision of any other Loan Document);
- (e) any Credit Party shall fail to observe or perform any covenant, condition or agreement contained in Section 5.12, and such failure shall continue unremedied for a period of seven days after notice thereof from the Agent to the Borrower (which shall be given at the request of any Lender);
- (f) any Credit Party shall fail to observe or perform any covenant, condition or agreement contained in this Agreement (other than those specified in clauses (a), (b) or (d) above) or any other Loan Document, and such failure shall continue unremedied for a period of 30 days after notice thereof from the Agent to the Borrower (which notice will be given at the request of any Lender);
- (g) any Credit Party shall fail to make any payment whether of principal or interest, and regardless of amount, in respect of any Material Indebtedness, when and as the same shall become due and payable within 2 Business Days after receipt of notice thereof;
- (h) any event or condition occurs that results in any Material Indebtedness becoming due prior to its scheduled maturity or that enables or permits (with or without the giving of notice, the lapse of time or both) the holder or holders of any Material Indebtedness or any trustee or agent on its or their behalf to cause any Material Indebtedness to become due, or to require the prepayment, repurchase, redemption or defeasance thereof, prior to its scheduled maturity; provided that this Section 7.1(h) shall not apply to secured Indebtedness that becomes due as a result of the voluntary sale or transfer of the property or assets securing such Indebtedness so long as the proceeds of such sale or transfer are sufficient to, and are applied to, reduce such secured Indebtedness to nil within 2 Business Days after receipt of notice thereof;

(i) any Credit Party or the Parent:

- (i) becomes insolvent, or generally does not or becomes unable to pay its debts or meet its liabilities as the same become due, or admits in writing its inability to pay its debts generally, or declares any general moratorium on its indebtedness, or proposes a compromise or arrangement between it and any class of its creditors;
- (ii) commits an act of bankruptcy under the BIA, or makes an assignment of its property for the general benefit of its creditors under the BIA, or makes a proposal (or files a notice of its intention to do so) under the BIA;
- (iii) institutes any proceeding seeking to adjudicate it an insolvent, or seeking liquidation, dissolution, winding-up, reorganization, compromise, arrangement, adjustment, protection, moratorium, relief, stay of proceedings of creditors generally (or any class of creditors), or composition of it or its debts or any other relief, under any federal, provincial or foreign Applicable Law now or hereafter in effect relating to bankruptcy, winding-up, insolvency, reorganization, receivership, plans of arrangement or relief or protection of debtors (including the BIA, the *Companies' Creditors Arrangement Act* (Canada), the *United States Bankruptcy Code* and any applicable corporations legislation) or at common law or in equity, or files an answer admitting the material allegations of a petition filed against it in any such proceeding;
- (iv) applies for the appointment of, or the taking of possession by, a receiver, interim receiver, receiver/manager, sequestrator, conservator, custodian, administrator, trustee, liquidator or other similar official for it or any substantial part of its property; or
- (v) threatens to do any of the foregoing, or takes any action, corporate or otherwise, to approve, effect, consent to or authorize any of the actions described in this Section 7.1(i) or in Section 7.1(j), or otherwise acts in furtherance thereof or fails to act in a timely and appropriate manner in defense thereof,
- (j) any petition is filed, application made or other proceeding instituted against or in respect of any Credit Party or the Parent:
 - (i) seeking to adjudicate it an insolvent;
 - (ii) seeking a receiving order against it under the BIA;
 - (iii) seeking liquidation, dissolution, winding-up, reorganization, compromise, arrangement, adjustment, protection, moratorium, relief, stay of proceedings of creditors generally (or any class of creditors), or composition of it or its debts or any other relief under any federal, provincial or foreign Applicable Law now or hereafter in effect relating to bankruptcy, winding-up, insolvency, reorganization, receivership, plans of arrangement or relief or protection of debtors (including the BIA, the *Companies' Creditors Arrangement Act* (Canada) or the *United States Bankruptcy Code* and any applicable corporations legislation) or at common law or in equity; or

(iv) seeking the entry of an order for relief or the appointment of, or the taking of possession by, a receiver, interim receiver, receiver/manager, sequestrator, conservator, custodian, administrator, trustee, liquidator or other similar official for it or any substantial part of its property;

and such petition, application or proceeding continues undismissed, or unstayed and in effect, for a period of 30 days after the institution thereof, provided that if an order, decree or judgment is granted or entered (whether or not entered or subject to appeal) against the Credit Party or the Parent, as applicable, thereunder in the interim, such grace period will cease to apply, and provided further that if the Credit Party or the Parent, as applicable, files an answer admitting the material allegations of a petition filed against it in any such proceeding, such grace period will cease to apply;

- (k) any other event occurs which, under the Applicable Laws of any applicable jurisdiction, has an effect equivalent to any of the events referred to in either of Sections 7.1(i) or (j);
- (l) one or more judgments for the payment of money in a cumulative amount in excess of Cdn.\$4,000,000 (or its then equivalent in any other currency) in the aggregate is rendered against the Borrower, any other Credit Party or any combination thereof and the Borrower or the other Credit Party has not (i) provided for its discharge in accordance with its terms within 30 days from the date of entry thereof, or (ii) procured a stay of execution thereof within 30 days from the date of entry thereof and within such period, or such longer period during which execution of such judgment has not been stayed, appealed such judgment and caused the execution thereof to be stayed during such appeal, provided that if enforcement and/or realization proceedings are lawfully commenced in respect thereof in the interim, such grace period will cease to apply;
- (m) any property of any Credit Party having a fair market value in excess of Cdn.\$4,000,000 (or its then equivalent in any other currency) in the aggregate is seized (including by way of execution, attachment, garnishment, levy or distraint), or any Lien thereon securing Indebtedness in excess of Cdn.\$4,000,000 (or its then equivalent in any other currency) is enforced, or such property has become subject to any charging order or equitable execution of a Governmental Authority, or any writ of execution or distress warrant exists in respect of the Borrower, any other Credit Party or the property of any of them, or any sheriff or other Person becomes lawfully entitled by operation of law or otherwise to seize or distrain upon such property and in any case such seizure, enforcement, execution, attachment, garnishment, distraint, charging order or equitable execution, or other seizure or right, continues in effect and is not released or discharged for more than 45 days or such longer period during which entitlement to the use of such property continues with the Credit Party (as the case may be), and the Credit Party (as the case may be) is contesting the same in good faith and by appropriate proceedings, provided that if the property is removed from the use of the Credit Party (as the case may be), or is sold, in the interim, such grace period will cease to apply;
- (n) one or more final judgments, not involving the payment of money and not otherwise specified in this Section 7.1(n), has been rendered against any Credit Party, the result of which could reasonably be expected to result in a Material Adverse Effect, so long as the Credit Party (as the case may be) has not (i) provided for its discharge in accordance with its terms within 30 days from the date of entry thereof, or (ii) procured a stay of execution thereof within 30 days from the date of entry thereof and within such period, or such longer period during which execution of such judgment has been stayed, appealed such judgment and caused the execution thereof to be stayed during such appeal, provided that if enforcement and/or realization proceedings are lawfully commenced in respect thereof in the interim, such grace period will cease to apply;

- (o) this Agreement, any other Loan Document or any material obligation or other provision hereof or thereof at any time for any reason (other than through the fault of the Agent or the Lenders, except where the Borrower refuses to reinstate such obligation or provision) terminates or ceases to be in full force and effect and a legally valid, binding and enforceable obligation of any Credit Party, is declared to be void or voidable or is repudiated, or the validity, binding effect, legality or enforceability hereof or thereof is at any time contested by any Credit Party, or any Credit Party denies that it has any or any further liability or obligation hereunder or thereunder or any action or proceeding is commenced to enjoin or restrain the performance or observance by any Credit Party of any material terms hereof or thereof or to question the validity or enforceability hereof or thereof, or at any time it is unlawful or impossible for any Credit Party to perform any of its material obligations hereunder or thereunder;
- (p) any Lien purported to be created by any Security Document shall cease to be, or shall be asserted by any Credit Party not to be, a valid, perfected, first priority (except as otherwise expressly provided in this Agreement or such Security Document) Lien in Collateral with a fair market value or book value (whichever is greater) in excess, individually or in the aggregate, of \$1,000,000 (except where such Lien ceases to be a valid, perfected first priority Lien through the failure or inaction of the Agent or the Lenders);
 - (q) a Material Adverse Change shall occur;
- (r) the Parent fails to directly own (i) 50% or more of the aggregate issued and outstanding limited partnership units in the capital of the Borrower or (ii) 100% of the aggregate issued and outstanding shares in the capital of the General Partner; or
- (s) any Credit Party other than as disclosed in Schedule 3.11 hereof as of the date hereof (A) institutes any step to terminate or wind up, in whole or in part, a Defined Benefits Plan, (B) has instituted against it by a Governmental Authority any steps to terminate or wind up, in whole or in part, a Defined Benefits Plan, including the issuance of a notice of intended decision to wind up, (C) has instituted against it or issued to it any order or proposal in respect of a Defined Benefits Plan that directly or indirectly requires the payment of or results in the obligation to pay, any monetary amount in respect of such Defined Benefits Plan in excess of \$1,000,000, or (D) winds up or terminates, in whole or in part, a Defined Benefits Plan.

7.2 Remedies.

(a) If an Event of Default has occurred and is continuing, the Agent may, in its discretion, and shall, at the direction of the Required Lenders, do one or more of the following at any time or times and in any order, without notice to or demand on the Borrower: (i) reduce the Commitments, or the advance rates against Eligible Accounts and/or Eligible Inventory used in computing the Borrowing Base, or reduce one or more of the other elements used in computing the Borrowing Base; (ii) restrict the amount of or refuse to make Revolving Loans; (iii) restrict or refuse to provide Letters of Credit and F/X Contracts; (iv) terminate the Commitments; (v) declare any or all Obligations to be immediately due and payable; and (vi) pursue its other rights and remedies under the Loan Documents and applicable law and equity.

(b) If an Event of Default has occurred and is continuing and without limiting any rights or remedies arising under the Security Documents, (i) the Agent shall have for the benefit of the Lenders, in addition to all other rights of the Agent and the Lenders, the rights and remedies of a secured party under applicable law in the jurisdiction where the Collateral is located and all rights and remedies provided for in the Loan Documents; (ii) the Agent may, at any time, take possession of the Collateral and keep it on the Borrower's or any Guarantor's premises, at no cost to the Agent or any Lender, or remove any part of it to such other place or places as the Agent may desire, or the Borrower or any Guarantor shall, upon the Agent's demand, at the Borrower's cost, assemble the Collateral and make it available to the Agent at a place convenient to the Agent; and (iii) the Agent may sell and deliver any Collateral at public or private sales, for cash, upon credit or otherwise, at such prices and upon such terms as the Agent deems advisable, in its sole discretion, and may postpone or adjourn any sale of the Collateral by an announcement at the time and place of sale or of such postponed or adjourned sale without giving a new notice of sale. Without in any way requiring notice to be given in the following manner, the Borrower and each of the Guarantors agree that any notice by the Agent of sale, disposition or other intended action hereunder or in connection herewith, whether required by applicable law or otherwise, shall constitute reasonable notice to the Borrower and Guarantors if such notice is mailed by registered or certified mail, return receipt requested, postage prepaid, or is delivered personally against receipt, at least fifteen (15) days prior to such action to the Borrower's address specified in or pursuant to Section 9.1. If any Collateral is sold on terms other than payment in full at the time of sale, no credit shall be given against the Obligations until the Agent or the Lenders receive payment, and if the buyer defaults in payment, the Agent may resell the Collateral without further notice to the Borrower or any Guarantor. If the Agent seeks to take possession of all or any portion of the Collateral by judicial process, the Borrower and each of the Guarantors irrevocably waives: (A) the posting of any bond, surety or security with respect thereto which might otherwise be required; (B) any demand for possession prior to the commencement of any suit or action to recover the Collateral; and (C) any requirement that the Agent retain possession and not dispose of any Collateral until after trial or final judgment. The Borrower and each of the Guarantors agree that the Agent and Lenders have no obligation to preserve rights to the Collateral or marshal any Collateral for the benefit of any Person. The Agent is hereby granted a license or other right to use, without charge, all of the Borrower's property, whether or not constituting Collateral, including its real estate, Equipment and intellectual property rights (including labels, patents, copyrights, name, trade secrets, trade names, trademarks, and advertising matter, or any similar property), in completing production of, advertising or selling any Collateral, and the Borrower's rights under all licenses and all franchise agreements shall inure to the Agent's benefit for such purpose. The proceeds of sale shall be applied first to all expenses of sale, including legal fees, and then to the Obligations. The Agent will return any excess to the Borrower and Guarantors and the Borrower shall remain liable for any deficiency.

(c) If an Event of Default has occurred and is continuing, to the maximum extent permitted by law, the Borrower and each of the Guarantors hereby waive all rights to notice and hearing prior to the exercise by the Agent of the Agent's rights to repossess the Collateral without judicial process or to reply, attach or levy upon the Collateral without notice or hearing.

(d) During the continuance of an Event of Default, the Agent may, and upon the direction of the Required Lenders the Agent shall, apply any and all payments received by the Agent in respect of any Obligations as set forth below. Notwithstanding any provision herein to the contrary, all payments made by or for the account of the Credit Parties to the Agent after any or all of the Obligations have been accelerated (so long as such acceleration has not been rescinded), including proceeds of Collateral, shall be applied as follows:

first, to payment of costs and expenses, including legal costs, of the Agent payable or reimbursable by the Credit Parties under the Loan Documents;

second, to payment of legal costs of Lenders payable or reimbursable by the Borrower under this Agreement;

third, to payment of all accrued unpaid interest on the Obligations and fees owed to Agent, Lenders and the Issuing Bank;

fourth, to payment of all Loans, reimbursement obligations in respect of Letters of Credit, F/X Exposure and Cover);

fifth, to payment of any other amounts owing which constitute Obligations; and

sixth, any remainder shall be for the account of and paid to whoever may be lawfully entitled thereto.

In carrying out the foregoing, (i) amounts received shall be applied in the numerical order provided until exhausted prior to the application to the next succeeding category, and (ii) each of the Lenders or other Persons entitled to payment shall receive an amount equal to its pro rata share of amounts available to be applied pursuant to each applicable category.

(e) If the Agent receives any payment from or for the account of a Credit Party in any currency other than the currency in which the Obligation is denominated, the Agent may convert the payment (including the proceeds of realization upon any Collateral) in accordance with its normal practice into the currency in which such Obligation is denominated.

ARTICLE 8 THE AGENT

- **8.1** Appointment of Agent. Each Lender hereby designates Canadian Imperial Bank of Commerce as Agent to act as herein specified and as specified in the other Loan Documents. Each Lender hereby irrevocably authorizes the Agent to take such action on its behalf under the provisions of the Loan Documents and to exercise such powers and to perform such duties thereunder as are specifically delegated to or required of the Agent by the terms thereof and such other powers as are reasonably incidental thereto. The Agent may perform any of its duties hereunder by or through its agents or employees.
- **8.2** <u>Limitation of Duties of Agent</u>. The Agent shall have no duties or responsibilities except those expressly set forth with respect to the Agent in this Agreement and as specified in the other Loan Documents. Neither the Agent nor any of its Related Parties shall be liable for any action taken or omitted by it hereunder or in connection herewith, unless caused by its or their gross negligence or willful misconduct. The duties of the Agent shall be mechanical and administrative in nature; the Agent shall not have, by reason of this Agreement or the other Loan Documents, a fiduciary relationship in respect of any Lender. Nothing in this Agreement or the other Loan Documents, expressed or implied, is intended to or shall be so construed as to impose upon the Agent any obligations in respect of this Agreement except as expressly set forth herein. The Agent shall be under no duty to take any discretionary action permitted to be taken by it pursuant to this Agreement or the other Loan Documents unless it is requested in writing to do so by the Required Lenders.

8.3 Lack of Reliance on the Agent.

- (a) <u>Independent Investigation</u>. Independently, and without reliance upon the Agent, each Lender, to the extent it deems appropriate, has made and shall continue to make (i) its own independent investigation of the financial condition and affairs of the Borrower and its Subsidiaries in connection with the taking or not taking of any action in connection herewith, and (ii) its own appraisal of the creditworthiness of the Borrower and its Subsidiaries, and, except as expressly provided in this Agreement and the other Loan Documents, the Agent shall have no duty or responsibility, either initially or on a continuing basis, to provide any Lender with any credit or other information with respect thereto, whether coming into its possession before the consummation of the Transactions or at any time or times thereafter.
- (b) Agent Not Responsible. The Agent shall not be responsible to any Lender for any recitals, statements, information, representations or warranties herein or in any document, certificate or other writing delivered in connection herewith or for the execution, effectiveness, genuineness, validity, enforceability, collectability, priority or sufficiency of this Agreement or the other Loan Documents or the financial condition of the Borrower and its Subsidiaries or be required to make any inquiry concerning either the performance or observance of any of the terms, provisions or conditions of this Agreement or the other Loan Documents, or the financial condition of the Borrower and its Subsidiaries, or the existence or possible existence of any Default or Event of Default.
- **8.4** Certain Rights of the Agent. If the Agent shall request instructions from the Lenders or the Required Lenders (as the case may be) with respect to any act or action (including the failure to act) in connection with this Agreement or the other Loan Documents, the Agent shall be entitled to refrain from such act or taking such action unless and until the Agent shall have received written instructions from the Lenders or the Required Lenders, as applicable, and the Agent shall not incur liability to any Person by reason of so refraining. Without limiting the foregoing, no Lender shall have any right of action whatsoever against the Agent as a result of the Agent acting or refraining from acting under this Agreement and the other Loan Documents in accordance with the instructions of the Required Lenders, or, to the extent required by Section 9.2, all of the Lenders.
- **8.5** Reliance by Agent. The Agent shall be entitled to rely, and shall be fully protected in relying, upon any note, writing, resolution, notice, statement, certificate, telex, teletype or facsimile message, electronic mail, cablegram, radiogram, order or other documentary teletransmission or telephone message believed by it to be genuine and correct and to have been signed, sent or made by the proper Person. The Agent may consult with legal counsel (including counsel for the Borrower), independent public accountants and other experts selected by it and shall not be liable for any action taken or omitted to be taken by it in good faith in accordance with the advice of such counsel, accountants or experts.
- **8.6** <u>Indemnification of Agent</u>. To the extent the Agent is not reimbursed and indemnified by the Borrower, each Lender will reimburse and indemnify the Agent, in proportion to its aggregate Applicable Percentage, for and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses (including reasonable counsel fees and disbursements) or disbursements of any kind or nature whatsoever which may be imposed on, incurred by or asserted against the Agent in performing its duties hereunder, in any way relating to or arising out of this Agreement or any other Loan Document; provided that no Lender shall be liable to the Agent for any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements resulting from the Agent's gross negligence (it being acknowledged that ordinary negligence does not necessarily constitute gross negligence) or willful misconduct.

- **8.7** The Agent in its Individual Capacity. With respect to its obligations under this Agreement and the Loans made by it, Canadian Imperial Bank of Commerce, in its capacity as a Lender hereunder, shall have the same rights and powers hereunder as any other Lender and may exercise the same as though it were not performing the duties, if any, specified herein; and the terms "Lenders", "Required Lenders", and any similar terms shall, unless the context clearly otherwise indicates, include Canadian Imperial Bank of Commerce, in its capacity as a Lender hereunder. The Agent may accept deposits from, lend money to, and generally engage in any kind of banking, trust, financial advisory or other business with the Borrower or any affiliate of the Borrower as if it were not performing the duties, if any, specified herein, and may accept fees and other consideration from the Borrower for services in connection with this Agreement and otherwise without having to account for the same to the Lenders.
- **8.8** May Treat Lender as Owner. The Borrower and the Agent may deem and treat each Lender as the owner of the Loans recorded on the Register maintained pursuant to Section 9.4(c) for all purposes hereof until a written notice of the assignment or transfer thereof shall have been filed with the Agent. Any request, authority or consent of any Person who at the time of making such request or giving such authority or consent is the owner of a Loan shall be conclusive and binding on any subsequent owner, transferee or assignee of such Loan.

8.9 Successor Agent.

- (a) Agent Resignation. The Agent may resign at any time by giving written notice thereof to the Lenders, the Issuing Bank and the Borrower. Upon any such resignation or removal, the Required Lenders shall have the right, upon five Business Days' notice to the Borrower, to appoint a successor Agent (who shall not be a non-resident of Canada within the meaning of the ITA), subject to the approval of the Borrower, such approval not to be unreasonably withheld. If no successor Agent shall have been so appointed by the Required Lenders, and shall have accepted such appointment, within 30 days after the retiring Agent's giving of notice of resignation or the Required Lenders' removal of the retiring Agent, then, upon five Business Days' notice to the Borrower, the retiring Agent may, on behalf of the Lenders, appoint a successor Agent (subject to approval of the Borrower, such approval not to be unreasonably withheld), which shall be a financial institution organized under the laws of Canada having a combined capital and surplus of at least Cdn.\$100,000,000 or having a parent company with combined capital and surplus of at least Cdn.\$100,000,000.
- (b) Rights, Powers, etc. Upon the acceptance of any appointment as Agent hereunder by a successor Agent, such successor Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Agent, and the retiring Agent shall be discharged from its duties and obligations under this Agreement. After any retiring Agent's resignation or removal hereunder as Agent, the provisions of this Article 8 shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Agent under this Agreement.

8.10 No Independent Legal Action by Lenders. No Lender may take any independent legal action to enforce any obligation of the Borrower hereunder. Each Lender hereby acknowledges that, to the extent permitted by Applicable Law, the Security Documents and the remedies provided thereunder to the Lenders are for the benefit of the Lenders collectively and acting together and not severally, and further acknowledges that each Lender's rights hereunder and under the Security Documents are to be exercised collectively, not severally, by the Agent upon the decision of the Required Lenders. Accordingly, notwithstanding any of the provisions contained herein or in the Security Documents, each of the Lenders hereby covenants and agrees that it shall not be entitled to take any action hereunder or thereunder, including any declaration of default hereunder or thereunder, but that any such action shall be taken only by the Agent with the prior written agreement of the Required Lenders, provided that, notwithstanding the foregoing, in the absence of instructions from the Lenders (or the Required Lenders) and where in the sole opinion of the Agent the exigencies of the situation so warrant such action, the Agent may without notice to or consent of the Lenders (or the Required Lenders) take such action on behalf of the Lenders as it deems appropriate or desirable in the interests of the Lenders. Each Lender hereby further covenants and agrees that upon any such written consent being given by the Required Lenders, it shall co-operate fully with the Agent to the extent requested by the Agent, and each Lender further covenants and agrees that all proceeds from the realization of or under the Security Documents, to the extent permitted by Applicable Law, are held for the benefit of all of the Lenders and shall be shared among the Lenders rateably in accordance with this Agreement, and each Lender acknowledges that all costs of any such realization (including all amounts for which the Agent is required to be indemnified under the provisions hereof) shall be shared among the Lenders rateably in accordance with this Agreement. Each Lender covenants and agrees to do all acts and things and to make, execute and deliver all agreements and other instruments, so as to fully carry out the intent and purpose of this Section and each Lender hereby covenants and agrees that it shall not seek, take, accept or receive any security for any of the obligations and liabilities of the Borrower hereunder or under the other Loan Documents, or any other document, instrument, writing or agreement ancillary hereto or thereto, other than such security as is provided hereunder or thereunder, and that it shall not enter into any agreement with any of the parties hereto or thereto relating in any manner whatsoever to the Credit(s), unless all of the Lenders shall at the same time obtain the benefit of any such security or agreement, as the case may be.

8.11 Quebec Security. For greater certainty, and without limiting the powers of the Agent or any other Person acting as an agent or mandatary for the Agent hereunder or under any of the other Loan Documents, the Borrower hereby acknowledges that, for purposes of holding any security granted by the Borrower or any Restricted Subsidiary on property pursuant to the laws of the Province of Quebec to secure obligations of the Borrower or any Restricted Subsidiary under any debenture, the Agent shall be the holder of an irrevocable power of attorney (fondé de pouvoir) (within the meaning of the Civil Code of Quebec) for all present and future Lenders and in particular for all present and future holders of any such debenture. Each Lender hereby irrevocably constitutes, to the extent necessary, the Agent as the holder of an irrevocable power of attorney (fondé de pouvoir) (within the meaning of Article 2692 of the Civil Code of Quebec) in order to hold security granted by the Borrower or any Restricted Subsidiary in the Province of Quebec to secure the obligations of the Borrower or any Restricted Subsidiary under any debenture. Each assignee of a Lender shall be deemed to have confirmed and ratified the constitution of the Agent as the holder of such irrevocable power of attorney (fondé de pouvoir) by execution of an Assignment and Transfer. Notwithstanding the provisions of section 32 of the An Act respecting the special powers of legal persons (Quebec), the Agent may acquire and be the holder of any debenture. The Borrower hereby acknowledges that such debenture constitutes a title of indebtedness, as such term is used in Article 2692 of the Civil Code of Quebec.

8.12 F/X Contracts. The obligations of the Credit Parties in respect of an F/X Contract between the Borrower and an F/X Bank or a Lender Affiliate are secured by the Security Documents, pari passu with the obligations of the Credit Parties under the Loan Documents, provided that all decisions regarding the administration and enforcement of the security interests granted under the Security Documents shall be made by the Agent and the Lenders under this Agreement, and while this Agreement remains in effect, any F/X Bank or Lender Affiliate shall (in such capacity) have no voting rights under this Agreement and no other right whatsoever to participate in the administration or enforcement of such security interests. For the avoidance of doubt but without limitation, any or all of the Security Documents or any rights contained therein may be amended or released by the Agent without the consent of any F/X Bank or Lender Affiliate. Each Lender that is or becomes an F/X Bank shall be bound as such by virtue of its execution and delivery of this Agreement or an assignment and assumption agreement substantially in the form of Exhibit E, as applicable, notwithstanding that such capacity as F/X Bank may not be identified on its signature line.

ARTICLE 9 MISCELLANEOUS

- 9.1 Notices. (a) Except in the case of notices and other communications expressly permitted to be given by telephone (and subject to paragraph (b) below), all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by facsimile in each case to the addressee, as follows:
 - if to the Borrower or any other Credit Party:

Suite 11200 700 West Pender Street Vancouver, B.C. V6C 1G8 Attention: Genevieve Stannus Facsimile: (604) 684-1094

with a copy to:

Sangra Moller LLP 1000 Cathedral Place 925 West Georgia Street Vancouver, B.C. V6C 3L2 Attention: Harj Sangra Facsimile: (604) 669-8803

(ii) if to the Agent:

Canadian Imperial Bank of Commerce 199 Bay Street, 4th Floor Toronto, Ontario M5L 1A9

Attention: Senior Director, Portfolio Manager

Facsimile: (416) 861-9422

with a copy to:

Canadian Imperial Bank of Commerce 199 Bay Street, 11th Floor Toronto, Ontario M5L 1A9 Attention: Nick Chan

Facsimile: (416) 304-4573 Email: nick.chan@cibc.com

- (iii) if to any Lender or any Issuing Bank, to it at its address (or facsimile number) set forth opposite its name in the execution page(s) of this Agreement or the applicable Assignment and Transfer Agreement, as the case may be.
- (b) Notices and other communications to the Lenders hereunder may be delivered or furnished by electronic communications pursuant to procedures approved by the Agent. The Agent or the Borrower may, in its discretion, agree to accept notices and other communication to it hereunder by electronic communications pursuant to procedures approved by it; <u>provided</u> that approval of such procedures may be limited to particular notices or communications.
- (c) Any party hereto may change its address or facsimile number for notices and other communications hereunder by notice to the other parties hereto. All notices and other communications given to any party hereto in accordance with the provisions of this Agreement shall be deemed to have been given on the date of receipt.

9.2 Waivers; Amendments.

- (a) No failure or delay by the Agent or any Lender in exercising any right or power hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right or power, or any abandonment or discontinuance of steps to enforce such a right or power, preclude any other or further exercise thereof or the exercise of any other right or power. The rights and remedies of the Agent and the Lenders hereunder are cumulative and are not exclusive of any rights or remedies that they would otherwise have. No waiver of any provision of this Agreement or consent to any departure by the Borrower therefrom shall in any event be effective unless the same shall be permitted by Section 9.2(b), and then such waiver or consent shall be effective only in the specific instance and for the purpose for which given. Without limiting the generality of the foregoing, the making of a Loan or issuance of a Letter of Credit shall not be construed as a waiver of any Default, regardless of whether the Agent or any Lender may have had notice or knowledge of such Default at the time.
- (b) Neither this Agreement nor any other Loan Document (or any provision hereof or thereof) may be waived, amended or modified except pursuant to an agreement or agreements in writing entered into by the Borrower and the Required Lenders or by the Borrower and the Agent with the consent of the Required Lenders (and for greater certainty, any such waiver, amendment or modification shall not require any consent or other agreement of any Credit Party other than the Borrower, notwithstanding that any such Credit Party may be a party to this Agreement or any other Loan Document); provided that no such agreement shall:
 - (i) increase the amount or extend the expiry date of any Commitment of any Lender;

- (ii) reduce the principal amount of any Loan or reduce the rate of interest or any fee applicable to any Loan;
- (iii) postpone the scheduled date of payment of the principal amount of any Loan, or any interest thereon, or any fees payable in respect thereof, or reduce the amount of, waive or excuse any such payment, or postpone the scheduled date of expiration of any Commitment;
- (iv) change any aspect of this Agreement in a manner that would alter the pro rata sharing of payments required herein;
- (v) change any of the provisions of this Section 9.2 or the definition of "**Required Lenders**" or any other provision hereof specifying the number or percentage of Lenders required to waive, amend or modify any rights hereunder or make any determination or grant any consent hereunder;
- (vi) waive any Event of Default under Section 7.1(i), (j) or (k); or
- (vii) release the Borrower or any Subsidiary from any material obligations under the Security Documents and other instruments contemplated by this Agreement, release or discharge any of the Liens arising under the Security Documents, permit the creation of any Liens, other than Permitted Liens, on any of the assets subject to the Liens arising under the Security Documents, lower the priority of any Lien arising under any of the Security Documents, or lower the priority of any payment obligation of the Borrower or any Subsidiary under any of the Loan Documents or, in the case of clauses (i), (ii), (iii) or (iv), without the prior written consent of each Lender directly affected thereby;

in each case without the prior written consent of each Lender; or, in the case of the matters referred to in clauses (i), (ii), (iii) and (iv), without the prior written consent of each Lender directly affected thereby and provided further that no such agreement shall amend, modify or otherwise affect the rights or duties of the Agent hereunder, without the prior written consent of the Agent. For greater certainty, the Agent may release and discharge the Liens constituted by the Security Documents to the extent necessary to enable the Borrower to complete any asset sale which is not prohibited by this Agreement or the other Loan Documents.

9.3 Expenses; Indemnity; Damage Waiver.

(a) The Borrower shall pay (i) all reasonable Out-of-Pocket Expenses incurred by the Agent and its Affiliates, including the reasonable fees, charges and disbursements of counsel for the Agent and all applicable Taxes, in connection with the syndication of the credit facilities provided for herein and the preparation and administration of this Agreement and the other Loan Documents, (ii) all reasonable Out-of-Pocket Expenses incurred by the Agent and its Affiliates, including the reasonable fees, charges and disbursements of counsel for the Agent and applicable Taxes, in connection with any amendments, modifications or waivers of the provisions hereof or of any of the other Loan Documents, (whether or not the transactions contemplated hereby or thereby shall be consummated), and (iii) all Out-of-Pocket Expenses incurred by the Agent or any Lender, including the fees, charges and disbursements of any counsel for the Agent or any Lender and all applicable Taxes, in connection with the enforcement or protection of their rights in connection with this Agreement, including its rights under this Section, or in connection with the Loans made hereunder, including all such Out-of-Pocket Expenses incurred during any workout, restructuring or negotiations in respect of such Loans.

- (b) The Borrower shall indemnify the Agent and each Lender, as well as each Related Party and each assignee of any of the foregoing Persons (each such Person and each such assignee being called an "Indemnitee") against, and hold each Indemnitee harmless from, any and all losses, claims, cost recovery actions, damages, expenses and liabilities of whatsoever nature or kind and all Out-of-Pocket Expenses and all applicable Taxes to which any Indemnitee may become subject arising out of or in connection with (i) the execution or delivery of the Loan Documents or any agreement or instrument contemplated thereby, the performance by the parties thereto of their respective obligations thereunder, and the consummation of the Transactions or any other transactions thereunder, (ii) any Loan or Letter of Credit or any actual or proposed use of the proceeds therefrom, including any refusal by the Issuing Bank to honour a demand for payment under a Letter of Credit if the documents presented in connection with such demand do not strictly comply with the terms of such Letter of Credit, (iii) any actual or alleged presence or release of Hazardous Materials on or from any property owned or operated by the Borrower or any of its Subsidiaries, or any Environmental Liability related in any way to the Borrower or any of its Subsidiaries, (iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory and regardless of whether any Indemnitee is a party thereto, (v) any other aspect of this Agreement and the other Loan Documents, or (vi) the enforcement of any Indemnitee's rights hereunder and any related investigation, defence, preparation of defence, litigation and enquiries, in each case regardless of whether or not the Acquisition is consummated; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such losses, claims, damages, liabilities or related expenses are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence (it being acknowledged that ordinary negligence does not necessarily constitute gross negligence) or wilful misconduct of or material breach of this Agreement by such Indemnitee.
- (c) To the extent that the Borrower fails to pay any amount required to be paid under Sections 9.3 (a) or (b), each Lender severally agrees to pay to the Agent such Lender's Applicable Percentage (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought) of such unpaid amount; <u>provided</u> that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Agent, in its capacity as such.
- (d) The Borrower shall not assert, and hereby waives (to the fullest extent permitted by applicable Law), any claim against any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, any Loan Document, or any agreement or instrument contemplated thereby, the Transactions, any Loan or Letter of Credit or the use of the proceeds thereof.
- (e) Any inspection of any property of the Borrower or any of its Subsidiaries made by or through the Agent or any Lender is for purposes of administration of the Credits only, and neither the Borrower nor any of its Subsidiaries is entitled to rely upon the same (whether or not such inspections are at the expense of the Borrower).

- (f) By accepting or approving anything required to be observed, performed, fulfilled or given to the Agent or the Lenders pursuant to the Loan Documents, neither the Agent nor the Lenders shall be deemed to have warranted or represented the sufficiency, legality, effectiveness or legal effect of the same, or of any term, provision or condition thereof, and such acceptance or approval thereof shall not constitute a warranty or representation to anyone with respect thereto by the Agent or the Lenders.
- (g) The relationship between the Borrower and the Agent and the Lenders is, and shall at all times remain, solely that of borrowers and lenders. Neither the Agent nor the Lenders shall under any circumstance be construed to be partners or joint venturers of the Borrower or its Affiliates. Neither the Agent nor the Lenders shall under any circumstance be deemed to be in a relationship of confidence or trust or a fiduciary relationship with the Borrower or its Affiliates, or to owe any fiduciary duty to the Borrower or its Affiliates. Neither the Agent nor the Lenders undertake or assume any responsibility or duty to the Borrower or its Affiliates to select, review, inspect, supervise, pass judgment upon or inform the Borrower or its Affiliates of any matter in connection with their property or the operations of the Borrower or its Affiliates. The Borrower and its Affiliates shall rely entirely upon their own judgment with respect to such matters, and any review, inspection, supervision, exercise of judgment or supply of information undertaken or assumed by the Agent or the Lenders in connection with such matters is solely for the protection of the Agent and the Lenders, and neither the Borrower nor any other Person is entitled to rely thereon.
- (h) The Agent and the Lenders shall not be responsible or liable to any Person for any loss, damage, liability or claim of any kind relating to injury or death to Persons or damage to Property caused by the actions, inaction or negligence of the Borrower or any Subsidiary and/or their Affiliates and the Borrower hereby indemnifies and holds the Agent and the Lenders harmless from any such loss, damage, liability or claim.
- (i) This Agreement is made for the purpose of defining and setting forth certain obligations, rights and duties of the Borrower, the Agent and the Lenders in connection with the Loans, and is made for the sole benefit of the Borrower, the Agent and the Lenders, and the Agent's and each Lender's successors and assigns. Except as provided in Sections 9.3(b) and 9.4, no other Person shall have any rights of any nature hereunder or by reason hereof.
 - (j) All amounts due under this Section 9.3 shall be payable not later than three Business Days after written demand therefor.

9.4 Successors and Assigns.

(a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that (i) the Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of each Lender (and any attempted assignment or transfer by the Borrower without such consent shall be null and void), and (ii) no Lender may assign or otherwise transfer its rights or obligations hereunder except in accordance with this Section. Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby and, to the extent expressly contemplated hereby, the Related Parties of each of the Agent and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.

- (b) Any Lender may assign to one or more assignees (treating any fund that invests in bank loans and any other fund that invests in bank loans and is managed by the same investment advisor of such fund or by an Affiliate of such investment advisor as a single assignee) all or a portion of its rights and obligations under this Agreement and the other Loan Documents (including all or a portion of its Commitments and the Loans at the time owing to it); provided that (i) except in the case of an assignment of any Commitment to an assignee that is a Lender with a Commitment immediately prior to giving effect to such assignment, each of the Agent and the Borrower must give its prior written consent to such assignment (which consent shall not be unreasonably withheld or delayed); and provided further that (ii) the Borrower's consent shall not be required with respect to any assignment made at any time after the occurrence and during the continuance of an Event of Default, (iii) except in the case of an assignment to a Lender or a Lender Affiliate or an assignment of the entire remaining amount of the assigning Lender's Commitment, the amount of the Commitment of the assigning Lender subject to each such assignment (determined as of the date on which the Assignment and Transfer relating to such assignment is delivered to the Agent) shall not be less than Cdn.\$5,000,000 (or, in the case of a U.S. Dollar-denominated Commitment, the U.S.\$ Equivalent of Cdn.\$5,000,000), unless each of the Borrower and the Agent otherwise consent in writing and the amount held by each Lender after each such assignment shall not be less than Cdn.\$5,000,000 (or, in the case of a U.S. Dollar-denominated Commitment, the U.S.\$ Equivalent of Cdn.\$5,000,000), unless each of the Borrower and the Agent otherwise consent in writing, (iv) each partial assignment in respect of a Commitment and the related Loans shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement in respect of such Commitment and the related Loans, (v) the parties to each assignment shall execute and deliver to the Agent an Assignment and Transfer, together with (except in the case of an assignment to a Lender or a Lender Affiliate) a processing and recordation fee of Cdn.\$7,500, payable by the assigning Lender, (vi) in the case of an assignment to a Foreign Lender, such Foreign Lender shall not be entitled to require any payment under Section 2.14(a)(i) as a result of any withholding tax exigible in respect of any payment by the Borrower to such Foreign Lender hereunder, and (vii) the assignee, if it shall not be a Lender, shall deliver to the Agent an Administrative Questionnaire. The Agent shall provide the Borrower and each Lender with written notice of any change in (or new) address of a Lender disclosed in an Administrative Questionnaire. Subject to acceptance and recording thereof pursuant to Section 9.4(d), from and after the effective date specified in each Assignment and Transfer, the assignee thereunder shall be a party hereto and, to the extent of the interest assigned by such Assignment and Transfer, shall have all of the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Transfer, be released from its obligations under this Agreement (and, in the case of an Assignment and Transfer covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be entitled to the benefits of Sections 2.12, 2.13, and 2.14 and 9.3). Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this Section 9.4 shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with Section 9.4(e).
- (c) The Agent, acting for this purpose as an agent of the Borrower, shall maintain at one of its offices in Toronto, Ontario a copy of each Assignment and Transfer delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitment of, and principal amount of the Loans and the amount of the Reimbursement Obligations owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, and the Borrower, the Agent, and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement. The Register shall be available for inspection by the Borrower and any Lender at any reasonable time and from time to time upon reasonable prior notice.

- (d) Upon its receipt of a duly completed Assignment and Transfer executed by an assigning Lender and an assignee, the assignee's completed Administrative Questionnaire (unless the assignee shall already be a Lender hereunder), the processing and recordation fee referred to in Section 9.4(b) and any written consent to such assignment required by Section 9.4(b), the Agent shall accept such Assignment and Transfer and record the information contained therein in the Register. No assignment shall be effective for purposes of this Agreement unless it has been recorded in the Register as provided in this Section 9.4(d).
- (e) Any Lender may, without notice to the Borrower or the consent of the Borrower or the Agent, sell participations to one or more Persons (a "Participant") in all or a portion of such Lender's rights and obligations under this Agreement and the other Loan Documents (including all or a portion of its Commitment and the Loans owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, and (iii) the Borrower, the Agent, and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, modification or waiver described in the first proviso to Section 9.2(b) that affects such Participant. Subject to Section 9.4(f), the Borrower agrees that each Participant shall be entitled to the benefits of Sections 2.12, 2.13 and 2.14 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to this Section 9.4(b). To the extent permitted by Applicable Law, each Participant also shall be entitled to the benefits of Section 9.8 as though it were a Lender, provided that such Participant agrees to be subject to Section 2.15(c) as though it were a Lender.
- (f) A Participant shall not be entitled to receive any greater payment under Section 2.13 or 2.14 than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of Section 2.14 unless the Borrower is notified of the participation sold to such Participant and such Participant agrees, for the benefit of the Borrower, to comply with Section 2.14(e) as though it were a Lender.
- (g) Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank, and Section 9.4 shall not apply to any such pledge or assignment of a security interest; provided that no such pledge or assignment of a security interest shall release a Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.
- (h) Any assignment or grant of a participation pursuant to Section 9.4 shall constitute neither a repayment by the Borrower to the assigning or granting Lender of any Loan included therein, nor a new advance of any such Loan to the Borrower by such Lender or by the Assignee or Participant, as the case may be. The parties acknowledge that the Borrower's obligations hereunder with respect to any such Loans will continue and will not constitute new obligations as a result of such assignment or participation.

- 9.5 <u>Survival</u>. All covenants, agreements, representations and warranties made by the Borrower herein and in the certificates or other instruments delivered in connection with or pursuant to this Agreement shall be considered to have been relied upon by the other parties hereto and shall survive the execution and delivery of this Agreement and the making of any Loans and issuance of any Letters of Credit, regardless of any investigation made by any such other party or on its behalf and notwithstanding that the Agent or any Lender may have had notice or knowledge of any Default or incorrect representation or warranty at the time any credit is extended hereunder, and shall continue in full force and effect as long as the principal of or any accrued interest on any Loan or any fee or any other amount payable under this Agreement is outstanding and unpaid or any Letter of Credit is outstanding and so long as the Commitments have not expired or terminated. Sections 2.5(m), 2,12, 2.13, 2.14 and 9.3 and Article 8 shall survive and remain in full force and effect, regardless of the consummation of the Transactions, the repayment of the Loans, the expiration or termination of the Letters of Credit and the Commitments or the termination of this Agreement or any provision hereof.
- 9.6 Counterparts; Integration; Effectiveness. This Agreement may be executed in counterparts (and by different parties hereto on different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Agreement, the other Loan Documents and any separate letter agreements with respect to fees payable to the Agent, constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof. Except as provided in Section 4.1, this Agreement shall become effective when it shall have been executed by the Agent and when the Agent shall have received counterparts hereof which, when taken together, bear the signatures of each of the other parties hereto, and thereafter shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns. Delivery of an executed original counterpart of a signature page of this Agreement by facsimile or other electronically scanned method of delivery shall be as effective as delivery of a manually executed original counterpart of this Agreement.
- **9.7** Severability. Any provision of this Agreement held to be invalid, illegal or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such invalidity, illegality or unenforceability without affecting the validity, legality and enforceability of the remaining provisions hereof, and the invalidity of a particular provision in a particular jurisdiction shall not invalidate such provision in any other jurisdiction.
- **9.8** Right of Set-Off. If an Event of Default shall have occurred and be continuing, each Lender and each of its Affiliates is hereby authorized at any time and from time to time, to the fullest extent permitted by Applicable Law, to set off and apply any and all deposits (general or special, time or demand, provisional or final) at any time held and other obligations at any time owing by such Lender or Affiliate to or for the credit or the account of the Borrower against any of and all of the obligations of the Borrower now or hereafter existing under this Agreement held by such Lender, irrespective of whether or not such Lender shall have made any demand under this Agreement and although such obligations may be unmatured. The rights of each Lender under this Section are in addition to other rights and remedies (including other rights of set off) which such Lender may have.

9.9 Governing Law; Jurisdiction; Consent to Service of Process.

- (a) This Agreement shall be construed in accordance with and governed by the Laws of the Province of British Columbia.
- (b) The Borrower hereby irrevocably and unconditionally submits, for itself and its property, to the non-exclusive jurisdiction of the Courts of the Province of British Columbia, and any appellate court thereof, in any action or proceeding arising out of or relating to this Agreement, or any other Loan Document or for recognition or enforcement of any judgment, and each of the parties hereto hereby irrevocably and unconditionally agrees that all claims in respect of any such action or proceeding may be heard and determined in British Columbia. Each of the parties hereto agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Applicable Law. Nothing in this Agreement shall affect any right that the Agent or any Lender may otherwise have to bring any action or proceeding relating to this Agreement or any other Loan Document against the Borrower or its properties in the courts of any other jurisdiction.
- (c) The Borrower hereby irrevocably and unconditionally waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any suit, action or proceeding arising out of or relating to this Agreement in any court referred to in this Section 9.9. Each of the parties hereto hereby irrevocably waives, to the fullest extent permitted by Applicable Law, any *forum non conveniens* defence to the maintenance of such action or proceeding in any such court.
- (d) Each party to this Agreement irrevocably consents to service of process in the manner provided for notices in Section 9.1. Nothing in this Agreement will affect the right of any party to this Agreement to serve process in any other manner permitted by Applicable Law.
- 9.10 WAIVER OF JURY TRIAL. EACH PARTY HERETO HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, ANY OTHER LOAN DOCUMENT, OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.
- **9.11** <u>Headings</u>. Article and Section headings and the Table of Contents used herein are for convenience of reference only, are not part of this Agreement and shall not affect the construction of, or be taken into consideration in interpreting, this Agreement.

9.12 Confidentiality. Each of the Agent and each Lender agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to each of their, and each of their Affiliates', directors, officers, employees, agents and advisors, including accountants, legal counsel and other advisors (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent requested by any rating agency, regulatory authority or other Governmental Authority, or their legal counsel, (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process, (d) to any other party to this Agreement, (e) in connection with the exercise of any remedies under any Loan Document or any suit, action or proceeding relating to any Loan Document or the enforcement of rights thereunder, (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any actual or prospective assignee of or Participant (or such assignee's or Participant's advisors) in any of its rights or obligations under this Agreement, or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to the Borrower and its obligations, (g) any financial institution, credit reporting agency or credit bureau, (h) any Person with whom the Borrower may have or proposes to have financial dealings, or (i) with the consent of the Borrower. For greater certainty, the Borrower acknowledges that from time to time, the Borrower may request the Agent to facilitate the provision of certain financial services offered by CIBC (the "CIBC Services"). In such circumstances, CIBC policies and procedures ("CIBC's Policies") will apply in respect of all transactions undertaken by CIBC in connection with the provision of the CIBC Services, including any required due diligence investigation and related business approval processes conducted in respect of the Borrower. The Borrower consents to the disclosure of Information by the Agent to CIBC for the purpose of facilitating compliance with CIBC's Policies. For the purposes of this Section, "Information" means all information received from the Borrower or any Credit Party relating to the Borrower, any of the Credit Parties, or their respective business, other than Information that is (i) is or becomes publicly available other than as a result of a breach of this Section, (ii) any such information that is or becomes available to the Agent, the Issuing Bank, any Lender or CIBC on a nonconfidential basis prior to disclosure by the Borrower, or (iii) was already in the possession of the Agent, the Issuing Bank, or any CIBC Lender prior to its disclosure by the Borrower; provided that, in the case of information received from the Borrower after the date hereof, such information is clearly identified as confidential in writing at the time of delivery. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

9.13 Press Releases and Related Materials. Each Credit Party agrees that neither it nor its Affiliates will in the future issue any press releases or other public disclosure using the name of the Agent or referring to this Agreement, or the other Loan Documents without prior notice to the Agent unless (and only to the extent that) such Credit Party or Affiliate is required to do so by law or pursuant to any applicable regulatory policy protocol or rule. Each Credit Party consents to the publication by the Agent or any Lender of advertising material relating to the financing transactions contemplated by this Agreement using its name, product photographs, logo or trademark. The Agent or such Lender shall provide a draft of any advertising material to Borrower for review and comment prior to the publication thereof. The Agent reserves the right to provide to industry trade organizations information necessary and customary for inclusion in league table measurements.

9.14 Anti-Money Laundering Legislation.

- (a) The Borrower acknowledges that, pursuant to the *Proceeds of Crime (Money Laundering) and Terrorist Financing Act* (Canada) and other applicable anti-money laundering, anti-terrorist financing, government sanction and "know your client" Laws (collectively, including any guidelines or orders thereunder, "**AML Legislation**"), the Lenders and the Agent may be required to obtain, verify and record information regarding the Borrower, its directors, authorized signing officers, direct or indirect shareholders or other Persons in control of the Borrower, and the transactions contemplated hereby. The Borrower shall promptly provide all such information, including supporting documentation and other evidence, as may be reasonably requested by any Lender or the Agent, or any prospective assignee or participant of a Lender or the Agent, in order to comply with any applicable AML Legislation, whether now or hereafter in existence.
- (b) If the Agent has ascertained the identity of the Borrower or any authorized signatories of the Borrower for the purposes of applicable AML Legislation, then the Agent:
 - (i) shall be deemed to have done so as an agent for each Lender, and this Agreement shall constitute a "written agreement" in such regard between each Lender and the Agent within the meaning of applicable AML Legislation; and
 - (ii) shall provide to each Lender copies of all information obtained in such regard without any representation or warranty as to its accuracy or completeness.

Notwithstanding the preceding sentence and except as may otherwise be agreed in writing, each of the Lenders agrees that the Agent has no obligation to ascertain the identity of the Borrower or any authorized signatories of the Borrower on behalf of any Lender, or to confirm the completeness or accuracy of any information it obtains from the Borrower or any such authorized signatory in doing so.

- **9.15** No Strict Construction. The parties hereto have participated jointly in the negotiation and drafting of this Agreement. If an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties hereto and no presumption or burden of proof shall arise favouring or disfavouring any party by virtue of the authorship of any provisions of this Agreement.
- **9.16 Paramountcy**. In the event of any inconsistency between the provisions of this Agreement and the provisions of any other Loan Document, the provisions of this Agreement shall prevail.
- 9.17 LIMITATION OF LIABILITY. NO CLAIM MAY BE MADE BY THE BORROWER, ANY OTHER CREDIT PARTY OR, ANY LENDER OR OTHER PERSON AGAINST THE AGENT, ANY LENDER, OR THE AFFILIATES, DIRECTORS, OFFICERS, EMPLOYEES, OR AGENTS OF ANY OF THEM FOR ANY SPECIAL, INDIRECT, CONSEQUENTIAL OR PUNITIVE DAMAGES IN RESPECT OF ANY CLAIM FOR BREACH OF CONTRACT OR ANY OTHER THEORY OF LIABILITY ARISING OUT OF OR RELATED TO THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, OR ANY ACT, OMISSION OR EVENT OCCURRING IN CONNECTION THEREWITH, AND THE BORROWER, EACH GUARANTOR, EACH LENDER AND THE AGENT HEREBY WAIVES, RELEASES AND AGREES NOT TO SUE UPON ANY CLAIM FOR SUCH DAMAGES, WHETHER OR NOT ACCRUED AND WHETHER OR NOT KNOWN OR SUSPECTED TO EXIST IN ITS FAVOUR.

[Remainder of this page intentionally left blank. Signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed by their respective authorized officers as of the day and year first above written.

Address: ZELLSTOFF CELGAR LIMITED
PARTNERSHIP, By its General Partner,
ZELLSTOFF CELGAR LIMITED

Attention:

Facsimile No.: By: /s/ Genevieve Stannus

Name: Genevieve Stannus

Title: Treasurer

SIGNATURE PAGE TO THIRD AMENDED AND RESTATED CREDIT AGREEMENT DATED AS OF JULY 16, 2018 BETWEEN CANADIAN IMPERIAL BANK OF COMMERCE AND ZELLSTOFF CELGAR LIMITED PARTNERSHIP.

Signature Page to Third A&R Credit Agreement

Address:

199 Bay Street, 4th Floor Toronto, Ontario M5L 1A9

Attention: Portfolio Manager

Facsimile No.:(416) 861-9422

CANADIAN IMPERIAL BANK OF COMMERCE, as Agent and as Lender

By: /s/ David Carson

Name: David Carson

Title: Authorized Signatory

By: /s/ Anthony Tsuen

Name: Anthony Tsuen Title: Authorized Signatory

SIGNATURE PAGE TO THIRD AMENDED AND RESTATED CREDIT AGREEMENT DATED AS OF JULY 16, 2018 BETWEEN CANADIAN IMPERIAL BANK OF COMMERCE AND ZELLSTOFF CELGAR LIMITED PARTNERSHIP.

Signature Page to Third A&R Credit Agreement

-100-

I, David M. Gandossi, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Mercer International Inc. (the "Registrant");
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the Registrant as of, and for, the periods presented in this report;
- 4. The Registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the Registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the Registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the Registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the Registrant's internal control over financial reporting that occurred during the Registrant's most recent fiscal quarter (the Registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the Registrant's internal control over financial reporting; and
- 5. The Registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Registrant's auditors and the audit committee of the Registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the Registrant's internal control over financial reporting.

Dated: July 26, 2018

/s/ David M. Gandossi
David M. Gandossi
Chief Executive Officer

I, David K. Ure, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Mercer International Inc. (the "Registrant");
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the Registrant as of, and for, the periods presented in this report;
- 4. The Registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f)) and 15d-15(f)) for the Registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the Registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the Registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the Registrant's internal control over financial reporting that occurred during the Registrant's most recent fiscal quarter (the Registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the Registrant's internal control over financial reporting; and
- 5. The Registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Registrant's auditors and the audit committee of the Registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the Registrant's internal control over financial reporting.

Dated: July 26, 2018

/s/ David K. Ure
David K. Ure
Chief Financial Officer

- I, David M. Gandossi, Chief Executive Officer of Mercer International Inc. (the "Company"), certify pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the *Sarbanes-Oxley Act of 2002*, that, to my knowledge:
 - (1) the Quarterly Report on Form 10-Q of the Company for the quarterly period ended June 30, 2018 (the "Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
 - (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: July 26, 2018

/s/ David M. Gandossi
David M. Gandossi
Chief Executive Officer

A signed original of this written statement required by Section 906 of the *Sarbanes-Oxley Act of 2002* has been provided to Mercer International Inc. and will be retained by Mercer International Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

This certification accompanies the Report pursuant to Section 906 of the *Sarbanes-Oxley Act of 2002* and shall not, except to the extent required by the *Sarbanes-Oxley Act of 2002*, be deemed filed by the Company for purposes of Section 18 of the *Securities Exchange Act of 1934*, as amended.

- I, David K. Ure, Chief Financial Officer of Mercer International Inc. (the "Company"), certify pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the *Sarbanes-Oxley Act of 2002*, that, to my knowledge:
 - (1) the Quarterly Report on Form 10-Q of the Company for the quarterly period ended June 30, 2018 (the "Report") fully complies with the requirements of Section 13(a) or 15 (d) of the *Securities Exchange Act of 1934*; and
 - (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: July 26, 2018

/s/ David K. Ure	
David K. Ure	
Chief Financial Officer	•

A signed original of this written statement required by Section 906 of the *Sarbanes-Oxley Act of 2002* has been provided to Mercer International Inc. and will be retained by Mercer International Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

This certification accompanies the Report pursuant to Section 906 of the *Sarbanes-Oxley Act of 2002* and shall not, except to the extent required by the *Sarbanes-Oxley Act of 2002*, be deemed filed by the Company for purposes of Section 18 of the *Securities Exchange Act of 1934*, as amended.